### **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

		FC	ORM 10-Q		
×	QUARTERLY REPORT PURSUA	ANT TO SECTIO	N 13 OR 15(d) OF THE S	ECURITIES EXCHA	ANGE ACT OF 1934
		For the quarte	rly period ended June 30, 2021 OR		
	TRANSITION REPORT PURSUA	ANT TO SECTIO	N 13 OR 15(d) OF THE S	ECURITIES EXCHA	ANGE ACT OF 1934
			ition period from to ion file number 001-38150		
	KA		MACEUTICALS egistrant as specified in its charter)	, INC.	
	<b>Delaware</b> (State or other jurisdiction incorporation or organizat			27-0604595 (I.R.S. Employer Identification No.)	
	490 Arsenal Way, Suite ! Watertown, MA (Address of principal executive			<b>02472</b> (Zip Code)	
		(Registrant's telep	(781) 996-5252 shone number, including area code)		
		Securities registered	pursuant to Section 12(b) of the	Act	
	<u>Title of each class</u> Common Stock, \$0.001 par value per share	<u>T</u>	<u>rading symbol(s)</u> KALA		ang <u>e on which registered</u> lobal Select Market
	e by check mark whether the registrant (1) has fi 1ths (or for such shorter period that the registrant				
	e by check mark whether the registrant has subm ceding 12 months (or for such shorter period that				405 of Regulation S-T during
	e by check mark whether the registrant is a large ny. See the definitions of "large accelerated filer,				
Large a	accelerated filer  Acce	lerated filer □	Non-accelerated filer ∑	3	Smaller reporting company ⊠ Emerging growth company ⊠
	merging growth company, indicate by check mark ting standards provided pursuant to Section 13(a			on period for complying with	any new or revised financial
Indicat	e by check mark whether the registrant is a shell	company (as defined in	Rule 12b-2 of the Exchange Act). Y	es □ No ⊠	
There	were 64,971,096 shares of Common Stock, \$0.00	11 par value per share, o	utstanding as of August 4, 2021.		
_					

#### TABLE OF CONTENTS

PART I – F	INANCIAL INFORMATION	Page 3
Item 1.	Financial Statements (Unaudited)	3
	Condensed Consolidated Balance Sheets as of June 30, 2021 and December 31, 2020	3
	Condensed Consolidated Statement of Operations and Comprehensive Loss for the three and six months ended June 30, 2021 and 2020	4
	Condensed Consolidated Statements of Changes in Stockholders' Equity for the three and six months ended June 30, 2021 and 2020	5
	Condensed Consolidated Statements of Cash Flows for the six months ended June 30, 2021 and 2020	7
	Notes to Condensed Consolidated Financial Statements	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	23
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	38
Item 4.	Controls and Procedures	38
PART II –	OTHER INFORMATION	38
Item 1.	<u>Legal Proceedings</u>	38
Item 1A.	Risk Factors	39
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	86
Item 6.	<u>Exhibits</u>	87
<u>SIGNATUI</u>	RES	88

#### PART I – FINANCIAL INFORMATION

#### Item 1. Financial Statements.

# KALA PHARMACEUTICALS, INC. CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED)

(In thousands, except share and per share amounts)

		June 30, 2021	De	cember 31, 2020
Assets				
Current assets:				
Cash and cash equivalents	\$	144,635	\$	77,264
Short-term investments		5,000		76,276
Accounts receivable, net		11,859		9,604
Inventory		8,046		5,229
Prepaid expenses and other current assets		2,180		3,006
Total current assets		171,720		171,379
Non-current assets:				
Property and equipment, net		3,416		3,166
Long-term inventory		9,157		6,219
Right-of-use assets		27,953		27,853
Restricted cash and other long-term assets		3,218		12,989
Total assets	\$	215,464	\$	221,606
Liabilities and Stockholders' Equity				
Current liabilities:				
Accounts payable	\$	2,844	\$	1,724
Accrued expenses and other current liabilities		21,255		18,971
Current portion of lease liabilities		1,977		1,530
Total current liabilities		26,076		22,225
Long-term liabilities:				
Long-term lease liabilities		27,063		27,143
Long-term debt		78,055		72,243
Total long-term liabilities		105,118		99,386
Total liabilities		131,194		121,611
Commitments and Contingencies (Note 14)				
Stockholders' equity:				
Common stock, \$0.001 par value; 120,000,000 shares authorized as of June 30, 2021 and				
December 31, 2020; 64,770,219 and 58,915,375 shares issued and outstanding as of June				
30, 2021 and December 31, 2020, respectively		65		59
Additional paid-in capital		550,898		499,715
Accumulated other comprehensive income		_		4
Accumulated deficit	_	(466,693)		(399,783)
Total stockholders' equity		84,270		99,995
Total liabilities and stockholders' equity	\$	215,464	\$	221,606

#### KALA PHARMACEUTICALS, INC.

#### CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS

#### (UNAUDITED)

#### (In thousands, except share and per share amounts)

	Three Months Ended June 30,					Six Months Ended June 30,			
		2021	_	2020		2021	_	2020	
Product revenues, net	\$	3,051	\$	833	\$	6,317	\$	1,904	
Costs and expenses:									
Cost of product revenues		1,016		759		1,771		1,113	
Selling, general and administrative		27,986		15,301		55,685		30,709	
Research and development		3,094		6,053		6,220		11,487	
Total costs and expenses		32,096		22,113		63,676		43,309	
Loss from operations		(29,045)		(21,280)		(57,359)		(41,405)	
Other income (expense):									
Interest and other income		33		102		76		400	
Interest and other expense		(2,091)		(2,134)		(4,232)		(4,262)	
Loss on extinguishment of debt		(5,395)		<u> </u>		(5,395)		_	
Total interest and other expense		(7,453)		(2,032)		(9,551)		(3,862)	
Net loss	\$	(36,498)	\$	(23,312)	\$	(66,910)	\$	(45,267)	
Net loss per share—basic and diluted	\$	(0.57)	\$	(0.42)	\$	(1.06)	\$	(0.94)	
Weighted average shares outstanding—basic and diluted	$\epsilon$	54,554,506		55,703,882		63,113,194		48,232,933	
Net loss	\$	(36,498)	\$	(23,312)	\$	(66,910)	\$	(45,267)	
Other comprehensive loss:									
Change in unrealized gains on investments		(3)		17		(4)		17	
Total other comprehensive loss		(3)		17		(4)		17	
Total comprehensive loss	\$	(36,501)	\$	(23,295)	\$	(66,914)	\$	(45,250)	

#### ${\bf KALA\ PHARMACEUTICALS,\ INC.}$

## CONDENSED CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY (UNAUDITED)

(In thousands, except share and per share amounts)

	Three Months Ended June 30, 2021												
	Common	Stock	•	Additional		Other				Total			
	\$0.001 Par	Valu	e	Paid-In	Co	omprehensive	A	ccumulated	Sto	ockholders'			
	Shares Amount			Capital		Income		Deficit	Equity				
Balance as of March 31, 2021	63,805,108	\$	64	\$ 539,920	\$	3	\$	(430,195)	\$	109,792			
At the market offering, net of sales													
agent commission of \$170	837,257		1	6,020		_				6,021			
Exercise of stock options	20,074		—	64		_		_		64			
Issuance of vested restricted stock													
units	107,780		—	_		_				_			
Stock-based compensation expense	_		—	4,894		_		_		4,894			
Change in fair value of investments	_		_	_		(3)				(3)			
Net loss			_					(36,498)		(36,498)			
Balance as of June 30, 2021	64,770,219	\$	65	\$ 550,898	\$		\$	(466,693)	\$	84,270			

	Three Months Ended June 30, 2020													
		Accumulated												
	Common	Stock		Additional		Other				Total				
	\$0.001 Par	<b>Value</b>		Paid-In	C	omprehensive	Accumulated		Stockholders'					
	Shares	Shares Amount			Income			Deficit		Equity				
Balance as of March 31, 2020	54,585,636	\$	55	\$ 458,866	\$	_	\$	(317,411)	\$	141,510				
Exercise of stock options	249,870		_	772		_		_		772				
Common stock offering, net of														
offering costs of \$464	979,371		1	7,215		_		_		7,216				
Stock-based compensation														
expense	_		—	2,774		_				2,774				
Warrant exercises	16,144		—	_		_		_		_				
Change in fair value of														
investments	_		—	_		(17)				(17)				
Net loss			_					(23,312)		(23,312)				
Balance as of June 30, 2020	55,831,021	\$	56	\$ 469,627	\$	(17)	\$	(340,723)	\$	128,943				

	Six Months Ended June 30, 2021											
	Common S	Stock		Additional		Accumulated			Total			
	\$0.001 Par	\$0.001 Par Value		Paid-In	Other Comprehensive		Accumulated	Sto	ockholders'			
	Shares	Shares Amo		Capital	Income		Deficit		Equity			
Balance as of December 31, 2020	58,915,375	\$	59	\$ 499,715	\$	4	\$ (399,783)	\$	99,995			
At the market offering, net of												
offering costs \$1,161	5,583,329		6	40,724		_	_		40,730			
Exercise of stock options	88,888		_	248		_	_		248			
Issuance of vested restricted stock												
units	107,780		_	_		_	_		_			
Issuance of common stock under												
employee stock purchase plan	74,847		_	431		_	_		431			
Stock-based compensation expense	_		_	9,780		_	_		9,780			
Change in fair value of investments	_		—	_		(4)	_		(4)			
Net loss	_		_	_		_	(66,910)		(66,910)			
Balance as of June 30, 2021	64,770,219	\$	65	\$ 550,898	\$	_	\$ (466,693)	\$	84,270			

				Six Mont	ths E	Ended June 30, 2020			
	Common S	Stock		Additional		Accumulated		Total	
	\$0.001 Par	Valu	e	Paid-In	Other Comprehensive		Accumulated	Sto	ockholders'
	Shares	An	nount	Capital	Income		Deficit		Equity
Balance as of December 31, 2019	36,086,254	\$	36	\$ 325,112	\$	_	\$ (295,456)	\$	29,692
At the market offering, net of sales									
agent commission of \$388	2,352,671		3	12,543		_	_		12,546
Exercise of stock options	311,028		_	938		_	_		938
Common stock offering, net of									
issuance cost and underwriting fees									
of \$8,475	16,979,371		17	125,406		_	_		125,423
Issuance of common stock under									
employee stock purchase plan	85,553		_	270		_	_		270
Stock-based compensation expense	_		_	5,358		_	_		5,358
Warrant exercises	16,144		_	_		_	_		_
Change in fair value of investments	_		_	_		(17)	_		(17)
Net loss			_			<u> </u>	(45,267)		(45,267)
Balance as of June 30, 2020	55,831,021	\$	56	\$ 469,627	\$	(17)	\$ (340,723)	\$	128,943

## KALA PHARMACEUTICALS, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

(In thousands)

		Six Mont June		nded
	_	2021	_	2020
Cash flows from operating activities:		(55.04.0)		
Net loss	\$	(66,910)	\$	(45,267)
Adjustments to reconcile net loss to cash used in operating activities:				
Depreciation and amortization		504		454
Non-cash operating lease cost		1,118		952
Loss on extinguishment of debt		5,395		
Amortization of debt discount and other non-cash interest		645		513
Stock-based compensation		9,412		5,271
Amortization of discount on available-for-sale securities		22		_
Change in operating assets and liabilities:				
Accounts receivable		(2,255)		7,282
Prepaid expenses and other current assets		826		1,351
Inventory		(5,387)		(1,306)
Accounts payable		1,250		(973)
Accrued expenses and other current liabilities		1,966		(6,782)
Lease liabilities and other long-term liabilities	_	(834)		(629
Net cash used in operating activities		(54,248)		(39,134
Cash flows from investing activities:				
Purchases of property and equipment and other assets		(545)		(915
Purchases of short-term investments		_		(56,535
Proceeds from sales or maturities of short-term investments		71,250		_
Net cash provided by (used in) investing activities		70,705		(57,450
Cash flows from financing activities:	_			
Proceeds from issuance of debt, net of debt issuance costs of \$2,218		77,782		
Payment of principal, prepayment premium and exit fee on debt		(78,010)		_
Proceeds from common stock offerings, net of offering costs		40,730		137,969
Payment of principal on finance lease		(17)		(16
Proceeds from exercise of stock options and issuance of common stock under employ	/ee			•
stock purchase plan		679		1,207
Net cash provided by financing activities	_	41,164		139,160
Net increase in cash, cash equivalents and restricted cash:		57,621		42,576
Cash, cash equivalents and restricted cash at beginning of period		89,756		98,031
Cash, cash equivalents and restricted cash at end of period	\$	147,377	\$	140,607
	<u> </u>	117,077	<u> </u>	110,007
Reconciliation of cash, cash equivalents and restricted cash:	¢	147,377	ď	140,607
Cash, cash equivalents, and restricted cash at end of period	Þ	(2,742)	Ф	
Less restricted cash (Note 8)	<u></u>		<u>_</u>	(12,584
Cash and cash equivalents at end of period	<u>\$</u>	144,635	<u>&gt;</u>	128,023
Non-cash investing and financing activities:				
Purchases of property and equipment in accrued expenses	\$	318	\$	_
Supplemental disclosure:				
Cash paid for interest	\$	3,030	\$	3,749
Right-of-use assets obtained in exchange of operating lease obligations	· ·	1,218		
3		, -		

(Unaudited)

(In thousands, except share and per share amounts)

#### 1. NATURE OF BUSINESS AND BASIS OF PRESENTATION

**Nature of Business**— Kala Pharmaceuticals, Inc. (the "Company") was incorporated on July 7, 2009, and is a biopharmaceutical company focused on the discovery, development and commercialization of innovative therapies for diseases of the eye. The Company has applied its AMPPLIFY® mucus-penetrating particle ("MPP") Drug Delivery Technology to loteprednol etabonate ("LE"), a corticosteroid designed for ocular applications, resulting in the U.S. Food and Drug Administration's (the "FDA") approval of EYSUVIS® (loteprednol etabonate ophthalmic suspension) 0.25%, for the short-term (up to two weeks) treatment of the signs and symptoms of dry eye disease, and INVELTYS® (loteprednol etabonate ophthalmic suspension) 1% as the first and only topical twice-daily ocular corticosteroid for treatment of post-operative inflammation and pain following ocular surgery.

In January 2019, the Company launched its first commercial product, INVELTYS, in the United States and began shipping its second commercial product, EYSUVIS, to wholesalers in the United States in late December 2020 with the full promotional launch commencing in early January 2021. The Company is engaged in the commercialization of EYSUVIS and INVELTYS, research and development activities, raising capital and recruiting skilled personnel. The Company is subject to a number of risks similar to those of other companies conducting high-risk, research and development of pharmaceutical product candidates and launching products for the first time. Principal among these risks are dependence on key individuals and intellectual property, competition from other products and companies and the technical risks associated with the successful research, development and marketing of its product candidates. The Company's success is dependent upon its ability to successfully commercialize its products, the success of its research and development efforts, its ability to obtain regulatory approval of its product candidates, its ability to raise additional capital when needed and, ultimately, attain profitable operations.

The Company is also progressing its pipeline of proprietary preclinical development programs targeted to address front and back of the eye diseases. These preclinical development programs, all of which are new chemical entities, include its receptor Tyrosine Kinase Inhibitor program, that is designed to inhibit the vascular endothelial growth factor pathway, for the treatment of retinal diseases, including wet age-related macular degeneration; its selective glucocorticoid receptor modulators, which are a novel class of therapies designed to modify the downstream activity of the receptors to exhibit the anti-inflammatory and immunomodulatory properties of the corticosteroid class of therapies without their associated side effects; and its novel surface targeting steroid designed to target the ocular surface and thus have the potential to have fewer side effects compared to traditional topical steroids. The Company owns all intellectual property and worldwide rights to these pipeline preclinical development programs.

**Liquidity**— Since inception, the Company has incurred significant losses from operations and negative cash flows from operations. As of June 30, 2021, the Company had an accumulated deficit of \$466,693. As the Company commenced a full promotional launch of EYSUVIS in early January 2021 and commercially launched its first product, INVELTYS, in January 2019, it has had only limited revenues to date from product sales and has financed operations primarily through proceeds from its initial public offering of common stock ("IPO"), follow-on public common stock offerings and sales of its common stock under its at-the-market offering ("ATM Offering") facility, private placements of preferred stock, borrowings under credit facilities, convertible promissory notes and warrants. The Company has devoted substantially all of its financial resources and efforts to research and development, including preclinical studies and clinical trials, and engaging in activities to commercialize EYSUVIS and INVELTYS. Although the Company expects to continue to generate revenue from sales of EYSUVIS and INVELTYS, there can be no assurance as to the amount or timing of any such revenue, and it expects to continue to incur significant expenses and operating losses. Net losses may fluctuate significantly from quarter-to-quarter and year-to-year.

The Company expects that its cash, cash equivalents and short-term investments as of June 30, 2021, together with anticipated net revenue from sales of EYSUVIS and INVELTYS, will enable it to fund its operating expenses, debt service obligations and capital expenditure requirements for at least twelve months from the date these condensed consolidated financial statements were issued. This evaluation is based on relevant conditions and events that are known and reasonably knowable at the date that the condensed consolidated financial statements are issued. As a result, the Company could deplete its available capital resources sooner than it currently expects.

(Unaudited)

(In thousands, except share and per share amounts)

COVID-19 — In order to safeguard the health of its employees from the ongoing novel coronavirus pandemic, or COVID-19, the Company is following, and will continue to follow, recommendations from the U.S. Centers for Disease Control and Prevention, as well as federal, state and local governments, regarding working-from-home practices for non-essential employees. As a result, all office-based personnel are permitted to work from home, and the Company's laboratory facilities that support its early-stage research activities are being utilized as necessary. In addition, the Company previously suspended its sales force from substantially all in-person interactions with physicians and was limited to conducting educational and promotional activities virtually. The Company's sales force has since resumed substantially all in-person interactions in the field, but if the Company suspends all or some in-person interactions with physicians in the future, or to the extent physicians limit in-person interactions, the Company may be limited to conducting educational and promotional activities virtually, which may continue to hamper its ability to market and commercialize EYSUVIS and INVELTYS

In addition, government restrictions have at times led to moratoria on elective ocular surgeries in many jurisdictions, which had significantly reduced, and may in the future continue to significantly reduce, the demand for INVELTYS, which is indicated for the treatment of post-operative inflammation and pain following ocular surgery. The extent of the impact of the COVID-19 pandemic on the Company's commercialization efforts of EYSUVIS and INVELTYS and its operational and financial performance will depend on certain developments, including the length and severity of this pandemic, the timing and extent of any resurgence of the COVID-19 virus or any variant strains of the virus, the availability and effectiveness of vaccines, and the impact of the foregoing on our customers, employees and vendors and government agencies, all of which are uncertain and cannot be predicted. The Company cannot reasonably estimate the extent to which the disruption may materially impact its condensed consolidated results of operations or financial position.

Use of Estimates—The preparation of condensed consolidated financial statements in conformity with accounting principles generally accepted in the United States of America ("U.S. GAAP") requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue, expense, and related disclosures. The Company bases estimates and assumptions on historical experience when available and on various factors that it believes to be reasonable under the circumstances. The Company evaluates its estimates and assumptions on an ongoing basis. Estimates and assumptions relied upon in preparing these condensed consolidated financial statements relate to, but are not limited to, revenue recognition, inventory, the present value of lease liabilities and the corresponding right-of-use assets, the fair value of warrants, stock-based compensation, accrued expenses and the recoverability of the Company's net deferred tax assets and related valuation allowance. Actual results may differ from these estimates under different assumptions or conditions.

**Net Loss per Share**—Basic net loss per share is computed using the weighted average number of common shares outstanding during the period. Diluted net loss per share is computed using the weighted average number of common shares outstanding during the period and, if dilutive, the weighted average number of potential shares of common stock, including the assumed exercise of stock options and warrants and the issuance of unvested restricted stock units ("RSUs") and performance-based restricted stock units ("PSUs").

The weighted average number of common shares included in the computation of diluted net loss gives effect to all potentially dilutive common equivalent shares, including outstanding stock options, warrants and unvested RSUs and PSUs. Common stock equivalent shares are excluded from the computation of diluted net loss per share if their effect is antidilutive. In periods in which the Company reports a net loss attributable to common stockholders, diluted net loss per share attributable to common stockholders is the same as basic net loss per share attributable to common stockholders since dilutive common shares are not assumed to have been issued if their effect is anti-dilutive. The Company reported a net loss attributable to common stockholders for each of the three and six months ended June 30, 2021 and 2020.

(Unaudited)

(In thousands, except share and per share amounts)

As of June 30, 2021 and 2020, potentially dilutive securities excluded from the calculation of diluted net loss per share because including such securities would have an anti-dilutive effect consisted of outstanding options to purchase 10,043,748 and 8,573,573 shares of the Company's common stock, respectively, an aggregate of 1,282,830 and 957,097 unvested RSUs and PSUs, respectively, and an aggregate of 215,172 and 248,505 unexercised warrants, respectively.

Unaudited Interim Financial Information—The condensed consolidated financial statements of the Company included herein have been prepared, without audit, pursuant to the rules and regulations of the Securities Exchange Commission ("SEC"). Certain information and footnote disclosures normally included in financial statements prepared in accordance with U.S. GAAP have been condensed or omitted from this report, as is permitted by such rules and regulations. The accompanying condensed consolidated financial statements reflect all adjustments consisting of normal, recurring adjustments, that are necessary for a fair presentation of the financial position, results of operations, statement of stockholders' equity and cash flows for the interim periods presented. Interim results are not necessarily indicative of results for a full year. Accordingly, these condensed consolidated financial statements should be read in conjunction with the financial statements and notes thereto included in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2020 (the "Annual Report").

The unaudited condensed consolidated financial statements include the accounts of Kala Pharmaceuticals, Inc. and its wholly owned subsidiary, Kala Pharmaceuticals Security Corporation. All intercompany transactions and balances have been eliminated in consolidation.

#### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The Company's significant accounting policies are described in Note 2, "Summary of Significant Accounting Policies," to the consolidated financial statements included in the Annual Report. There have been no material changes to the significant accounting policies during the three months ended June 30, 2021.

#### **Recent Accounting Pronouncements**

In June 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2016-13, *Financial Instruments - Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments* ("ASU 2016-13"). ASU 2016-13 significantly changes the impairment model for most financial assets and certain other instruments. ASU 2016-13 will require immediate recognition of estimated credit losses expected to occur over the remaining life of many financial assets, which will generally result in earlier recognition of allowances for credit losses on loans and other financial instruments. In November 2019, the FASB issued ASU 2019-10, *Financial Instruments - Credit Losses (Topic 326)*, *Derivatives and Hedging (Topic 815)*, *and Leases (Topic 842)* ("ASU 2019-10"), which is effective for Smaller Reporting Companies ("SRCs") as defined by the SEC for fiscal years beginning after December 15, 2022, including interim periods within those fiscal years. Upon adoption, beginning January 1, 2023, the Company does not expect ASU 2019-10 to have a material effect on its condensed consolidated financial statements.

#### 3. FAIR VALUE OF FINANCIAL INSTRUMENTS

The Company has short-term investments which are considered financial instruments that are measured on a recurring basis. ASC 820, *Fair Value Measurements and Disclosures*, establishes a fair value hierarchy for those instruments measured at fair value that distinguishes between assumptions based on market data (observable inputs) and its own assumptions (unobservable inputs). The hierarchy consists of three levels:

- Level 1—Quoted prices in active markets for identical assets or liabilities.
- Level 2—Observable inputs (other than Level 1 quoted prices), such as quoted prices in active markets for similar assets or liabilities, quoted prices in markets that are not active for identical or similar assets or liabilities, or other inputs that are observable or can be corroborated by observable market data.

(Unaudited)

(In thousands, except share and per share amounts)

Level 3—Unobservable inputs that are supported by little or no market activity and that are significant to
determining the fair value of the assets or liabilities, including pricing models, discounted cash flow
methodologies and similar techniques.

The Company's financial instruments consist primarily of cash equivalents and short-term investments in money market funds and short-term securities. Cash equivalents and short-term investments are reported at their respective fair values on the Company's condensed consolidated balance sheets. See Note 4, "Investments" for additional information.

The following tables set forth the fair value of the Company's financial assets by level within the fair value hierarchy as of June 30, 2021 and December 31, 2020:

			June	30, 202	1	
	I	Fair Value	Level 1		Level 2	Level 3
Assets:			 	· ·		
Cash equivalents	\$	126,118	\$ 126,118	\$	_	\$ _
Short-term investments		5,000	5,000		_	_
Total Assets	\$	131,118	\$ 131,118	\$		\$ 

		December 31, 2020									
	F	Fair Value		Level 1		Level 2		Level 3			
Assets:											
Cash equivalents	\$	63,811	\$	63,811	\$	_	\$	_			
Short-term investments		76,276		76,276		_		_			
Total Assets	\$	140,087	\$	140,087	\$	_	\$	_			

During the three and six months ended June 30, 2021 and the year ended December 31, 2020, there were no transfers between Level 1, Level 2, and Level 3.

#### 4. INVESTMENTS

Investments by security type consisted of the following as of June 30, 2021 and December 31, 2020:

		Gross Gross						
	Α	mortized	Ur	realized	Un	realized		Fair
		Cost		Gains	1	Losses		Value
U.S. government agencies securities	\$	5,000	\$	_	\$		\$	5,000
Total	\$	5,000	\$		\$		\$	5,000

(Unaudited)

(In thousands, except share and per share amounts)

		December 31, 2020						
	·			Gross		Gross		_
	A	mortized		Unrealized		Unrealized		Fair
		Cost		Gains		Losses		Value
U.S. treasury securities	\$	26,744	\$	2	\$		\$	26,746
U.S. government agencies securities		49,528		2		_		49,530
Total	\$	76,272	\$	4	\$		\$	76,276

As of June 30, 2021 and December 31, 2020, all of the Company's investments had a contractual maturity within one year. The fair value of all of the Company's investments are classified as short-term on its condensed consolidated balance sheets.

#### 5. REVENUE & ACCOUNTS RECEIVABLE, NET

The Company accounts for revenue in accordance with ASC Topic 606, *Revenue from Contracts with Customers*. Under ASC Topic 606, an entity recognizes revenue when its customer obtains control of promised goods or services, in an amount that reflects the consideration that the entity expects to be entitled in exchange for those goods or services. The Company performs the following five steps to recognize revenue under ASC Topic 606: (i) identify the contract(s) with a customer; (ii) identify the performance obligations in the contract; (iii) determine the transaction price; (iv) allocate the transaction price to the performance obligations in the contract; and (v) recognize revenue when (or as) the entity satisfies a performance obligation. The Company only recognizes revenue when it is probable that it will collect the consideration to which it is entitled in exchange for the goods or services that will be transferred to the customer.

#### Product revenues, net

The Company sells EYSUVIS and INVELTYS primarily to wholesalers in the United States (collectively, "Customers"). These Customers subsequently resell the Company's products to specialty and other retail pharmacies. In addition to agreements with Customers, the Company enters into arrangements with third-party payors that provide for government-mandated and/or privately-negotiated rebates, chargebacks and discounts for the purchase of the Company's products.

The Company also holds inventory at a third-party pharmacy on a consignment basis and records revenue when control of the product transfers to the customer upon sale to the end user. The amount of inventory held on a consignment basis as of June 30, 2021 and December 31, 2020 was immaterial.

The goods promised in the Company's product sales contracts represent a single performance obligation. The Company recognizes revenue from product sales at the point the Customer obtains control of the product, which occurs upon delivery. The transaction price ("net sales price") that is recognized as revenue for product sales includes the selling price to the Customer and an estimate of variable consideration. Components of variable consideration include prompt pay and other discounts, product returns, government rebates, third-party payor rebates, coverage gap rebates, incentives such as patient co-pay assistance, and other fees paid to Customers and other third-party payors where a distinct good or service is not received. Variable consideration is recorded on the condensed consolidated balance sheet as either a reduction of accounts receivable, if payable to a Customer, or as a current liability, if payable to a third-party other than a Customer. The Company considers all relevant information when estimating variable consideration such as assessment of its current and anticipated sales and demand forecasts, actual payment history, information from third parties regarding the payor mix for products, information from third parties regarding the units remaining in the distribution channel, specific known market events and trends, industry data and current contractual and statutory requirements that are reasonably available. The Company includes estimated amounts for variable consideration in the net sales price to the extent it is determined probable that a significant reversal of cumulative revenue recognized will not occur when the uncertainty associated with the variable consideration is resolved.

(Unaudited)

(In thousands, except share and per share amounts)

Payment terms with Customers do not exceed one year and, therefore, the Company does not account for a significant financing component in its arrangements. The Company expenses incremental cost of obtaining a contract with a Customer when incurred as the period of benefit is generally less than one year.

Reserves for Variable Consideration:

Trade Discounts and Allowances

The Company provides its Customers with certain trade discounts and allowances including discounts for prompt payments and other discounts and fees paid for distribution, data and administrative services. These discounts and fees are based on contractually-determined percentages and are recorded as a reduction of revenue and accounts receivable in the period in which the related product revenue is recognized.

#### Chargebacks

Chargebacks for fees and discounts to providers represent the estimated obligations resulting from contractual commitments to sell products to qualified healthcare providers at prices lower than the list prices charged to Customers who directly purchase the product from the Company. Customers charge the Company for the difference between what they pay for the product and the ultimate selling price to the qualified healthcare providers. These components of variable consideration are established in the same period that the related revenue is recognized, resulting in a reduction of product revenue and accounts receivable. Reserves for chargebacks consist of credits the Company expects to issue for units that remain in the distribution channel at the end of each reporting period and that the Company expects will be sold to qualified healthcare providers, as well as chargebacks that Customers have claimed, but for which the Company has not yet issued a credit.

#### Product Returns

Consistent with industry practice, the Company has a product returns policy that provides Customers right of return for product purchased within a specified period prior to and subsequent to the product's expiration date. The Company estimates the amount of its products that may be returned and presents this amount as a reduction of revenue in the period the related product revenue is recognized, in addition to establishing a liability. The Company's estimates for product returns are based upon available industry data and its own sales information, including its visibility into the inventory remaining in the distribution channel as well as historical returns, which develop over time.

#### Commercial Payor and Medicare Part D Rebates

The Company contracts with certain third-party payors, primarily pharmacy benefit managers ("PBMs") and health plans ("Plans"), for the payment of rebates with respect to utilization of its product. These rebates are based on contractual percentages applied to the amount of product prescribed to patients who are covered by the PBMs or the Plans with which it contracts. The Company estimates the rebates for commercial and Medicare Part D payors based on the contractual discount percentage, the various payor mix for EYSUVIS and INVELTYS as well as future rebates that will be made for product that has been recognized as revenue but remains in the distribution channel at the end of each reporting period. The Company also estimates the number of patients in the prescription drug coverage gap for whom it will owe an additional liability under the Medicare Part D program. Such estimates are recorded in the same period the related revenue is recognized, resulting in a reduction of product revenue and the establishment of a current liability.

#### Government Rebates

The Company is subject to discount obligations under Medicaid and other government programs. For Medicaid, reserves are based on actual payment history, and estimates of future Medicaid beneficiary utilization applied to the Medicaid unit rebate formula established by the Centers for Medicaid and Medicare Services. The Company's liability for these rebates consists of estimates of claims for the current period and estimated future claims that will be made for

(Unaudited) (In thousands, except share and per share amounts)

product that has been recognized as revenue but remains in the distribution channel at the end of each reporting period. These reserves are recorded in the same period the related revenue is recognized, resulting in a reduction of product revenue and the establishment of a current liability.

#### Co-pay Assistance Programs

The Company offers co-pay assistance programs (the "co-pay programs"), which are intended to provide financial assistance to patients who may or may not be covered by commercial insurance or, with respect to INVELTYS, who opt out of Medicare Part D programs. The calculation of accruals for the co-pay programs is based on actual claims processed during the period as well as an estimate of the number and cost per claim that the Company expects to receive associated with product that has been recognized as revenue but remains in the distribution channel at the end of each reporting period. Allowances for estimated co-pay claims are recorded in the same period the related revenue is recognized, resulting in a reduction of product revenue and the establishment of a current liability.

The following tables summarize activity in each of the Company's product revenue provision and allowance categories for the three and six months ended June 30, 2021 and 2020:

	,	Trade Discounts,			
		Allowances and			Rebates and
		Chargebacks (1)	Pr	oduct Returns (2)	Incentives (3)
Balance as of December 31, 2020	\$	1,157	\$	600	\$ 4,904
Provision related to current period sales		2,201		245	10,216
Changes in estimate related to prior period sales		3		30	(26)
Credit/payments made		(1,823)		(419)	(7,880)
Balance as of March 31, 2021	\$	1,538	\$	456	\$ 7,214
Provision related to current period sales		2,103		356	11,193
Changes in estimate related to prior period sales		30		142	(353)
Credit/payments made		(2,260)		(253)	(9,801)
Balance as of June 30, 2021	\$	1,411	\$	701	\$ 8,253

	IIau	e Discounts,			
	Allo	wances and			Rebates and
	Cha	rgebacks (1)	Produc	t Returns (2)	Incentives (3)
Balance as of December 31, 2019	\$	1,783	\$	180	\$ 10,044
Provision related to current period sales		725		_	4,576
Changes in estimate related to prior period sales		2		(66)	93
Credit/payments made		(1,114)		_	(8,090)
Balance as of March 31, 2020	\$	1,396	\$	114	\$ 6,623
Provision related to current period sales		335		_	2,194
Changes in estimate related to prior period sales		3		(51)	(234)
Credit/payments made		(1,336)		1	(3,013)
Balance as of June 30, 2020	\$	398	\$	64	\$ 5,570

Trade Discounts

- (1) Trade discounts, allowances and chargebacks include fees for distribution service fees, prompt pay and other discounts, and chargebacks. Estimated trade discounts, allowances and chargebacks are deducted from gross revenue at the time revenues are recognized and are recorded as a reduction to accounts receivable on the Company's condensed consolidated balance sheets.
- (2) Estimated provisions for product returns are deducted from gross revenues at the time revenues are recognized and are included in accrued expenses and other current liabilities on the Company's condensed consolidated balance sheets.
- (3) Rebates and incentives include managed care rebates, government rebates, co-pay program incentives, and sales incentives and allowances. Estimated provisions for rebates and discounts are deducted from gross revenues at the

(Unaudited)

(In thousands, except share and per share amounts)

time revenues are recognized and are included in accrued expenses and other current liabilities on the Company's condensed consolidated balance sheets.

#### Accounts Receivable, net

Accounts receivable are reported on the condensed consolidated balance sheets at outstanding amounts due from Customers for product sales. The Company deducts sales discounts for prompt payments and other discounts, contractual fees for service arrangements and chargebacks from accounts receivable. The Company evaluates the collectability of accounts receivable on a regular basis, by reviewing the financial condition and payment history of Customers, an overall review of collections experience on other accounts, and economic factors or events expected to affect future collections experience. An allowance for doubtful accounts is recorded when a receivable is deemed to be uncollectible.

The Company recorded no allowance for doubtful accounts as of June 30, 2021 or December 31, 2020. The Company recorded an allowance of \$1,411 and \$1,157 for expected sales discounts, related to prompt pay discounts and other discounts, contractual fee for service arrangements and chargebacks, to wholesalers and distributors as of June 30, 2021 and December 31, 2020, respectively.

#### 6. INVENTORY

Inventory consists of the following:

		June 30,	December 31,		
	_	2021	2020		
Raw materials	\$	1,485	\$	801	
Work in progress		9,580		6,437	
Finished goods		6,138		4,210	
Total inventory	\$	17,203	\$	11,448	

As of June 30, 2021, the Company had \$8,046 of current inventory and \$9,157 of long-term inventory. As of December 31, 2020, the Company had \$5,229 of current inventory and \$6,219 of long-term inventory.

#### 7. ACCRUED EXPENSES

Accrued expenses consist of the following:

		June 30, 2021		cember 31,
				2020
A	σħ	0.261	ф	F 22.4
Accrued revenue reserves (1)	\$	8,261	\$	5,224
Compensation and benefits		8,035		9,676
Commercial costs		2,121		2,103
Contract manufacturing		773		336
Professional services		605		926
Interest		533		_
Development costs		377		154
Other		550		552
Accrued expenses	\$	21,255	\$	18,971

<sup>(1)</sup> There were additional revenue reserves included in accounts payable of \$693 and \$280, as of June 30, 2021 and December 31, 2020, respectively.

(Unaudited)

(In thousands, except share and per share amounts)

#### 8. LEASES

#### Operating leases

In connection with the lease of the Company's corporate headquarters (the "Watertown Lease"), the Company issued a letter of credit to the landlord for \$2,042. The Company secured the letter of credit for the full amount of the letter with cash on deposit, which is reported as restricted cash on the condensed consolidated balance sheets as of June 30, 2021 and December 31, 2020.

For the three and six months ended June 30, 2021, the variable lease expense for the Watertown Lease, which includes common area maintenance and real estate taxes, was \$334 and \$697, respectively. For the three and six months ended June 30, 2020, the variable lease expense for the Watertown Lease, which includes common area maintenance and real estate taxes, was \$611 and \$1,068, respectively. The remaining lease term was 10.3 years as of June 30, 2021.

#### Vehicle Fleet lease

During the year ended December 31, 2019, the Company entered into a master fleet lease agreement (the "Vehicle Fleet Lease"), pursuant to which it leased 65 vehicles. In connection with the Vehicle Fleet Lease, the Company issued a letter of credit for \$450, which was reported as restricted cash on the condensed consolidated balance sheets as of June 30, 2021 and December 31, 2020. The lease has an expected term of three years, which commenced upon the delivery of the vehicles in March 2019.

During the six months ended June 30, 2021, the Company modified the Vehicle Fleet Lease to add 54 additional vehicles to the fleet. The new component of the lease has an expected term of approximately three years, which commenced upon the delivery of the additional vehicles in March 2021.

The components of lease expense and related cash flows were as follows:

	Three Months Ended			Six Months Ended June 30,				
		June 30, 2021 2020			2021			2020
Lease cost								
Operating lease cost	\$	1,268	\$	1,185	\$	2,484	\$	2,371
Variable lease cost		584		658		1,093		1,068
Total lease cost	\$	1,852	\$	1,843	\$	3,577	\$	3,439
	_							
Operating cash outflows from operating leases	\$	1,659	\$	1,686	\$	3,243	\$	3,130

The weighted average remaining lease term and weighted average discount rate of operating leases are as follows:

	June 30,	December 31,
	2021	2020
Weighted average remaining lease term	9.9 years	10.3 years
Weighted average discount rate	9.6%	9.8%

(Unaudited)

(In thousands, except share and per share amounts)

#### 9. DEBT

Athyrium Credit Facility

On October 1, 2018, the Company entered into a credit agreement (the "Athyrium Credit Facility") with Athyrium Opportunities III Acquisition LP ("Athyrium") for up to \$110,000. The Athyrium Credit Facility provided for a Term Loan A in the aggregate principal amount of \$75,000 (the "Term Loan A"), and a Term Loan B in the aggregate principal amount of \$35,000 (the "Term Loan B"). On October 1, 2018, the Company borrowed the entire principal amount of the Term Loan A. The maturity date of the Athyrium Credit Facility was October 1, 2024, the six-year anniversary of the close.

As of December 31, 2020, the unpaid principal balance under the Athyrium Credit Facility was \$75,000 and the unamortized debt discount was \$3,088. On May 4, 2021, the Company repaid all amounts owed under the Athyrium Credit Facility and terminated all commitments by Athyrium to extend further credit thereunder and all guarantees and security interests granted by the Company to the lenders thereunder. In connection with the termination of the Athyrium Credit Facility, the Company paid to the lenders a prepayment premium of \$2,250 and an exit fee of \$750. The transaction resulted in a loss on extinguishment of debt of \$5,395, consisting of the prepayment premium, the unamortized debt discount and issuance costs and the unaccreted exit fee.

#### Loan and Security Agreement

On May 4, 2021 (the "Closing Date"), the Company entered into a Loan and Security Agreement (the "Loan Agreement") with Oxford Finance LLC, in its capacity as lender (in such capacity, the "Lender"), and in its capacity as collateral agent (in such capacity, the "Agent"), pursuant to which a term loan of up to an aggregate principal amount of \$125,000 is available to the Company, consisting of (i) a tranche A term loan that was disbursed on the Closing Date in the aggregate principal amount of \$80,000; (ii) a contingent tranche B term loan in the aggregate principal amount of \$20,000 available to the Company through June 30, 2023 and within 90 days of the Company achieving trailing 6-month product revenue equal to or greater than \$75,000, subject to certain other terms and conditions; and (iii) a contingent tranche C term loan in the aggregate principal amount of \$25,000 available to the Company through December 31, 2023 and within 90 days of the Company achieving trailing 6-month product revenue equal to or greater than \$100,000, subject to certain other terms and conditions. The Company utilized substantially all of the proceeds from the tranche A term loan to repay the Athyrium Credit Facility (as more fully described above).

The term loans bear interest at a floating rate equal to the greater of (i) 30-day LIBOR and (ii) 0.11%, plus 7.89%. The Loan Agreement provides for interest-only payments until December 1, 2024 if neither the tranche B term loan nor the tranche C term loan are made, and until June 1, 2025 if either the tranche B term loan or the tranche C term loan is made (the "Amortization Date"). The aggregate outstanding principal balance of the term loans are required to be repaid in monthly installments starting on the Amortization Date based on a repayment schedule equal to (i) 18 months if neither the tranche B term loan nor the tranche C term loan is made and (ii) 12 months if either the tranche B term loan or the tranche C term loan is made. All unpaid principal and accrued and unpaid interest with respect to each term loan is due and payable in full on May 1, 2026 (the "Maturity Date").

The Company paid a facility fee of \$400 on the Closing Date and has agreed to pay a facility fee of \$100 upon closing of the tranche B term loan and a \$125 facility fee upon the closing of the tranche C term loan. The Company will be required to make a final payment fee of 7.00% of the original principal amount of any funded term loan payable on the earlier of (i) the prepayment of the term loan in full or (ii) the Maturity Date. At the Company's option, the Company may elect to prepay all, but not less than all, of the outstanding loans, subject to a prepayment fee equal to the following percentage of the principal amount being prepaid: 3.00% if an advance is prepaid during the first 12 months following the applicable advance date, 2.00% if an advance is prepaid after 12 months but prior to 24 months following the applicable advance date, and 1.00% if an advance is prepaid any time after 24 months following the applicable advance date but prior to the Maturity Date.

(Unaudited)

(In thousands, except share and per share amounts)

In connection with its entry into the Loan Agreement, the Company granted the Agent a security interest in substantially all of the Company's personal property owned or later acquired, including intellectual property. The Loan Agreement also contains customary representations and warranties and affirmative and negative covenants, as well as customary events of default. Certain of the customary negative covenants limit the ability of the Company and certain of its subsidiaries, among other things, to incur future debt, grant liens, make investments, make acquisitions, distribute dividends, make certain restricted payments and sell assets, subject in each case to certain exceptions.

The Loan Agreement includes features requiring (i) additional interest rate upon an event of default accrued at an additional 5%, and (ii) the Lender's right to declare all outstanding principal and interest immediately payable upon an event of default. These two features were analyzed and determined to be embedded derivatives to be valued as separate financial instruments. These embedded derivatives were bundled and valued as one compound derivative in accordance with the applicable accounting guidance for derivatives and hedging transactions. The Company determined that, due to the unlikely event of default, the embedded derivatives have a *de minimis* value as of June 30, 2021. The derivative liability will be remeasured at fair value at each reporting date, with changes in fair value being recorded as other income (expense) in the condensed consolidated statements of operations and comprehensive loss.

In addition, in connection with the Loan Agreement, the Company paid certain fees to the Lender and other third-party service providers. The fees paid to the Lender were recorded as a debt discount while the fees paid to other third-party service providers were recorded as debt issuance cost. These costs were amortized using the effective interest method over the term of the Loan Agreement. The amortization of debt discount and debt issuance cost is included in interest expense within the condensed consolidated statements of operations and comprehensive loss. As of June 30, 2021, the effective interest rate was 10.41%, which takes into consideration the non-cash accretion of the exit fee and the amortization of the debt discount and issuance costs.

During the three months ended June 30, 2021 and June 30, 2020, the Company recognized interest expense of \$1,860 and \$2,095, respectively, for the Loan Agreement and Athyrium Credit Facility. This consisted of amortization of debt discount of \$150 and \$223 for the periods ended June 30, 2021 and June 30, 2020, respectively, and the contractual coupon interest expense of \$1,710 and \$1,872 for the periods ended June 30, 2021 and June 30, 2020, respectively. During the six months ended June 30, 2021 and June 30, 2020, the Company recognized interest expense of \$3,952 and \$4,183, respectively. This consisted of amortization of debt discount of \$390 and \$439 for the periods ended June 30, 2021 and June 30, 2020, respectively, and the contractual coupon interest expense of \$3,562 and \$3,744 for the periods ended June 30, 2021 and June 30, 2020, respectively.

(Unaudited)

(In thousands, except share and per share amounts)

The components of the carrying value of the debt as of June 30, 2021 and December 31, 2020 are detailed below:

	June 30, 2021	D	ecember 31, 2020
Principal loan balance	\$ 80,000	\$	75,000
Unamortized debt discount and issuance cost	(2,150)		(3,088)
Cumulative accretion of exit fee	205		331
Long-term debt, net	\$ 78,055	\$	72,243

The annual principal payments due under the Loan Agreement as of June 30, 2021 were as follows:

Vanre	Ending	December	21
rears	F.naiin9	December	.31.

2021 (remaining six months)	\$ _
2022	_
2023	_
2024	4,445
2025	53,333
Thereafter	 22,222
Total	\$ 80,000

#### 10. WARRANTS

The following table summarizes the common stock warrants outstanding as of June 30, 2021 and December 31, 2020, each exercisable into the number of shares of common stock set forth below as of the specified dates:

				Shares E	xercisable at
Issued	Exercise Price	Expiration Date	Exercisable From	June 30, 2021	December 31, 2020
2013	\$ 7.50	April 2021	July 2017	_	33,333
2014	\$ 7.50	November 2024	July 2017	16,000	16,000
2016	\$ 8.27	October 2026	September 2017	14,512	14,512
2018	\$ 12.18	October 2025	October 2018	184,660	184,660
				215,172	248,505

#### 11. EQUITY FINANCINGS

On August 9, 2018, the Company filed a shelf registration statement on Form S-3 with the SEC, which was declared effective on August 27, 2018 (the "2018 Shelf Registration"). Under the 2018 Shelf Registration, the Company could initially offer and sell up to \$250,000 of a variety of securities including common stock, preferred stock, warrants, depositary shares, debt securities, purchase contracts, purchase units or any combination of such securities during the three-year period that commenced upon the 2018 Shelf Registration becoming effective.

In connection with the filing of the 2018 Shelf Registration, the Company entered into a sales agreement (the "2018 Sales Agreement") with Jefferies, LLC ("Jefferies") pursuant to which the Company may issue and sell, from time to time, up to an aggregate of \$50,000 of its common stock in an ATM Offering through Jefferies, as sales agent. Through the first quarter of 2020, the Company issued an aggregate of 4,945,605 shares of its common stock under the ATM Offering, resulting in net proceeds to the Company of \$25,605. On March 10, 2020, the Company notified Jefferies that it was suspending and terminating the prospectus related to the 2018 Sales Agreement.

On March 11, 2020, the Company sold 16,000,000 shares of its common stock (the "2020 Offering Shares") in an underwritten offering (the "2020 Offering"), pursuant to the 2018 Shelf Registration, at a public offering price of \$7.89 per share, resulting in net proceeds of \$118,207, after underwriting discounts, commissions, and offering

(Unaudited)

(In thousands, except share and per share amounts)

expenses. In addition, the underwriters of the 2020 Offering were granted the option for a period of 30 days to purchase up to an additional 2,400,000 shares of common stock offered in the public offering at the public offering price, less underwriting discounts, commissions and offering expenses. On April 3, 2020, the underwriters exercised their option and purchased an additional 979,371 shares of common stock at \$7.89 per share, resulting in net proceeds to the Company of \$7,216, after underwriting discounts, commissions, and offering expenses. The total number of shares sold by the Company in the 2020 Offering was 16,979,371, resulting in total net proceeds to the Company, after underwriting discounts, commissions, and offering expenses, of \$125,423. Under the 2018 Shelf Registration, the Company has issued an aggregate of 30,549,976 shares of common stock, including under the ATM Offering, resulting in aggregate gross proceeds of \$231,666. There was \$18,334 of securities available to be issued under the 2018 Shelf Registration as of June 30, 2021.

On May 7, 2020, the Company filed a shelf registration statement on Form S-3 with the SEC, which was declared effective on May 19, 2020 (the "2020 Shelf Registration"). Under the 2020 Shelf Registration, the Company may offer and sell up to \$350,000 of a variety of securities including common stock, preferred stock, warrants, depositary shares, debt securities or units during the three-year period that commenced upon the 2020 Shelf Registration becoming effective. In connection with the filing of the 2020 Shelf Registration, the Company entered into an amended and restated sales agreement with Jefferies pursuant to which it may issue and sell, from time to time, up to an aggregate of \$75,000 of its common stock under its ATM Offering through Jefferies, as a sales agent. During the fourth quarter of 2020, the Company issued an aggregate of 2,821,059 shares of its common stock under the ATM Offering, resulting in net proceeds of \$20,612. During the three and six months ended June 30, 2021, the Company issued and sold an additional 837,257 and 5,583,329 shares of its common stock under its ATM Offering, respectively, resulting in net proceeds of \$6,021 and \$40,730, respectively. As of June 30, 2021, there was \$11,859 of shares of common stock remaining under the ATM Offering and, excluding the shares of common stock that may be offered under the ATM Offering, there was approximately \$275,000 of securities available to be issued under the 2020 Shelf Registration.

#### 12. STOCK-BASED COMPENSATION

During the six months ended June 30, 2021, the Company granted options for the purchase of 1,604,964 shares of common stock including 148,700 non-statutory stock options granted to new employees and 556,333 restricted stock units. In January 2021, employees of the Company purchased an aggregate of 74,847 shares under the ESPP. In July 2021, employees of the Company purchased an aggregate of 200,877 shares under the ESPP.

The assumptions used in determining fair value of the stock options granted during the six months ended June 30, 2021 are as follows:

	Six Months Ended June 30, 2021
Expected volatility	72.9% - 74.2%
Risk-free interest rate	0.50% - 1.07%
Expected dividend yield	0%
Expected term (in years)	6.03 - 6.10

During the six months ended June 30, 2021 the weighted average grant-date fair value of options granted was \$4.37.

In June 2020, the Company issued 693,537 PSUs to certain executives and other employees tied to certain performance criteria, which will vest, if at all, as to 50% on the first anniversary of satisfying the performance criteria and the remaining 50% vesting upon the second anniversary of satisfying the performance criteria. The Company has determined that the performance criteria for these awards has been achieved but not all of the awards have vested as of June 30, 2021. As of June 30, 2021, a total of 1,370,830 RSUs and PSUs were outstanding, consisting of 1,282,830 unvested shares and 88,000 vested and deferred shares.

(Unaudited)

(In thousands, except share and per share amounts)

Stock-based compensation expense was classified in the condensed consolidated statements of operations and comprehensive loss as follows for the three and six months ended June 30, 2021 and 2020:

	Three Months Ended June 30,			Six Months Ende June 30,				
	202	2021 2020		2020 2021		2020		
Cost of product revenues	\$	37	\$	9	\$	71	\$	28
Research and development	ç	986		551		1,952		1,274
Selling, general and administrative	3,6	687		1,933		7,389		3,686
Total	\$ 4,7	710	\$	2,493	\$	9,412	\$	4,988

#### 13. INCOME TAXES

The Company did not record a provision or benefit for income taxes during the three and six months ended June 30, 2021 and 2020. The Company continues to maintain a full valuation allowance for its U.S. federal and state deferred tax assets.

The Company has evaluated the positive and negative evidence bearing upon its ability to realize the deferred tax assets. Management has considered the Company's history of cumulative net losses incurred since inception and its generation of limited revenue from product sales since inception and has concluded that it is more likely than not that the Company will not realize the benefits of the deferred tax assets. Management reevaluates the positive and negative evidence at each reporting period.

Realization of the future tax benefits is dependent on many factors, including the Company's ability to generate taxable income within the net operating loss carryforward period. Under the provisions of Section 382 of the Internal Revenue Code of 1986, as amended, certain substantial changes in the Company's ownership, including a sale of the Company, or significant changes in ownership due to sales of equity, may have limited, or may limit in the future, the amount of net operating loss carryforwards, which could be used annually to offset future taxable income. The Company previously completed an analysis and determined that an ownership change has materially limited the net operating loss carryforwards and research and development tax credits available to offset future tax liabilities. The Company may be further limited by any changes that may have occurred or may occur subsequent to December 31, 2020.

The Company files its corporate income tax returns in the United States and various states. All tax years since the date of incorporation remain open to examination by the major taxing jurisdictions (state and federal) to which the Company is subject, as carryforward attributes generated in years past may still be adjusted upon examination by the Internal Revenue Service ("IRS") or other authorities if they have or will be used in a future period. The Company is not currently under examination by the IRS or any other jurisdictions for any tax year.

As of June 30, 2021 and December 31, 2020, the Company had no uncertain tax positions. The Company's policy is to recognize interest and penalties related to income tax matters as a component of income tax expense, of which no interest or penalties were recorded for the three and six months ended June 30, 2021 and 2020.

(Unaudited)

(In thousands, except share and per share amounts)

#### 14. COMMITMENTS AND CONTINGENCIES

**License Agreement**—In 2009, the Company entered into an exclusive license agreement with The Johns Hopkins University ("JHU"), as amended in November 2012, May 2014, August 2014, October 2014, June 2018 and July 2020, which licensed to the Company a portfolio of specified patent rights and remains in full force and effect. Pursuant to the terms of the agreement, as amended, the Company agreed to pay an initial license fee, minimum annual payments beginning in 2017, certain development and commercial milestone payments, royalties on product sales and reimburse all or a portion of the costs associated with the preparation, filing, prosecution and maintenance of the agreed-upon patents and patent applications to JHU.

After 2016 and until the first commercial sale of product, which occurred in January 2019, the minimum annual payment was \$38. Upon the first commercial sale of INVELTYS, the annual minimum payment increased to \$113. The Company is obligated to pay JHU low single-digit running royalties based upon a percentage of net sales of the licensed products, which is applied to the annual minimum payment. The Company also has an obligation to pay JHU certain one-time development and commercial milestone payments. During the three months ended June 30, 2021 and 2020, the Company paid JHU \$33 and \$11, respectively, in royalty payments associated with the sales of EYSUVIS and INVELTYS. During the six months ended June 30, 2021 and 2020, the Company paid JHU \$55 and \$23, respectively, in royalty payments associated with the sales of EYSUVIS and INVELTYS. The Company paid JHU a \$150 milestone payment during the six months ended June 30, 2021, which was triggered by the first commercial sale of EYSUVIS in the United States in December 2020.

The Company recorded other expenses related to the JHU agreement of \$21 and \$31 for the three months ended June 30, 2021 and 2020, respectively. The Company recorded a credit for other expenses related to the JHU agreement of \$30 and other expenses of \$76 for the six months ended June 30, 2021 and 2020, respectively.

**Litigation**—The Company is not currently subject to any material legal proceedings.

**Other Commitments** — The Company entered into a commercial supply agreement with Woodstock Sterile Solutions, Inc. (formerly known as Catalent Pharma Solutions, LLC) to manufacture commercial supplies of EYSUVIS and INVELTYS.

The Company has the following minimum purchase obligations for EYSUVIS and INVELTYS as of June 30, 2021:

rear's Entaing December 51,	
2021 (remaining six months)	\$ 2,295
2022	5,390
2023	6,285
2024	7,875
2025	8,199
Thereafter	17,925
Total minimum purchase commitments	\$ 47,969

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

The following discussion and analysis of our financial condition and results of operations should be read together with our unaudited condensed consolidated financial statements and related notes thereto appearing elsewhere in this Quarterly Report on Form 10-Q and our Annual Report on Form 10-K for the year ended December 31, 2020, which was filed with the Securities and Exchange Commission on February 25, 2021.

This Quarterly Report on Form 10-Q contains forward-looking statements that involve substantial risks and uncertainties. The words "anticipate," "believe," "continue" "could," "estimate," "expect," "intend," "may," "might," "plan," "potential," "predict," "project," "should," "target," "would," and similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain these identifying words. There are a number of important risks and uncertainties that could cause our actual results to differ materially from those indicated by forward-looking statements. We may not actually achieve the plans, intentions or expectations disclosed in our forward-looking statements, and you should not place undue reliance on our forward-looking statements. Actual results or events could differ materially from the plans, intentions and expectations disclosed in the forward-looking statements we make. We have included important factors in the cautionary statements included in this Quarterly Report on Form 10-Q, particularly in the section entitled "Risk Factors" in Part II, Item 1A that we believe could cause actual results or events to differ materially from the forward-looking statements that we make. Our forward-looking statements do not reflect the potential impact of any future acquisitions, mergers, dispositions, joint ventures or investments that we may make.

We are a biopharmaceutical company focused on the discovery, development and commercialization of innovative therapies for diseases of the eye. We have worldwide rights to a portfolio of innovative products and product candidates that include two marketed therapies utilizing our proprietary mucus-penetrating particle, or MPP, drug delivery technology, which we refer to as our AMPPLIFY® technology, to address medical needs for the front of the eye, and a pipeline of proprietary new chemical entities, or NCEs, targeted to address front and back of the eye diseases.

Our two marketed products are EYSUVIS® (loteprednol etabonate ophthalmic suspension) 0.25%, for the short-term (up to two weeks) treatment of the signs and symptoms of dry eye disease, and INVELTYS® (loteprednol etabonate ophthalmic suspension) 1%, a topical twice-a-day ocular steroid for the treatment of post-operative inflammation and pain following ocular surgery. Both products apply our AMPPLIFY technology to loteprednol etabonate, or LE, a corticosteroid designed for ocular applications. The AMPPLIFY technology, uses selectively-sized nanoparticles that each have a proprietary coating. We believe that these two key attributes enable even distribution of drug particles on mucosal surfaces and significantly increase drug delivery to target tissues by enhancing mobility of drug particles through mucus and preventing drug particles from becoming trapped and eliminated by mucus. We commercially launched INVELTYS in January 2019 and began shipping EYSUVIS to wholesalers in the U.S. in late December 2020. Full promotional launch of EYSUVIS began in January 2021.

We are also progressing our pipeline of proprietary preclinical development programs targeted to address front and back of the eye diseases. These preclinical development programs, all of which are new chemical entities, include our receptor Tyrosine Kinase Inhibitor program, that is designed to inhibit the vascular endothelial growth factor pathway, for the treatment of retinal diseases, including wet age-related macular degeneration; our selective glucocorticoid receptor modulators, which are a novel class of therapies designed to modify the downstream activity of the receptors to exhibit the anti-inflammatory and immunomodulatory properties of the corticosteroid class of therapies without their associated side effects; and our novel surface targeting steroid designed to target the ocular surface and thus have the potential to have fewer side effects compared to traditional topical steroids. We own all intellectual property and worldwide rights to these pipeline preclinical development programs.

We have retained worldwide commercial rights for EYSUVIS, INVELTYS and our preclinical development programs. Starting with FDA approval of INVELTYS, we have built a commercial infrastructure with our own focused, specialty sales force which now includes 105 territory sales managers, or TSMs, 14 regional sales leaders, two area sales leaders and four directors of national accounts. We plan to further expand our sales force up to 125 TSMs by year-end, subject to continued growth in payor coverage and the status of the COVID-19 pandemic. Our sales representatives promote both EYSUVIS and INVELTYS. We expect to commercialize in the United States any of our product

candidates that receive marketing approval as well. We also expect to explore commercialization of EYSUVIS for the treatment of dry eye disease in certain markets outside the United States, including the European Union, or EU, utilizing a variety of collaboration, distribution and other marketing arrangements with one or more third parties.

Since inception, we have incurred significant losses from operations and negative cash flows from operations. Our net losses were \$36.5 million and \$66.9 million for the three and six months ended June 30, 2021, respectively, and \$104.3 million for the year ended December 31, 2020. As of June 30, 2021, we had an accumulated deficit of \$466.7 million. As we commenced a full promotional launch of EYSUVIS in early January 2021 and commercially launched our first product, INVELTYS, in January 2019, we have had only limited revenues to date from product sales and have financed our operations primarily through proceeds from our initial public offering, or IPO, follow-on public common stock offerings and sales of our common stock under our sales agreement with Jefferies, LLC, or Jefferies, in at-the-market offerings, or ATM Offerings, private placements of preferred stock, borrowings under credit facilities and the Loan and Security Agreement, or Loan Agreement, with Oxford Finance LLC, or Oxford Finance, convertible promissory notes and warrants. We have devoted substantially all of our financial resources and efforts to research and development, including preclinical studies and clinical trials and engaging in activities to commercialize EYSUVIS and INVELTYS. Although we expect to continue to generate revenue from sales of EYSUVIS and INVELTYS, there can be no assurance as to the amount or timing of any such revenue, and we expect to continue to incur significant expenses and operating losses. Our net losses may fluctuate significantly from quarter-to-quarter and year-to-year.

#### **Business Impact of COVID-19 Pandemic**

In order to safeguard the health of our employees from the COVID-19 pandemic, we follow, and will continue to follow, recommendations from the U.S. Centers for Disease Control and Prevention, as well as federal, state, and local governments, regarding working-from-home practices for non-essential employees. As a result, all office-based personnel are permitted to work from home, and our laboratory facilities that support our early-stage research activities are being utilized as necessary. In addition, we previously suspended our sales force from substantially all in-person interactions with physicians and were limited to conducting educational and promotional activities virtually. Our sales force has since resumed substantially all in-person interactions in the field, but to the extent we restrict, or are restricted from, in-person interactions with physicians in the future, we may be limited to conducting educational and promotional activities virtually, which may continue to hamper our ability to market and commercialize EYSUVIS and INVELTYS.

In addition, government restrictions have at times led to moratoria on elective ocular surgeries in many jurisdictions, which had significantly reduced, and may in the future continue to significantly reduce, the demand for INVELTYS, which is indicated for the treatment of inflammation and pain following ocular surgery. The extent of the impact of the COVID-19 pandemic on our commercialization efforts of EYSUVIS and INVELTYS and our operational and financial performance will depend on certain developments, including the length and severity of this pandemic, the timing and extent of any resurgence of the COVID-19 virus or any variant strains of the virus, the availability and effectiveness of vaccines, and the full extent of the impact on our customers, employees and vendors and government agencies, all of which are uncertain and cannot be predicted.

Management is actively monitoring the COVID-19 pandemic and its effects on our financial condition, liquidity, operations, customers, sales force, contractors, and workforce. For additional information on risks posed by the COVID-19 pandemic, please see Part II, Item 1A – "Risk Factors" of this Quarterly Report on Form 10-Q, including the risk factor entitled "The ongoing novel coronavirus pandemic and the efforts to prevent its spread have adversely impacted our operations and the market for INVELTYS, could impact the launch and commercialization of EYSUVIS and may continue to adversely affect our business, results of operations and financial condition."

#### **Financial Operations Overview**

#### Product Revenues, Net

We commenced generating product revenues from sales of INVELTYS in January 2019 and commenced generating revenue from EYSUVIS upon the shipment to wholesalers in the United States in late December 2020. Full promotional launch of EYSUVIS began in early January 2021. Our product revenues are recorded net of provisions relating to estimates for (i) trade discounts and allowances, such as discounts for prompt payment and other discounts and distributor fees, (ii) estimated rebates, chargebacks and co-pay assistance programs, and (iii) reserves for expected product returns. These estimates reflect current contractual and statutory requirements, known market events and trends, industry data and forecasted customer buying and payment patterns. Actual amounts may ultimately differ from these estimates. If actual results vary, estimates may be adjusted in the period such change in estimate becomes known, which could have an impact on earnings in the period of adjustment. Beginning in March 2020 and continuing through most of the second quarter of 2020, prescriptions of INVELTYS and revenue had been adversely affected by the ongoing COVID-19 pandemic as federal, state and local governments implemented restrictions on elective procedures, which included most ocular surgeries. While many deferred ocular surgeries have been rescheduled as individual states have released restrictions on elective procedures, and INVELTYS prescriptions have returned to quarterly growth, we are unable to project the specific timing or potential impact on future revenues given the continued uncertainty around the impact and duration of the restrictions related to the COVID-19 pandemic. We also cannot project the full extent of the impact that the COVID-19 pandemic may have on the full promotional launch and commercialization of EYSUVIS.

#### **Cost of Product Revenues**

Cost of product revenues consists primarily of materials, third-party manufacturing costs, freight and distribution costs, royalty expense, allocation of labor, quality control and assurance, reserves for defective inventory, reserves for excess and obsolete inventory and losses on inventory purchase commitments and other manufacturing overhead costs. We expensed cost of product revenues related to INVELTYS as research and development expenses prior to U.S. regulatory approval, which we received on August 22, 2018. We expensed cost of product revenues related to EYSUVIS as research and development expenses prior to the determination that FDA approval was probable and before the future economic benefit was expected to be realized. With respect to the ongoing COVID-19 pandemic, we expect that the cost of product revenues will be impacted consistent with the negative impact to product revenues, net. However, we are unable to predict the specific timing or specific impact on cost of product revenues given the continued uncertainty around the impact and duration of the restrictions related to the COVID-19 pandemic.

#### Selling, General and Administrative Expenses

Selling, general and administrative expenses consist primarily of salaries, benefits, commissions, stock-based compensation and travel expenses related to our commercial infrastructure and our executive, finance, human resources, legal, information technology and business development functions. Selling, general and administrative expenses also include external selling and marketing costs, costs to manufacture sample units and professional fees for auditing, tax, information technology, consultants, legal services and allocated facility-related costs not otherwise included in research and development expenses.

We anticipate that our selling, general and administrative expenses will increase in the future as we continue to build our commercial infrastructure to support the full promotional launch and commercialization of EYSUVIS and the continued commercialization of INVELTYS or of any product candidates for which we obtain marketing approval. We also anticipate that our selling, general and administrative expenses will increase if and as we increase our administrative headcount to support our continued research activities and development of our product candidates. Our sales force has resumed substantially all in-person interactions in the field with physicians, which were previously suspended due to the restrictions with respect to the ongoing COVID-19 pandemic. If we are forced to suspend all or some in-person sales force interactions again in the future as a result of the COVID-19 pandemic, selling, general and administrative expenses could be favorably impacted by a reduction in certain expenses associated with the restriction in activities for our sales force and other employees. We are unable to predict the specific amount of this impact if we are forced to resume such restrictions.

#### Research and Development Expenses

Research and development expenses consist of costs associated with our research activities, including compensation and benefits for full-time research and development employees, an allocation of facilities expenses, overhead expenses and certain outside expenses. Our research and development expenses include:

- employee-related expenses, including salaries, related benefits, travel and stock-based compensation;
- expenses incurred for the preclinical and clinical development of our product candidates and under
  agreements with contract research organizations, or CROs, including costs of manufacturing product
  candidates prior to the determination that FDA approval of a drug candidate is probable and before the future
  economic benefit of the drug is expected to be realized; and
- facilities, depreciation and other expenses, which include direct and allocated expenses for rent and maintenance of facilities and supplies.

We expense research and development costs as they are incurred. We expense costs relating to the production of inventory for our product candidates, as research and development expenses within our condensed consolidated statements of operations and comprehensive loss in the period incurred, unless we believe regulatory approval and subsequent commercialization of the product candidate is probable and we expect the future economic benefit from sales of the drug to be realized. Research and development costs that are paid in advance of performance are capitalized as a prepaid expense until incurred. We track outsourced development costs by development program but do not allocate personnel costs, payments made under our license agreements or other costs to specific product candidates or development programs. These costs are included in Employee-related costs and Other research and development costs in the line items in the tables under "Results of Operations".

We expect that our total research and development costs will decrease in 2021 as compared to the year ended December 31, 2020 as a result of the completion of our Phase 3 clinical trial of EYSUVIS, which we refer to as STRIDE 3 (STRIDE- Short Term Relief In Dry Eye), and as a result of the capitalization of EYSUVIS manufacturing costs as inventory beginning in the third quarter of 2020. We expect that research and development costs will increase if and as we continue to advance our preclinical development programs, identify product candidates and conduct preclinical studies and clinical trials. The process of conducting preclinical studies and clinical trials necessary to obtain regulatory approval is costly and time-consuming. We may never succeed in obtaining marketing approval for any of our product candidates. The probability of success for each product candidate may be affected by numerous factors, including preclinical data, clinical data, competition, manufacturing capability and commercial viability. With respect to the ongoing COVID-19 pandemic, we may incur reduced research and development costs resulting from any limitations that may be placed on our laboratory facilities that support our early-stage research. However, we are unable to predict the specific amount of this impact, nor are we able to predict the additional costs, if any, associated with personnel safely resuming their full activities.

Our research and development programs are at the early stage of development. Successful development and completion of preclinical studies and clinical trials is uncertain and may not result in approved products. Completion dates and completion costs can vary significantly for each product candidate and future product candidate and are difficult to predict. We will continue to make determinations as to which product candidates to pursue and how much funding to direct to each product candidate on an ongoing basis in response to the scientific and clinical success of each product candidate as well as ongoing assessments as to the commercial potential of product candidates and our ability to enter into collaborations with respect to each product candidate. We may need to raise additional capital and may seek collaborations in the future to advance our various product candidates. Additional private or public financings may not be available to us on acceptable terms, or at all. Our failure to raise capital as and when needed would have a material adverse effect on our financial condition and our ability to pursue our business strategy.

#### Interest Income

Interest income consists of interest earned on our cash, cash equivalents and short-term investments.

#### Interest Expense

Interest expense primarily consists of contractual coupon interest, amortization of debt discounts and debt issuance costs and accretion of the final payment fee recognized on our debt facility.

#### Loss on Extinguishment of Debt

Loss on extinguishment of debt primarily consists of unamortized debt discount and issuance costs, a prepayment premium and unaccreted exit fees on the Athyrium Credit Facility.

#### **Critical Accounting Policies and Significant Judgments and Estimates**

Our management's discussion and analysis of our financial condition and results of operations is based on our financial statements, which we have prepared in accordance with U.S. generally accepted accounting principles. We believe that several accounting policies are important to understanding our historical and future performance. We refer to these policies as critical because these specific areas generally require us to make judgments and estimates about matters that are uncertain at the time we make the estimate, and different estimates—which also would have been reasonable—could have been used. On an ongoing basis, we evaluate our estimates and judgments, including those described in greater detail below. We base our estimates on historical experience and other market-specific or other relevant assumptions that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying value of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

There have been no material changes to our critical accounting policies from those described in "Management's Discussion and Analysis of Financial Condition and Results of Operations" in our Annual Report on Form 10-K for the fiscal year ended December 31, 2020.

#### **Results of Operations**

#### Comparison of the Three Months Ended June 30, 2021 and 2020

The following table summarizes the results of our operations for the three months ended June 30, 2021 and 2020:

		Three Mo				
	2021 2020			2020	20 Ch	
		(in tho				
Product revenues, net	\$	3,051	\$	833	\$	2,218
Costs and expenses:						
Cost of product revenues		1,016		759		257
Selling, general and administrative		27,986		15,301		12,685
Research and development		3,094		6,053		(2,959)
Total operating expenses		32,096		22,113		9,983
Loss from operations	(	29,045)		(21,280)		(7,765)
Other income (expense)						
Interest income		33		102		(69)
Interest expense		(2,091)		(2,134)		43
Loss on extinguishment of debt		(5,395)				(5,395)
Net loss	\$ (	36,498)	\$	(23,312)	\$	(13,186)

#### **Table of Contents**

#### Product revenues, net

Product revenues, net was \$3.1 million for the three months ended June 30, 2021, consisting of \$1.7 million from EYSUVIS sales and \$1.4 million from INVELTYS sales, compared to \$0.8 million from INVELTYS sales for the three months ended June 30, 2020. There were no sales of EYSUVIS in the three months ended June 30, 2020. The increase in product revenues, net of \$2.3 million was driven primarily by sales of EYSUVIS, which we began shipping to wholesalers in the United States in late December 2020, an increase in the total units of INVELTYS sold in the three months ended June 30, 2021 as well as a higher per unit gross selling price of INVELTYS as compared to those sold during the three months ended June 30, 2020. These increases in INVELTYS sales were partially offset by higher estimated reserves per unit during the three months ended June 30, 2021 as compared to those estimated during the three months ended June 30, 2020. We expect product revenues to increase if and as we increase our market share and obtain and maintain coverage and adequate reimbursement for EYSUVIS and INVELTYS from third-party payors; however, revenues could continue to be negatively impacted in 2021 as a result of the COVID-19 pandemic.

#### Cost of product revenues

Cost of product revenues was \$1.0 million for the three months ended June 30, 2021, compared to \$0.8 million for the three months ended June 30, 2020, an increase of \$0.2 million. Cost of product revenues increased \$0.5 million due to an increase in total INVELTYS units sold during the three months ended June 30, 2021, compared to the three months ended June 30, 2020 as well as a higher INVELTYS cost per unit as a result of the units sold during the three months ended June 30, 2020 being partially manufactured prior to FDA approval and for which costs were expensed as research and development prior to FDA approval as compared to those units sold during the three months ended June 30, 2021. Offsetting these increases was a reserve for excess inventory of \$0.5 million recorded during the three months ended June 30, 2020, which did not occur in the three months ended June 30, 2021. The cost of product revenues attributable to EYSUVIS was \$0.2 million for the three months ended June 30, 2021. There were no sales of EYSUVIS in the three months ended June 30, 2020. We expect aggregate cost of product revenues to increase as we continue to commercialize INVELTYS and as a result of the launch of EYSUVIS, which we began shipping to wholesalers in the United States in late December 2020 and for which we commenced a full promotional launch in early January 2021.

#### Selling, general and administrative expenses

Selling, general and administrative expenses were \$28.0 million for the three months ended June 30, 2021, compared to \$15.3 million for the three months ended June 30, 2020, which was an increase of \$12.7 million. Selling, general and administrative expenses for the three months ended June 30, 2021 included a \$4.8 million increase in external sales and marketing costs as compared to the three months ended June 30, 2020, primarily as a result of the launch of EYSUVIS. External sales and marketing costs incurred during the three months ended June 30, 2020 primarily related to commercial activities for INVELTYS. Also contributing to the increase in selling, general and administrative expenses for the three months ended June 30, 2021 was a \$5.7 million increase in employee-related expenses primarily due to an increase in employee headcount related to the launch of EYSUVIS and merit-based pay, a \$1.6 million increase in stock-based compensation costs and a \$0.9 million increase in other selling, general and administrative expenses, which included facility related costs and certain medical affairs costs attributable to our commercial products. These increases were partially offset by a \$0.3 million decrease in costs for administrative and professional service fees. We anticipate that our selling, general and administrative expenses will increase in the future as we continue to commercialize EYSUVIS and INVELTYS and if and as we increase our administrative headcount to support our continued research and development activities and seek marketing approval for our product candidates.

#### Research and development expenses

The following table summarizes the research and development expenses incurred during the three months ended June 30, 2021 and 2020:

		Three Months Ended					
		Jun					
	2021 2020			020 Change			
		(in tho					
KPI-121 development costs	\$	16	\$	2,034	\$	(2,018)	
Employee-related costs		2,106		3,357		(1,251)	
Other research and development costs		972		662		310	
Total research and development	\$	3,094	\$	6,053	\$	(2,959)	

Research and development expenses were \$3.1 million for the three months ended June 30, 2021, compared to \$6.1 million for the three months ended June 30, 2020, a \$3.0 million decrease. The decrease was primarily the result of a \$2.0 million decrease in EYSUVIS development costs related to a decrease in external spend on STRIDE 3, our Phase 3 clinical trial of EYSUVIS, and a \$1.3 million decrease in employee-related costs largely due to the decrease in the allocation of employee time dedicated to research and development, partially offset by a \$0.3 million increase in other research and development costs, which included preclinical studies and other facility related costs. We expect research and development costs to increase if and as we advance our development programs and conduct any necessary preclinical studies and clinical trials and other development activities for product candidates.

#### Interest income

Interest income was less than \$0.1 million for the three months ended June 30, 2021, compared to \$0.1 million for the three months ended June 30, 2020, a decrease of less than \$0.1 million. Interest income consists of interest earned on our cash, cash equivalents and short-term investments. The decrease was attributable to lower interest rates during the three months ended June 30, 2021.

#### Interest expense

We incurred interest expense of \$2.1 million for the three months ended June 30, 2021, which remained consistent with interest expense of \$2.1 million for the three months ended June 30, 2020. Interest expense was comprised of the contractual coupon interest expense, the amortization of the debt discount and the accretion of the final payment fee associated with our Loan Agreement with Oxford Finance LLC and our Athyrium Credit Facility. During the three months ended June 30, 2021, \$75.0 million of indebtedness was outstanding under the Athyrium Credit Facility until we repaid such indebtedness in full on May 4, 2021. During the three months ended June 30, 2021, \$80.0 million of indebtedness was outstanding under our Loan Agreement after we drew down the tranche A term loan on May 4, 2021. During the three months ended June 30, 2020, \$75.0 million of indebtedness was outstanding under the Athyrium Credit Facility.

#### Loss on extinguishment of debt

The loss of extinguishment of debt was \$5.4 million for the three months ended June 30, 2021. Upon the repayment in full of all amounts owed under the Athyrium Credit Facility, the unamortized debt discount and issuance costs, prepayment premium and unaccreted exit fee were recorded as loss on extinguishment of debt for the three months ended June 30, 2021. There was no loss on extinguishment of debt for the three months ended June 30, 2020.

#### Comparison of the Six Months Ended June 30, 2021 and 2020

The following table summarizes the results of our operations for the six months ended June 30, 2021 and 2020:

	Six Months Ended					
	June 30,					
		2021	2020		Change	
	(in thousands)					
Product revenues, net	\$	6,317	\$	1,904	\$	4,413
Costs and expenses:						
Cost of product revenues		1,771		1,113		658
Selling, general and administrative		55,685		30,709		24,976
Research and development		6,220		11,487		(5,267)
Total operating expenses		63,676		43,309		20,367
Loss from operations		(57,359)		(41,405)		(15,954)
Other income (expense)						
Interest income		76		400		(324)
Interest expense		(4,232)		(4,262)		30
Loss on extinguishment of debt		(5,395)		_		(5,395)
Net loss	\$ (	(66,910)	\$	(45,267)	\$	(21,643)

#### Product revenues, net

Product revenues, net was \$6.3 million for the six months ended June 30, 2021, consisting of \$3.3 million from EYSUVIS sales and \$3.0 million from INVELTYS sales, compared to \$1.9 million from INVELTYS sales for the six months ended June 30, 2020. There were no sales of EYSUVIS in the six months ended June 30, 2020. The increase in product revenues, net of \$4.4 million was driven primarily by sales of EYSUVIS, which we began shipping to wholesalers in the United States in late December 2020, an increase in the total units of INVELTYS sold in the six months ended June 30, 2021 as well as a higher per unit gross selling price of INVELTYS as compared to those sold during the six months ended June 30, 2020. These increases in INVELTYS sales were partially offset by higher estimated reserves per unit during the six months ended June 30, 2021 as compared to those estimated during the six months ended June 30, 2020. We expect product revenues to increase if and as we increase our market share and obtain and maintain coverage and adequate reimbursement for EYSUVIS and INVELTYS from third-party payors; however, revenues could continue to be negatively impacted in 2021 as a result of the COVID-19 pandemic.

#### Cost of product revenues

Cost of product revenues was \$1.8 million for the six months ended June 30, 2021, compared to \$1.1 million in the six months ended June 30, 2020, an increase of \$0.7 million. Cost of product revenues increased \$0.8 million due to an increase in total INVELTYS units sold during the six months ended June 30, 2021, compared to the six months ended June 30, 2020 as well as a higher INVELTYS cost per unit as a result of the units sold during the six months ended June 30, 2020 being partially manufactured prior to FDA approval and for which costs were expensed as research and development prior to FDA approval as compared to those units sold during the six months ended June 30, 2021. Partially offsetting these increases was a reserve for excess inventory of \$0.5 million recorded during the six months ended June 30, 2020, which did not occur in the six months ended June 30, 2021. The cost of product revenues attributable to EYSUVIS was \$0.4 million for the six months ended June 30, 2021. There were no sales of EYSUVIS in the six months ended June 30, 2020. We expect aggregate cost of product revenues to increase as we continue to commercialize INVELTYS and as a result of the launch of EYSUVIS, which we began shipping to wholesalers in the United States in late December 2020 and for which we commenced a full promotional launch in early January 2021.

#### Selling, general and administrative expenses

Selling, general and administrative expenses were \$55.7 million for the six months ended June 30, 2021, compared to \$30.7 million for the six months ended June 30, 2020, which was an increase of \$25.0 million. Selling, general and administrative expenses for the six months ended June 30, 2021 included a \$10.3 million increase in external

sales and marketing costs as compared to the six months ended June 30, 2020, primarily as a result of the launch of EYSUVIS. External sales and marketing costs incurred during the six months ended June 30, 2020 primarily related to commercial activities for INVELTYS. Also contributing to the increase in selling, general and administrative expenses for the six months ended June 30, 2021 was a \$9.8 million increase in employee-related expenses primarily due to an increase in employee headcount related to the launch of EYSUVIS and merit-based pay, a \$3.4 million increase in stock-based compensation costs and a \$1.5 million increase in other selling, general and administrative expenses, which included facility related costs and certain medical affairs costs attributable to our commercial products. We anticipate that our selling, general and administrative expenses will increase in the future as we continue to commercialize EYSUVIS and INVELTYS and if and as we increase our administrative headcount to support our continued research and development activities and seek marketing approval for our product candidates.

#### Research and development expenses

The following table summarizes the research and development expenses incurred during the six months ended June 30, 2021 and 2020:

	Six Mon Jun	_				
	 2021 2020			Change		
	(in thousands)					
KPI-121 development costs	\$ 89	\$	3,715	\$	(3,626)	
Employee-related costs	4,311		6,234		(1,923)	
Other research and development costs	 1,820		1,538		282	
Total research and development	\$ 6,220	\$	11,487	\$	(5,267)	

Research and development expenses were \$6.2 million for the six months ended June 30, 2021, compared to \$11.5 million for the six months ended June 30, 2020, a \$5.3 million decrease. The decrease was primarily the result of a \$3.6 million decrease in EYSUVIS development costs related to a decrease in external spend on STRIDE 3, our Phase 3 clinical trial of EYSUVIS, and a \$1.9 million decrease in employee-related costs largely due to the decrease in the allocation of employee time dedicated to research and development, partially offset by a \$0.2 million increase in other research and development costs, which included preclinical studies and other facility related costs. We expect research and development costs to increase if and as we advance our development programs and conduct any necessary preclinical studies and clinical trials and other development activities for product candidates.

#### Interest income

Interest income was \$0.1 million for the six months ended June 30, 2021, compared to \$0.4 million for the six months ended June 30, 2020, a decrease of \$0.3 million. Interest income consists of interest earned on our cash, cash equivalents and short-term investments. The decrease was attributable to lower interest rates during the six months ended June 30, 2021.

#### Interest expense

We incurred interest expense of \$4.2 million for the six months ended June 30, 2021, compared to \$4.3 million for the six months ended June 30, 2020, a \$0.1 million decrease. Interest expense was comprised of the contractual coupon interest expense, the amortization of the debt discount and the accretion of the final payment fee associated with our Loan Agreement with Oxford Finance LLC and our Athyrium Credit Facility. During the six months ended June 30, 2021, \$75.0 million of indebtedness was outstanding under the Athyrium Credit Facility until we repaid such indebtedness in full on May 4, 2021. During the six months ended June 30, 2021, \$80.0 million of indebtedness was outstanding under our Loan Agreement after we drew down the tranche A term loan on May 4, 2021. During the six months ended June 30, 2020, \$75.0 million of indebtedness was outstanding under the Athyrium Credit Facility.

#### Loss on extinguishment of debt

The loss of extinguishment of debt was \$5.4 million for the six months ended June 30, 2021. Upon the repayment in full of all amounts owed under the Athyrium Credit Facility, the unamortized debt discount and issuance costs, prepayment premium and unaccreted exit fee were recorded as loss on extinguishment of debt for the six months ended June 30, 2021. There was no loss on extinguishment of debt for the six months ended June 30, 2020.

#### **Liquidity and Capital Resources**

Since our inception, we have incurred significant operating losses. As we commercially launched our first product, INVELTYS, in January 2019, and commenced a full promotional launch of our second product, EYSUVIS, in early January 2021, we have had limited revenues to date from product sales and have financed our operations primarily through proceeds from our IPO, follow-on public common stock offerings and sales of our common stock under our ATM Offerings, private placements of preferred stock, borrowings under credit facilities and the Loan Agreement, convertible promissory notes and warrants.

In July 2017, we completed an IPO pursuant to which we issued and sold 6,900,000 shares of our common stock, which included 900,000 shares sold pursuant to the exercise of the underwriters' option to purchase additional shares, at a price of \$15.00 per share. We received net proceeds of \$94.0 million after deducting underwriting discounts and commission of \$7.3 million and offering costs of \$2.2 million.

On August 9, 2018, we filed our shelf registration statement on Form S-3 that was declared effective by the SEC on August 27, 2018, or the 2018 Shelf Registration, under which we could initially offer and sell up to \$250.0 million of a variety of securities including common stock, preferred stock, warrants, depositary shares, debt securities, purchase contracts, purchase units or any combination of such securities during the three-year period that commenced upon the 2018 Shelf Registration becoming effective. Under the 2018 Shelf Registration, we may periodically offer one or more types of securities in amounts, at prices and on terms announced, if and when the securities are ever offered.

On October 1, 2018, we entered into the Athyrium Credit Facility with Athyrium for up to \$110.0 million. The Athyrium Credit Facility provided for a Term Loan A in the aggregate principal amount of \$75.0 million, and a Term Loan B in the aggregate principal amount of \$35.0 million which we did not draw down on. On May 4, 2021, concurrently with the closing of the Loan Agreement with Oxford Finance LLC and the initial borrowing of the tranche A loan, we utilized substantially all of the proceeds from the tranche A term loan to repay in full all outstanding amounts owed under our Athyrium Credit Facility, under which we had an aggregate principal amount of \$75.0 million of indebtedness outstanding. We terminated all commitments by Athyrium to extend further credit under the Athyrium Credit Facility and all guarantees and security interests granted by us thereunder. In connection with the termination of the Athyrium Credit Facility, we paid to the lenders a prepayment premium of \$2.25 million and an exit fee of \$0.8 million. The transaction resulted in a loss on extinguishment of debt of \$5.4 million, consisting of the prepayment premium, the unamortized debt discount and the unaccreted exit fee.

On October 5, 2018, we sold 7,500,000 shares of common stock in an underwritten offering pursuant to the 2018 Shelf Registration at a public offering price of \$8.25 per share, before underwriting discounts and commissions. In addition, the underwriters were granted an overallotment option to purchase an additional 1,125,000 shares of the common stock at the same public offering price, less underwriting discounts and commissions. On October 11, 2018, the underwriters exercised in full their option to purchase the overallotment shares. The total number of shares sold by us in the offering was 8,625,000 shares, resulting in net proceeds to us, after underwriting discounts and offering expenses, of \$66.1 million. In connection with the filing of the 2018 Shelf Registration, we entered into a sales agreement with Jefferies, pursuant to which we could issue and sell, from time to time, up to an aggregate of \$50.0 million of our common stock in an ATM Offering, through Jefferies, as sales agent. Through the first quarter of 2020, we issued an aggregate of 4,945,605 shares of our common stock under the ATM Offering, resulting in net proceeds to us of \$25.6 million. On March 10, 2020, we suspended and terminated the prospectus related to the ATM Offering.

On March 11, 2020, we sold 16,000,000 shares of our common stock in an underwritten offering pursuant to the 2018 Shelf Registration at a public offering price of \$7.89 per share, resulting in net proceeds of \$118.2 million, after underwriting discounts, commissions, and offering expenses. In addition, the underwriters of the offering were granted the option for a period of 30 days to purchase up to an additional 2,400,000 shares of common stock offered in the public offering at the public offering price, less underwriting discounts, commissions and offering expenses. On April 3, 2020, the underwriters exercised their option and purchased an additional 979,371 shares of common stock at \$7.89 per share, resulting in net proceeds to us of \$7.2 million, after underwriting discounts, commissions, and offering expenses. The total number of shares sold by us in the offering was 16,979,371, resulting in total net proceeds to us, after underwriting discounts, commissions and offering expenses, of \$125.4 million.

Under the 2018 Shelf Registration, we have issued an aggregate of 30,549,976 shares of common stock, including under the ATM Offering, resulting in aggregate gross proceeds to us of \$231.7 million. There was \$18.3 million of securities available to be issued under the 2018 Shelf Registration as of June 30, 2021.

On May 7, 2020, we filed our shelf registration statement on Form S-3 that was declared effective by the SEC on May 7, 2020, or the 2020 Shelf Registration, under which we may offer and sell up to \$350.0 million of a variety of securities including common stock, preferred stock, warrants, depositary shares, debt securities or units during the three-year period that commenced upon the 2020 Shelf Registration becoming effective. In connection with the filing of the 2020 Shelf Registration, we entered into an amended and restated sales agreement with Jefferies, pursuant to which we may issue and sell, from time to time, up to an aggregate of \$75.0 million of our common stock under our ATM Offering. During the fourth quarter of 2020, we issued an aggregate of 2,821,059 shares of our common stock under the ATM Offering, resulting in net proceeds to us of \$20.6 million. In the three and six months ended June 30, 2021, we issued and sold an additional 837,257 and 5,583,329 shares of our common stock under our ATM Offering, respectively, resulting in net proceeds to us of \$6.0 million and \$40.7 million, respectively. As of June 30, 2021, there was \$11.9 million of shares of common stock remaining under the ATM Offering that we may issue and sell in the future and, excluding the shares of common stock that may be offered under our ATM Offering, there was \$275.0 million of securities available to be issued under the 2020 Shelf Registration.

On May 4, 2021, we entered into the Loan Agreement with Oxford Finance, in its capacity as lender, or the Lender, and in its capacity as collateral agent (in such capacity, the "Agent"), pursuant to which a term loan of up to an aggregate principal amount of \$125.0 million is available to us, consisting of (i) a tranche A term loan that was disbursed on the closing date of the Loan Agreement in the aggregate principal amount of \$80.0 million; (ii) a contingent tranche B term loan in the aggregate principal amount of \$20.0 million available to us through June 30, 2023 and within 90 days of our achieving trailing 6-month product revenue equal to or greater than \$75.0 million, subject to certain other terms and conditions; and (iii) a contingent tranche C term loan in the aggregate principal amount of \$25.0 million available to us through December 31, 2023 and within 90 days of our achieving trailing 6-month product revenue equal to or greater than \$100 million, subject to certain other terms and conditions. The term loans bear interest at a floating rate equal to the greater of 30-day LIBOR and 0.11%, plus 7.89%. Certain of the customary negative covenants limit our and certain of our subsidiaries' ability, among other things, to incur future debt, grant liens, make investments, make acquisitions, distribute dividends, make certain restricted payments and sell assets, subject in each case to certain exceptions. The Loan Agreement provides for interest-only payments until December 1, 2024 if neither the tranche B term loan nor the tranche C term loan are made, and until June 1, 2025 if either the tranche B term loan or the tranche C term loan is made, or the Amortization Date. The aggregate outstanding principal balance of the term loans are required to be repaid in monthly installments starting on the Amortization Date based on a repayment schedule equal to (i) 18 months if neither the tranche B term loan nor the tranche C term loan is made and (ii) 12 months if either the tranche B term loan or the tranche C term loan is made. All unpaid principal and accrued and unpaid interest with respect to each term loan is due and payable in full on May 1, 2026, or the Maturity Date.

We paid a facility fee of \$400,000 on the closing date of the Loan Agreement and have agreed to pay a facility fee of \$100,000 upon closing of the tranche B term loan and a \$125,000 facility fee upon the closing of the tranche C term loan. We will be required to make a final payment fee of 7.00% of the original principal amount of any funded term loan payable on the earlier of (i) the prepayment of the term loan in full or (ii) the Maturity Date. At our option, the we may elect to prepay all, but not less than all, of the outstanding loans, subject to a prepayment fee equal to the following percentage of the principal amount being prepaid: 3.00% if an advance is prepaid during the first 12 months following

the applicable advance date, 2.00% if an advance is prepaid after 12 months but prior to 24 months following the applicable advance date, and 1.00% if an advance is prepaid any time after 24 months following the applicable advance date but prior to the Maturity Date.

#### Cash Flows

As of June 30, 2021 and 2020, we had \$149.6 million and \$184.6 million, respectively, in cash, cash equivalents and short-term investments. As of June 30, 2021, we had \$80.0 million in indebtedness, which represented the aggregate principal amount that was outstanding under the Loan Agreement with Oxford Finance LLC. As of June 30, 2020, we had \$75.0 million in indebtedness, which represented the aggregate principal amount that was outstanding under the Athyrium Credit Facility.

The following table summarizes our sources and uses of cash for the six months ended June 30, 2021 and 2020:

		naca		
		2021		2020
		ds)		
Net cash used in operating activities	\$	(54,248)	\$	(39,134)
Net cash provided by (used in) investing activities		70,705		(57,450)
Net cash provided by financing activities		41,164		139,160
Increase in cash and restricted cash	\$	57,621	\$	42,576

#### **Operating Activities**

Net cash used in operating activities for the six months ended June 30, 2021 was \$54.2 million compared to \$39.1 million for the six months ended June 30, 2020, an increase of \$15.1 million, primarily due to an \$11.7 million increase in the net loss adjusted for non-cash charges and the timing of working capital fluctuations which accounted for \$3.4 million of the increase. Notable working capital fluctuations include an increase to accounts receivable in the six months ended June 30, 2021 of \$2.3 million driven by an increase in sales largely due to the launch of EYSUVIS, whereas accounts receivable had decreased by \$7.3 million in the six months ended June 30, 2020 driven by a decrease in sales of INVELTYS primarily as a result of the COVID-19 pandemic. Inventory increased by a greater amount during the six months ended June 30, 2021 due to an increase in manufacturing activity for EYSUVIS and INVELTYS. Offsetting these increases was an increase in accrued expenses and other current liabilities during the six months ended June 30, 2021 of \$2.0 million, as compared to a decrease in accrued expenses and other current liabilities in the six months ended June 30, 2020 of \$6.8 million.

#### **Investing Activities**

Net cash provided by investing activities for the six months ended June 30, 2021 was \$70.7 million compared to net cash used of \$57.5 million for the same period in 2020, an increase of \$128.2 million. Net cash provided by investing activities for the six months ended June 30, 2021 primarily related to the sales or maturities of short-term investments \$71.3 million. Net cash used in investing activities for the six months ended June 30, 2020 primarily related to the purchases of short-term investments of \$56.5 million.

#### Financing Activities

Net cash provided by financing activities for the six months ended June 30, 2021 was \$41.2 million, a decrease of \$98.0 million compared to \$139.2 million in the six months ended June 30, 2020. Net cash provided by financing activities for the six months ended June 30, 2021 included \$77.8 million of net proceeds from the tranche A term loan under our Loan Agreement, \$40.7 million of net proceeds from the sale of shares of our common stock under the ATM Offering and \$0.7 million of proceeds from the exercise of stock options and the issuance of common stock under our employee stock purchase plan, partially offset by the repayment of indebtedness under our Athyrium Credit Facility of \$78.0 million. Net cash provided by financing activities for the six months ended June 30, 2020 consisted of \$125.4 million of net proceeds from the sale of shares of our underwritten offering pursuant to the 2018 Shelf Registration,

\$12.6 million of net proceeds from the sale of shares of our common stock under the ATM Offering and \$1.2 million of proceeds from the exercise of stock options and the issuance of common stock under our employee stock purchase plan.

#### **Funding Requirements**

We anticipate that our expenses will increase substantially as compared to prior periods as we continue to commercialize INVELTYS in the United States and execute our commercial launch plans for EYSUVIS, as a result of increased headcount, including management personnel to support our clinical, manufacturing and commercialization activities, expanded infrastructure, increased legal, compliance, accounting and investor and public relations expenses associated with being a public company and increased insurance premiums, among other factors. The increase in headcount includes the expansion of our sales force from 56 TSMs to 91 TSMs, which occurred in the fourth quarter of 2020, and from 91 TSMs to 105 TSMs, which occurred around the start of the third quarter 2021, with a subsequent expansion up to 125 TSMs expected by year-end, subject to continued growth in payor coverage and the status of the COVID-19 pandemic.

Our expenses will also increase if and as we:

- continue to grow our sales, marketing and distribution capabilities in connection with the commercialization
  of EYSUVIS, INVELTYS and any product candidates for which we may submit for and obtain marketing
  approval;
- continue to scale up our manufacturing processes and capabilities to support commercialization of EYSUVIS and INVELTYS:
- seek regulatory approval for EYSUVIS and INVELTYS outside of the United States;
- progress our current and any future preclinical development programs;
- in-license or acquire the rights to other products, product candidates or technologies;
- conduct clinical trials and other development activities and/or seek marketing approval for future product candidates;
- leverage our proprietary AMPPLIFY technology to seek to advance additional therapeutics into preclinical and clinical development;
- maintain, expand and protect our intellectual property portfolio;
- hire additional clinical, quality control, scientific, manufacturing, commercial and management personnel;
- expand our operational, financial and management systems; and
- increase our product liability insurance coverage as we expand our commercialization efforts for EYSUVIS and INVELTYS.

We expect to continue to incur significant expenses and operating losses. Net losses may fluctuate significantly from quarter-to-quarter and year-to-year. We anticipate that our cash, cash equivalents and short-term investments as of June 30, 2021, along with anticipated revenue from EYSUVIS and INVELTYS, will enable us to fund our operations, lease and debt service obligations, and capital expenditure requirements for at least two years. We have based this estimate on assumptions that may prove to be wrong, and our operating plan may change as a result of many factors currently unknown to us. As a result, we could deplete our available capital resources sooner or later than we currently expect.

Because of the numerous risks and uncertainties associated with pharmaceutical product development, we are unable to accurately predict the timing or amount of increased expenses or when, or if, we will be able to achieve profitability. Our expenses will increase from what we anticipate if:

- we elect or are required by the FDA or non-U.S. regulatory agencies to perform clinical trials or studies in addition to those expected;
- there are any delays in enrollment of patients in or completing our clinical trials or the development of our product candidates;
- we in-license or acquire rights to other products, product candidates or technologies; or
- there are any third-party challenges to our intellectual property portfolio, or the need arises to defend against
  intellectual property-related claims or enforce our intellectual property rights.

Our ability to become and remain profitable depends on our ability to generate revenue. While we began to generate revenue from the sales of EYSUVIS and INVELTYS in late December 2020 and January 2019, respectively, there can be no assurance as to the amount or timing of any future revenue from these products, and we may not achieve profitability. Achieving and maintaining profitability will require us to be successful in a range of challenging activities, including:

- successfully launching EYSUVIS and growing EYSUVIS revenues;
- successfully growing INVELTYS revenues;
- achieving an adequate level of market acceptance, and obtaining and maintaining coverage and adequate reimbursement from third-party payors for EYSUVIS, INVELTYS and any other products we commercialize;
- manufacturing at commercial scale, marketing, selling and distributing EYSUVIS and INVELTYS;
- maintaining regulatory and marketing approvals for EYSUVIS and INVELTYS;
- discovering, developing and successfully seeking marketing approval and commercialization of additional product candidates;
- hiring and building a full commercial organization required for marketing, selling and distributing those products for which we obtain marketing approval;
- obtaining, maintaining and protecting our intellectual property rights; and
- adapting our business in response to the current pandemic health event resulting from COVID-19 and its collateral consequences.

EYSUVIS and INVELTYS are our only products that have been approved for sale, and they have only been approved in the United States. We plan to seek approval in other jurisdictions, but may not do so successfully, or at all. Further, the successful commercialization of EYSUVIS and INVELTYS in the United States is subject to many risks. As a company, we have limited experience commercializing products, and we may not be able to do so successfully. There are numerous examples of unsuccessful product launches and failures to meet expectations of market potential, including by pharmaceutical companies with more experience and resources than us. Our revenue from sales of EYSUVIS and INVELTYS alone may not be sufficient for us to become profitable in the near future, if at all.

In addition, our recent commercialization efforts have been hampered by the operational restrictions on our sales force from quarantines, travel restrictions and bans and other governmental restrictions related to the COVID-19 pandemic. As a result of these restrictions, we previously suspended our sales force from substantially all in-person interactions with physicians and were limited to conducting educational and promotional activities virtually. Our sales force has since resumed substantially all in-person interactions in the field, but to the extent we restrict, or are restricted from, in-person interactions with physicians, we may be limited to conducting educational and promotional activities virtually, which may continue to hamper our ability to market INVELTYS and could adversely affect our ability to launch and market EYSUVIS. In addition, government restrictions have at times led to moratoria on elective ocular surgeries in many jurisdictions, which had significantly reduced, and may in the future continue to significantly reduce, the demand for INVELTYS, which is indicated for the treatment of post-operative inflammation and pain following ocular surgery. The extent of the impact of COVID-19 on our commercialization efforts will depend on the length and severity of this pandemic, including the extent there is any resurgence of the COVID-19 virus or any variant strains of the virus, the availability and effectiveness of vaccines, and the impact of the foregoing on our customers, employees and vendors and government agencies, which is uncertain and cannot be predicted.

We may never succeed in these activities and may never generate revenue that is sufficient to achieve profitability. Even if we do achieve profitability, we may not be able to sustain or increase profitability on a quarterly or annual basis. Our failure to become and remain profitable would decrease the value of our company and could impair our ability to raise capital, expand our business, maintain our research and development efforts, diversify our product offerings or even continue our operations. A decline in the value of our company could also cause you to lose all or part of your investment.

Until such time, if ever, as we can generate substantial product revenues, we expect to finance our cash needs through a combination of equity offerings, debt financings, collaborations, strategic alliances, licensing arrangements, royalty agreements, and marketing and distribution arrangements. To the extent that we raise additional capital through the sale of equity or convertible debt securities, your ownership interest will be diluted, and the terms of these securities may include liquidation or other preferences that adversely affect your rights as a common stockholder. Debt financing and preferred equity financing, if available, may involve agreements that include pledging of assets as collateral, covenants limiting or restricting our ability to take specific actions, such as incurring additional debt, making capital expenditures or declaring dividends. Our pledge of our assets as collateral to secure our obligations under our Loan Agreement may limit our ability to obtain additional debt financing. Under our Loan Agreement, we are also restricted from incurring future debt, granting liens, making investments, making acquisitions, distributing dividends on our common stock, making certain restricted payments and selling assets and making certain other uses of our cash, without the lenders' consent, subject in each case to certain exceptions.

We may need to raise additional capital in the future to advance our business. Additional private or public financings may not be available to us on acceptable terms, or at all. Additionally, the COVID-19 pandemic has already caused significant disruptions in the financial markets, and may again cause such disruptions, which could impact our ability to raise additional funds. The COVID-19 pandemic has also impacted, and may continue to impact, the volatility of our stock price and trading in our stock. Even after the COVID-19 pandemic has subsided, we may continue to experience adverse impacts to our business as a result of any economic recession or depression that has occurred or may occur in the future.

Our failure to raise capital as and when needed would have a material adverse effect on our financial condition and our ability to pursue our business strategy. If we raise additional funds through collaborations, strategic alliances, licensing arrangements, royalty agreements, or marketing and distribution arrangements, we may have to relinquish valuable rights to our technologies, future revenue streams, research programs or product candidates or grant licenses on terms that may not be favorable to us. If we are unable to raise additional funds through equity or debt financings when needed, we may be required to delay, limit, reduce or terminate our product development or current or future commercialization efforts or grant rights to develop and market products or product candidates that we would otherwise prefer to develop and market ourselves.

## **Off-Balance Sheet Arrangements**

We did not have during the periods presented, and we do not currently have, any off-balance sheet arrangements, as defined in the rules and regulations of the Securities and Exchange Commission.

## Item 3. Quantitative and Qualitative Disclosures About Market Risk.

Our financial instruments consist primarily of cash equivalents and short-term investments. Our cash equivalents as of June 30, 2021 consisted of money market accounts that have contractual maturities of less than 90 days from the date of acquisition. Our short-term investments as of June 30, 2021 consist of U.S. Government Agencies Securities. Due to the short-term maturities of our cash equivalents and short-term investments, and the fixed income nature of these investments, an immediate 10% change in interest rates would not have a material effect on the fair market value of our cash equivalents and short-term investments.

As of June 30, 2021, the aggregate principal amount outstanding under the Loan Agreement was \$80.0 million, which bears interest at a floating rate equal to the greater of 30-day LIBOR and 0.11%, plus 7.89% per annum. An immediate 10% change in the 30-day LIBOR rate would not have a material impact on our operating results or cash flows. As of June 30, 2020, the aggregate principal amount outstanding under the Athyrium Credit Facility was \$75.0 million, which bore interest at a fixed rate of 9.875% per annum. On May 4, 2021, we utilized substantially all of the proceeds from the tranche A term loan under the Loan Agreement to repay in full all outstanding amounts under the Athyrium Credit Facility.

#### Item 4. Controls and Procedures.

### **Evaluation of Disclosure Controls and Procedures.**

Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, evaluated the effectiveness of our disclosure controls and procedures as of June 30, 2021. The term "disclosure controls and procedures," as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended, or the Exchange Act, means controls and other procedures of a company that are designed to ensure that information required to be disclosed by a company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by a company in the reports that it files or submits under the Exchange Act is accumulated and communicated to the company's management, including its principal executive and principal financial officers, as appropriate to allow timely decisions regarding required disclosure. Management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving their objectives and management necessarily applies its judgment in evaluating the cost-benefit relationship of possible controls and procedures. Based on the evaluation of our disclosure controls and procedures as of June 30, 2021, our Chief Executive Officer and Chief Financial Officer concluded that, as of such date, our disclosure controls and procedures were effective at the reasonable assurance level.

# Changes in Internal Control over Financial Reporting.

There were no changes in our internal control over financial reporting that occurred during the three-month period ended June 30, 2021 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

## **PART II - OTHER INFORMATION**

# Item 1. Legal Proceedings.

We are not currently subject to any material legal proceedings.

### Item 1A. RISK FACTORS

Investing in our common stock involves a high degree of risk. You should carefully consider the risks and uncertainties described below together with all of the other information contained in our Annual Report on Form 10-K and this Quarterly Report on Form 10-Q, including our financial statements and the related notes appearing at the end of our Annual Report on Form 10-K and this Quarterly Report on Form 10-Q, before deciding to invest in our common stock. These risks, some of which have occurred and any of which may occur in the future, can have a material adverse effect on our business, prospects, operating results and financial condition. In such event, the trading price of our common stock could decline and you might lose all or part of your investment. The risks and uncertainties described below are not the only ones we face. Additional risks and uncertainties not presently known to us or that we presently deem less significant may also impair our business, prospects, operating results and financial condition.

# Risks Related to Our Financial Position and Need For Additional Capital

We have incurred significant losses from operations and negative cash flows from operations since our inception. We expect to incur additional losses and may never achieve or maintain profitability.

Since inception, we have incurred significant losses from operations and negative cash flows from operations. Our net losses were \$36.5 million and \$66.9 million for the three and six months ended June 30, 2021 and \$104.3 million for the year ended December 31, 2020. As of June 30, 2021, we had an accumulated deficit of \$466.7 million. In January 2019, we launched our first product, INVELTYS® (loteprednol etabonate ophthalmic suspension) 1% for the treatment of post-operative inflammation and pain following ocular surgery. On October 26, 2020, the U.S. Food and Drug Administration, or FDA, approved our second product, EYSUVIS® (loteprednol etabonate ophthalmic suspension) 0.25% for the short-term (up to two weeks) treatment of the signs and symptoms of dry eye disease. We began shipping EYSUVIS to wholesalers in the United States in late December 2020 and commenced a full promotional launch in early January 2021. We have had limited revenues to date from product sales. We have financed our operations primarily through proceeds from our initial public offering, or IPO, follow-on public offerings of common stock and sales under our at-the-market offering facility, or the ATM Offering, private placements of preferred stock, borrowings under credit facilities and the Loan and Security Agreement with Oxford Finance LLC, or the Loan Agreement, convertible promissory notes and warrants. We have devoted substantially all of our financial resources and efforts to research and development, including preclinical studies and clinical trials, and engaging in activities to launch and commercialize EYSUVIS and INVELTYS. Although we expect to continue to generate revenue from sales of EYSUVIS and INVELTYS, there can be no assurance as to the amount or timing of any such revenue, and we expect to continue to incur significant expenses and operating losses. We may never achieve or maintain profitability. Our net losses may fluctuate significantly from quarter-to-quarter and yearto-year.

We anticipate that our expenses will increase substantially as compared to prior periods as we continue to commercialize INVELTYS in the United States and execute our commercial launch plans for EYSUVIS, as a result of increased headcount, including management personnel to support our clinical, manufacturing and commercialization activities, expanded infrastructure, increased legal, compliance, accounting and investor and public relations expenses associated with being a public company and increased insurance premiums, among other factors. The increase in headcount includes the expansion of our sales force from 56 territory sales managers, or TSMs, to 91 TSMs, which occurred in the fourth quarter of 2020, and from 91 TSMs to 105 TSMs, which occurred around the start of the third quarter 2021, with a subsequent expansion up to 125 TSMs expected by year-end, subject to continued growth in payor coverage and the status of the COVID-19 pandemic.

Our expenses will also increase if and as we:

- continue to grow our sales, marketing and distribution capabilities in connection with the commercialization
  of EYSUVIS, INVELTYS and any product candidates for which we may submit for and obtain marketing
  approval;
- continue to scale up our manufacturing processes and capabilities to support commercialization of EYSUVIS and INVELTYS;

- seek regulatory approval for EYSUVIS and INVELTYS outside of the United States;
- progress our current and any future preclinical development programs;
- in license or acquire the rights to other products, product candidates or technologies;
- conduct clinical trials and other development activities and/or seek marketing approval for future product candidates;
- leverage our proprietary AMPPLIFY technology to seek to advance additional therapeutics into preclinical and clinical development;
- maintain, expand and protect our intellectual property portfolio;
- hire additional clinical, quality control, scientific, manufacturing, commercial and management personnel;
- expand our operational, financial and management systems; and
- increase our product liability insurance coverage as we expand our commercialization efforts for EYSUVIS and INVELTYS.

Because of the numerous risks and uncertainties associated with pharmaceutical product development, we are unable to accurately predict the timing or amount of increased expenses or when, or if, we will be able to achieve profitability. Our expenses will increase from what we anticipate if:

- we elect or are required by the FDA or non-U.S. regulatory agencies to perform clinical trials or studies in addition to those expected;
- there are any delays in enrollment of patients in or completing our clinical trials or the development of our product candidates;
- we in-license or acquire rights to other products, product candidates or technologies; or
- there are any third-party challenges to our intellectual property portfolio, or the need arises to defend against intellectual property-related claims or enforce our intellectual property rights.

Our ability to become and remain profitable depends on our ability to generate revenue. While we began to generate revenue from the sales of EYSUVIS and INVELTYS in late December 2020 and January 2019, respectively, there can be no assurance as to the amount or timing of any future revenue from these products, and we may not achieve profitability. Achieving and maintaining profitability will require us to be successful in a range of challenging activities, including:

- successfully launching EYSUVIS and growing EYSUVIS revenues;
- successfully growing INVELTYS revenues;
- achieving an adequate level of market acceptance, and obtaining and maintaining coverage and adequate reimbursement from third-party payors for EYSUVIS, INVELTYS and any other products we commercialize;
- manufacturing at commercial scale, marketing, selling and distributing EYSUVIS and INVELTYS;

- maintaining regulatory and marketing approvals for EYSUVIS and INVELTYS;
- discovering, developing and successfully seeking marketing approval and commercialization of additional product candidates;
- hiring and building a full commercial organization required for marketing, selling and distributing those products for which we obtain marketing approval;
- obtaining, maintaining and protecting our intellectual property rights; and
- adapting our business in response to the current pandemic health event resulting from COVID-19 and its
  collateral consequences.

EYSUVIS and INVELTYS are our only products that have been approved for sale, and they have only been approved in the United States. We plan to seek approval in other jurisdictions, but may not do so successfully, or at all. Further, the successful commercialization of EYSUVIS and INVELTYS in the United States is subject to many risks. As a company, we have limited experience commercializing products, and we may not be able to do so successfully. There are numerous examples of unsuccessful product launches and failures to meet expectations of market potential, including by pharmaceutical companies with more experience and resources than us. Our revenue from sales of EYSUVIS and INVELTYS alone may not be sufficient for us to become profitable in the near future, if at all.

In addition, our recent commercialization efforts have been hampered by the operational restrictions on our sales force from quarantines, travel restrictions and bans and other governmental restrictions related to the COVID-19 pandemic. As a result of these restrictions, we previously suspended our sales force from substantially all in-person interactions with physicians and were limited to conducting educational and promotional activities virtually. Our sales force has since resumed substantially all in-person interactions in the field, but to the extent we restrict, or are restricted from, in-person interactions with physicians, we may be limited to conducting educational and promotional activities virtually, which may continue to hamper our ability to market INVELTYS and could adversely affect our ability to launch and market EYSUVIS. In addition, government restrictions have at times led to moratoria on elective ocular surgeries in many jurisdictions, which had significantly reduced, and may in the future continue to significantly reduce, the demand for INVELTYS, which is indicated for the treatment of post-operative inflammation and pain following ocular surgery. The extent of the impact of COVID-19 on our commercialization efforts will depend on the length and severity of this pandemic, including the extent there is any resurgence of the COVID-19 virus or any variant strains of the virus, the availability and effectiveness of vaccines, and the impact of the foregoing on our customers, employees and vendors and government agencies, which is uncertain and cannot be predicted.

We may never succeed in these activities and may never generate revenue that is sufficient to achieve profitability. Even if we do achieve profitability, we may not be able to sustain or increase profitability on a quarterly or annual basis. Our failure to become and remain profitable would decrease the value of our company and could impair our ability to raise capital, expand our business, maintain our research and development efforts, diversify our product offerings or even continue our operations. A decline in the value of our company could also cause you to lose all or part of your investment.

Our limited operating history as a commercial company may make it difficult for you to evaluate the success of our business to date and to assess our future viability.

We are an early-stage commercial company. Our operations to date have been limited to organizing and staffing our company, acquiring rights to intellectual property, business planning, raising capital, developing EYSUVIS and INVELTYS and conducting other research and development activities, and commercially launching EYSUVIS and INVELTYS. We are in the process of transitioning from a company solely with a research and development focus to a company engaging in commercial activities. We may not be successful in such a transition. We only launched INVELTYS in January 2019 and are still in the process of executing our commercial launch plan for EYSUVIS, have no prior history of commercializing products, and, to date, have generated limited revenue from the sale of EYSUVIS and INVELTYS. In addition, our commercial operations and INVELTYS sales have been and continue to be negatively

impacted by COVID-19 and its collateral consequences. The effects of COVID-19 may also disrupt the launch and commercialization of EYSUVIS. Consequently, any predictions you make about our future success or viability may not be as accurate as they could be if we had a longer operating and commercialization history.

We expect our financial condition and operating results to fluctuate significantly from quarter-to-quarter and year-to-year due to a variety of factors, many of which are beyond our control. Accordingly, you should not rely upon the results of any quarterly or annual periods as indications of future operating performance.

We may need substantial additional funding. If we are unable to raise capital when needed, we could be forced to delay, reduce or eliminate our product development programs or commercialization efforts.

We expect to devote substantial financial resources to our ongoing and planned activities, particularly as we commercialize EYSUVIS and INVELTYS, and as we advance our preclinical activities for our product candidates. We also expect to incur significant additional expenses if and as we conduct further research and development activities, and potentially initiate clinical trials of, and seek regulatory approval for, any product candidates that we identify and advance, including product candidates from our receptor Tyrosine Kinase Inhibitor, or rTKI, program, novel surface targeting steroid program and novel selective glucocorticoid receptor modulators, or SEGRM, program.

Our expenses have increased relative to prior periods in connection with our launch and commercialization of EYSUVIS and INVELTYS, including costs associated with the addition and subsequent expansion of our specialty sales force and increased marketing, distribution and manufacturing capabilities. For example, with the approval of EYSUVIS, we increased our sales force from 56 TSMs to 91 TSMs and from seven RSLs to 14 RSLs during the fourth quarter of 2020 and then to 105 TSMs around the start of the third quarter 2021, with a subsequent expansion up to 125 TSMs expected by year-end, subject to continued growth in payor coverage and the status of the COVID-19 pandemic. Accordingly, we may need to obtain substantial additional funding in connection with our continuing operations. If we are unable to raise capital when needed or on attractive terms, we could be forced to delay, reduce or eliminate our research and development programs or any current or future commercialization efforts.

Our future capital requirements will depend on many factors, including:

- the costs and timing of commercialization activities for EYSUVIS and INVELTYS, including the costs and timing of expanding our sales force and establishing additional product sales, marketing, medical affairs, distribution and outsourced manufacturing capabilities;
- our ability to successfully commercialize and sell EYSUVIS and INVELTYS in the United States and the amount of revenue received from commercial sales;
- the progress, costs and results of any clinical activities for regulatory review of, and our success seeking approval and/or commercializing, EYSUVIS and INVELTYS outside of the United States;
- our ability to establish and maintain strategic collaborations, licensing or other agreements and the financial terms of such agreements;
- the scope, progress, results and costs of any product candidates that we may develop;
- the extent to which we successfully advance and/or in-license or acquire rights to other products, product candidates or technologies; and
- the costs and timing of preparing, filing and prosecuting patent applications, maintaining and protecting our intellectual property rights and defending against any intellectual property-related claims.

We expect to continue to incur significant expenses and operating losses. Net losses may fluctuate significantly from quarter-to-quarter and year-to-year. We expect that our cash, cash equivalents, and short-term investments of

\$149.6 million as of June 30, 2021, along with anticipated revenue from EYSUVIS and INVELTYS, will enable us to fund our operations, lease and debt service obligations, and capital expenditure requirements for at least two years. We have based this estimate on assumptions that may prove to be wrong, and our operating plan may change as a result of many factors currently unknown to us. As a result, we could deplete our available capital resources sooner than we currently expect.

Commercializing products is a time-consuming, expensive and uncertain process. Although we commercially launched INVELTYS in early 2019, began shipping EYSUVIS to wholesalers in the United States in late December 2020 and commenced a full promotional launch of EYSUVIS in early January 2021, our revenue from product sales of EYSUVIS and INVELTYS may not be sufficient for us to become profitable in the near future, if at all. In addition, other than our approved products, EYSUVIS and INVELTYS, all of our other development efforts are in the early stages of preclinical development. Identifying potential product candidates and conducting preclinical testing and clinical trials is a time-consuming, expensive and uncertain process that takes years to complete. Completion dates and completion costs can vary significantly for each product candidate and are difficult to predict. We may never generate the necessary data or results required to obtain marketing approval and achieve product sales from our preclinical development programs. Also, even if we successfully identify and develop product candidates from our preclinical development programs and those are approved, we may not achieve commercial success with them. Accordingly, we will need to rely on the commercial success of EYSUVIS and INVELTYS to generate product revenue for the foreseeable future.

We may require additional financing to achieve our business objectives. In addition, we may opportunistically raise additional capital due to favorable market conditions or strategic considerations, even if we believe we have sufficient funds for our current or future operating plans. Adequate additional financing may not be available to us on acceptable terms, or at all. If adequate funds are not available to us on a timely basis, we may be required to delay, limit, reduce or terminate preclinical studies, clinical trials or other development activities for one or more of our product candidates or delay, limit, reduce or terminate our establishment of sales and marketing capabilities or other activities that may be necessary to commercialize EYSUVIS and INVELTYS, or any product candidates for which we obtain approval.

Raising additional capital may cause dilution to our stockholders, restrict our operations or require us to relinquish rights to our technologies or product candidates.

Until such time, if ever, as we can generate substantial product revenues, we expect to finance our cash needs through a combination of equity offerings, debt financings, collaborations, strategic alliances, licensing arrangements, royalty agreements, and marketing and distribution arrangements. To the extent that we raise additional capital through the sale of equity or convertible debt securities, your ownership interest will be diluted, and the terms of these securities may include liquidation or other preferences that adversely affect your rights as a common stockholder. Debt financing and preferred equity financing, if available, may involve agreements that include pledging of assets as collateral, covenants limiting or restricting our ability to take specific actions, such as incurring additional debt, making capital expenditures or declaring dividends. Our pledge of our assets as collateral to secure our obligations under our Loan Agreement may limit our ability to obtain additional debt financing. Under the Loan Agreement, we are also restricted from paying dividends on our common stock, granting liens, making investments, making acquisitions, making certain restricted payments, selling assets and making certain other uses of our cash without the lenders' consent, subject in each case to certain exceptions.

If we raise additional funds through collaborations, strategic alliances, licensing arrangements, royalty agreements, or marketing and distribution arrangements, we may have to relinquish valuable rights to our technologies, future revenue streams, research programs or product candidates or grant licenses on terms that may not be favorable to us. If we are unable to raise additional funds through equity or debt financings when needed, we may be required to delay, limit, reduce or terminate our product development or current or future commercialization efforts or grant rights to develop and market products or product candidates that we would otherwise prefer to develop and market ourselves.

# Our substantial indebtedness may limit cash flow available to invest in the ongoing needs of our business.

We have a substantial amount of indebtedness. As of June 30, 2021, we had \$80.0 million of outstanding borrowings under the tranche A term loan under the Loan Agreement, bearing interest at a floating rate equal to the greater of 30-day LIBOR and 0.11%, plus 7.89%. The Loan Agreement provides for interest-only payments until December 1, 2024 if neither the tranche B term loan nor the tranche C term loan are made, and until June 1, 2025 if either the tranche B term loan or the tranche C term loan is made, or the Amortization Date. Beginning on the Amortization Date, we are required to repay the outstanding principal in monthly installments over a period of (i) 18 months if neither the tranche B term loan nor the tranche C term loan is made or (ii) 12 months if either the tranche B term loan or the tranche C term loan is made. All unpaid principal and interest is due in full on May 1, 2026, the date of maturity.

Our obligations under the Loan Agreement are secured by substantially all of our assets. We could in the future incur additional indebtedness beyond our borrowings under our Loan Agreement.

Our debt combined with our other financial obligations and contractual commitments could have significant adverse consequences, including:

- requiring us to dedicate a substantial portion of cash flow from operations or cash on hand to the payment of
  interest on, and principal of, our debt, which will reduce the amounts available to fund working capital,
  capital expenditures, product development efforts and other general corporate purposes;
- increasing our vulnerability to adverse changes in general economic, industry and market conditions;
- subjecting us to restrictive covenants that may reduce our ability to take certain corporate actions or obtain further debt or equity financing;
- limiting our flexibility in planning for, or reacting to, changes in our business and our industry; and
- placing us at a competitive disadvantage compared to our competitors that have less debt or better debt servicing options.

We intend to satisfy our current and future debt service obligations with our existing cash and anticipated product revenue. Nonetheless, we may not have sufficient funds or may be unable to arrange for additional financing to pay the amounts due under our existing debt and funds from external sources may not be available on acceptable terms, if at all. In addition, a failure to comply with the covenants under our Loan Agreement could result in an event of default and acceleration of amounts due. If an event of default occurs and the lender accelerates the amounts due under our Loan Agreement, we may not be able to make accelerated payments, and the lender could seek to enforce security interests in the collateral securing such indebtedness.

## Fluctuations in interest rates could materially affect the interest expense on our Loan Agreement.

Because our debt under the Loan Agreement bears interest at floating interest rates, increases in interest rates could materially increase our interest expense.

Further, our Loan Agreement uses LIBOR as a reference rate. On July 27, 2017, the United Kingdom's Financial Conduct Authority, or the FCA, which regulates LIBOR, announced that it intends to stop encouraging or compelling banks to submit rates for the calibration of LIBOR by the end of 2021. On November 30, 2020, ICE Benchmark Administration, the administrator of LIBOR, with the support of the United States Federal Reserve and the FCA, announced plans to consult on ceasing publication of LIBOR on December 31, 2021 for only the one week and two month LIBOR tenors, and on June 30, 2023 for all other LIBOR tenors. While this announcement extends the transition period to June 2023, the United States Federal Reserve concurrently issued a statement advising banks to stop new LIBOR issuances by the end of 2021. In light of these recent announcements, the future of LIBOR at this time is uncertain and any changes in the methods by which LIBOR is determined or regulatory activity related to LIBOR's

phaseout could cause LIBOR to perform differently than in the past or cease to exist. In June 2017, the Alternative Reference Rates Committee selected the Secured Overnight Financing Rate, or SOFR, a new index calculated by reference to short-term repurchase agreements backed by Treasury securities, as its preferred replacement for U.S. dollar LIBOR. Whether or not SOFR attains market acceptance as a LIBOR replacement tool remains in question. As such, the future of LIBOR and the potential alternatives at this time is uncertain. If the method for calculation of LIBOR changes, if LIBOR is no longer available or if lenders have increased costs due to changes in LIBOR or changes in law, we may suffer from potential increases in interest rate costs on our floating debt rate. Further, we may need to renegotiate our Loan Agreement and the floating loans thereunder to replace the interest rate calculated by reference to LIBOR with an interest rate calculated by reference to a new standard that is established.

If our estimates or judgments relating to our critical accounting policies, or any of our projections, prove to be inaccurate or financial reporting standards or interpretations change, our results of operations could be adversely affected.

The preparation of financial statements in conformity with generally accepted accounting principles in the United States requires management to make estimates and assumptions that affect the amounts reported in the consolidated financial statements and accompanying notes. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amounts of our assets, liabilities, revenues and expenses. Such estimates and judgments include revenue recognition, inventory, the present value of lease liabilities and the corresponding right-of-use assets, the fair value of warrants, stock-based compensation, accrued expenses and the recoverability of our net deferred tax assets and related valuation allowance. We base our estimates and judgments on historical experience, expected future experience and on various other assumptions that we believe to be reasonable under the circumstances. In addition, from time to time, we may rely on projections regarding our expected future performance that represent our management's thencurrent estimates. However, any of these estimates, judgments or projections, or the assumptions underlying them, may change over time or may otherwise prove to be inaccurate. Our results of operations may be adversely affected if our estimates, assumptions or projections change or if actual circumstances differ from those in our estimates or assumptions, which could cause our results of operations to fall below the expectations of securities analysts and investors, resulting in a decline in the trading price of our common stock.

For example, we rely on third-party data providers to collect and report estimates of prescription information and pipeline inventory levels as components of our estimations for revenue recognition. There is a limited amount of information available to such data providers to determine the actual number of total prescriptions for prescription products during such periods. Their estimates are based on a combination of data received from pharmacies and other distributors, and historical data when actual data is unavailable. Their calculations of changes in prescription levels between periods can be significantly affected by lags in data reporting from various sources or by changes in pharmacies and other distributors providing data. Such methods can from time to time result in significant inaccuracies in information when ultimately compared with actual results. Further, data for a single and limited period may not be representative of a trend or otherwise predictive of future results.

Additionally, we regularly monitor our compliance with applicable financial reporting standards and review new pronouncements and drafts thereof that are relevant to us. As a result of new standards, changes to existing standards and changes in their interpretation, we might be required to change our accounting policies, alter our operational policies and implement new or enhance existing systems so that they reflect new or amended financial reporting standards, or we may be required to restate our published financial statements. Such changes to existing standards or changes in their interpretation may have an adverse effect on our reputation, business, financial position, and profit.

# Risks Related to the Commercialization of EYSUVIS, INVELTYS and our Product Candidates

The ongoing novel coronavirus pandemic and the efforts to prevent its spread have adversely impacted our operations and the market for INVELTYS, could impact the launch and commercialization of EYSUVIS and may continue to adversely affect our business, results of operations and financial condition.

The outbreak of the COVID-19 pandemic and government measures taken in response to it, including from time to time quarantines, strict travel restrictions and bans, heightened border scrutiny and other measures, have had a significant impact, both direct and indirect, on businesses and commerce; supply chains have been disrupted; facilities and production have been suspended; and demand for certain goods and services, such as medical services and supplies, has spiked, while demand for other goods and services has fallen significantly.

In particular, from time to time moratoria have been put in place on routine medical appointments and elective surgeries in many jurisdictions, including ocular surgeries, which have adversely affected, and may adversely affect in the future, the market for INVELTYS, which is indicated for the treatment of inflammation and pain following ocular surgery, resulting in a significant reduction in the demand for INVELTYS. The COVID-19 pandemic has negatively impacted revenues from INVELTYS and we expect it to continue to do so until surgeries return to and remain at historical levels. In light of shelter-in-place orders and other mandated local travel and social interaction prohibitions, we previously suspended substantially all in-person interactions with physicians and were limited to conducting educational and promotional activities virtually. Our sales force has since resumed substantially all in-person interactions in the field, but to the extent we restrict, or are restricted from, in-person interactions with physicians, we may be limited to conducting educational and promotional activities virtually, which may continue to hamper our ability to market INVELTYS and could adversely affect our ability to launch and market EYSUVIS. Furthermore, while the majority of our day-to-day operations are continuing as our employees are permitted to work remotely, our laboratory facilities that support our early-stage research activities were partially limited, and may be limited again in the future, as a result of COVID-19.

Additionally, while we currently are not experiencing interruptions in our manufacturing of EYSUVIS or INVELTYS, any reinstatement of quarantines, travel restrictions and other measures may significantly impact the ability of employees of our third-party suppliers to get to their places of work to manufacture and deliver future supplies if and when needed.

The COVID-19 pandemic has already caused significant disruptions in the financial markets, and may again cause such disruptions, which could impact our ability to raise additional funds through public offerings and may also impact the volatility of our stock price and trading in our stock. Moreover, the significant ongoing impact of the pandemic on economies worldwide could result in more extensive adverse effects on our business and operations. The full extent of the impact of COVID-19 on our commercialization efforts will depend on the length and severity of this pandemic, the timing and extent of any resurgence of the COVID-19 virus or any variant strains of the virus, the availability and effectiveness of vaccines, and the impact of the foregoing on our customers, employees and vendors and government agencies, which is uncertain and cannot be predicted. We cannot be certain what the overall impact of the COVID-19 pandemic will be on our business and it has the potential to significantly and adversely affect our business, financial condition, results of operations and prospects.

EYSUVIS, INVELTYS or any of our product candidates that receive marketing approval may fail to achieve market acceptance by clinicians and patients, or adequate formulary coverage, pricing or reimbursement by third-party payors and others in the medical community, and the market opportunity for these products may be smaller than we estimate.

EYSUVIS, INVELTYS or any product candidate that we develop that receives marketing approval may fail to gain sufficient market acceptance by clinicians, patients, third-party payors and others in the medical community. While there are no drugs other than EYSUVIS currently approved in the United States for the short-term treatment of the signs and symptoms of dry eye disease, current treatments that are used in the United States for dry eye disease include over-the-counter artificial tears, Restasis®, Xiidra®, Cequa<sup>TM</sup>, off-label use of corticosteroids and various drugs that are produced by compounding pharmacies. Generic versions of Restasis are also expected to become available in the United

States in the near future. Our current expectations regarding market potential for EYSUVIS are based, in part, on market research data we have commissioned, which indicated that interest in prescribing EYSUVIS is high among surveyed eye care professionals, or ECPs. However, it is possible that ECPs may continue to rely on other existing treatments rather than EYSUVIS. In addition, generic versions of any products that compete with any of our products or product candidates would likely be offered at a substantially lower price than we offer our products and expect to offer for our product candidates, if approved. As a result, clinicians, patients and third-party payors may choose to rely on such products rather than our products or product candidates.

Common treatments in the United States for inflammation and pain following ocular surgery include corticosteroids. Our current estimates of potential future revenue from sales of INVELTYS are based, in part, on market research data we have commissioned, which indicated that a majority of surveyed ophthalmologists were likely to prescribe INVELTYS. However, doctors may continue to rely on ocular steroids other than INVELTYS and other treatments rather than INVELTYS. In addition, there are also non-topical formulations of ocular steroids that are available to patients. It is also possible that other therapeutics will be approved for treatment of inflammation and pain following ocular surgery with twice a day or less frequent dosing.

The market opportunity for EYSUVIS and INVELTYS may be further impacted by extraordinary events such as the current pandemic health event resulting from COVID-19 and its collateral consequences. For example, from time to time moratoria have been put in place on routine medical appointments and elective surgeries in many jurisdictions, including ocular surgeries such as cataract and refractive, which have adversely affected, and may adversely affect in the future, the market for INVELTYS, which is indicated for the treatment of post-operative inflammation and pain following ocular surgery, resulting in a significant reduction in the demand for INVELTYS. In light of shelter-in-place orders and other mandated local travel prohibitions, we previously suspended substantially all in-person interactions with physicians and were limited to conducting educational and promotional activities virtually. Our sales force has since resumed substantially all in-person interactions in the field, but to the extent we restrict, or are restricted from, in-person interactions with physicians, we may be limited to conducting educational and promotional activities virtually, which may continue to hamper our ability to market INVELTYS and could adversely affect our ability to successfully launch and market EYSUVIS.

Our assessment of the potential market opportunity for EYSUVIS, INVELTYS and our product candidates is based on industry and market data that we obtained from industry publications and research, surveys and studies conducted by third parties, some of which we commissioned. Industry publications and third-party research, surveys and studies generally indicate that their information has been obtained from sources believed to be reliable, although they do not guarantee the accuracy or completeness of such information. While we believe these industry publications and third-party research, surveys and studies are reliable, we have not independently verified such data. The potential market opportunity for the treatment of dry eye disease in particular is difficult to precisely estimate. The results from our physician and patient surveys may be less reflective of the dry eye disease population as a whole than a survey conducted with a larger sample size. Our estimates of the potential market opportunities for our product candidates include several key assumptions based on our industry knowledge, industry publications, third-party research and other surveys, which may be based on a small sample size and fail to accurately reflect market opportunities. While we believe that our internal assumptions are reasonable, no independent source has verified such assumptions. If any of our assumptions or estimates, or these publications, research, surveys or studies prove to be inaccurate, then the actual market for EYSUVIS, INVELTYS or any of our product candidates may be smaller than we expect, and as a result our product revenue may be limited and it may be more difficult for us to achieve or maintain profitability. The uncertainty with respect to the future progression of the COVID-19 pandemic and its long-term effects may adversely impact the accuracy of such estimates and our potential market opportunity for EYSUVIS and INVELTYS.

If EYSUVIS, INVELTYS or any of our product candidates for which we obtain marketing approval do not achieve adequate levels of acceptance, formulary coverage, pricing or reimbursement, we may not generate significant product revenues and we may not become profitable. The degree of market acceptance of EYSUVIS, INVELTYS or any product candidates for which we obtain marketing approval, will depend on a number of factors, including:

 the efficacy and potential advantages of our product or our product candidates compared to alternative treatments, including the existing standard of care;

- our ability to offer our products for sale at competitive prices, particularly in light of the lower cost of alternative treatments;
- the availability of third-party formulary coverage and adequate reimbursement, particularly by Medicare in light of the prevalence of dry eye disease and cataracts in persons over age 55;
- the clinical indications for which the product is approved;
- the convenience and ease of administration compared to alternative treatments;
- the willingness of the target patient population to try new therapies and of clinicians to prescribe these therapies;
- the strength of our marketing and distribution support;
- the timing of market introduction of competitive products;
- the prevalence and severity of any side effects; and
- any restrictions on the use of our products together with other medications.

Even if we are able to successfully commercialize EYSUVIS, INVELTYS or any product candidate that we may develop, the products may become subject to unfavorable pricing regulations, third-party coverage or reimbursement practices or healthcare reform initiatives, which could harm our business.

Our ability to successfully commercialize EYSUVIS, INVELTYS or any of our product candidates that we may develop successfully will depend, in part, on the extent to which coverage and adequate reimbursement for these products and related treatments will be available from government healthcare programs, private health insurers, managed care plans and other organizations. Government authorities and third-party payors, such as private health insurers and health maintenance organizations, decide which medications they will pay for and establish reimbursement levels. A primary trend in the U.S. healthcare industry and elsewhere is cost containment. Government authorities and third-party payors have attempted to control costs by limiting coverage and the amount of reimbursement for particular medications. Increasingly, third-party payors are requiring that drug companies provide them with predetermined discounts from list prices and are challenging the prices charged for medical products. Coverage and reimbursement may not be available for EYSUVIS, INVELTYS or any product candidate that we commercialize and, even if they are available, the level of reimbursement may be limited or not satisfactory.

Inadequate reimbursement may adversely affect the demand for, or the price of, EYSUVIS, INVELTYS or any product candidate for which we obtain marketing approval. Obtaining and maintaining adequate reimbursement for our products may be difficult. We may be required to conduct expensive pharmacoeconomic studies to justify coverage and reimbursement or the level of reimbursement relative to other therapies. If coverage and adequate reimbursement are not available or reimbursement is available only to limited levels, we may not be able to successfully commercialize EYSUVIS, INVELTYS or any product candidate for which we obtain marketing approval.

There may be significant delays in obtaining coverage and reimbursement for newly approved drugs, such as EYSUVIS, and coverage may be more limited than the indications for which the drug is approved by the FDA or similar regulatory authorities outside the United States. Moreover, eligibility for coverage and reimbursement does not imply that a drug will be paid for in all cases or at a rate that covers our costs, including research, development, manufacture, sale and distribution expenses. Interim reimbursement levels for new drugs, if applicable, may also not be sufficient to cover our costs and may not be made permanent. Reimbursement rates may vary according to the use of the drug and the clinical setting in which it is used, may be based on reimbursement levels already set for lower cost drugs and may be incorporated into existing payments for other services. Net prices for drugs may be reduced by mandatory discounts or rebates required by government healthcare programs or private payors and by any future relaxation of laws that presently

restrict imports of drugs from countries where they may be sold at lower prices than in the United States. Third-party payors often rely upon Medicare coverage policy and payment limitations in setting their own reimbursement policies. Our inability to promptly obtain coverage and adequate reimbursement rates from both government-funded and private payors for any approved products that we develop would compromise our ability to generate revenues and become profitable.

The regulations that govern marketing approvals, pricing, coverage and reimbursement for new drug products vary widely from country to country. Current and future legislation may significantly change the approval requirements in ways that could involve additional costs and cause delays in obtaining approvals. Some countries require approval of the sale price of a drug before it can be marketed. In many countries, the pricing review period begins after marketing or product licensing approval is granted. In some foreign markets, prescription pharmaceutical pricing remains subject to continuing governmental control even after initial approval is granted. As a result, we might obtain marketing approval for a product in a particular country, but then be subject to price regulations that delay our commercial launch of the product, possibly for lengthy time periods, and negatively impact the revenues we are able to generate from the sale of the product in that country. To obtain reimbursement or pricing approval in some countries, we may be required to conduct a clinical trial that compares the cost-effectiveness of our product candidate to other available therapies. Adverse pricing limitations may hinder our ability to recoup our investment in one or more product candidates, even if our product candidates obtain marketing approval.

There can be no assurance that EYSUVIS, INVELTYS or our product candidates, even if such product candidates are approved for sale in the United States or in other countries, will be considered medically reasonable and necessary for a specific indication or cost-effective by third-party payors, or that coverage and an adequate level of reimbursement will be available or that third-party payors' reimbursement policies will not adversely affect our ability to sell EYSUVIS, INVELTYS or our product candidates profitably.

If we are unable to maintain our sales, marketing and distribution capabilities, establish additional capabilities if and when necessary, or enter into sales, marketing and distribution agreements with third parties, we may not be successful in commercializing EYSUVIS, INVELTYS or any of our product candidates that we may develop if and when they are approved.

We established our sales and marketing infrastructure for the commercial launch of INVELTYS, our first product, and EYSUVIS, and, as a company, we have limited experience in the sales, marketing and distribution of therapeutic products. To achieve commercial success for any product for which we obtained marketing approval, we may need to establish additional sales, marketing and distribution capabilities, either ourselves or through collaborations or other arrangements with third parties.

In 2019, we completed the initial buildout of our specialty sales, marketing and distribution infrastructure in the United States to commercialize INVELTYS, which included a sales force of 57 TSMs, seven RSLs, and three directors of national accounts. During the fourth quarter of 2020, we expanded our sales force to include 91 TSMs, 14 RSLs, and two new area sales leaders. In 2021, increased our sales force from 91 TSMs to 105 TSMs by the start of the third quarter, with a subsequent expansion up to 125 TSMs expected by year-end subject to continued growth in payor coverage and the status of the COVID-19 pandemic. Our sales representatives promote both EYSUVIS and INVELTYS. There are risks involved with establishing, maintaining and expanding our own sales, marketing and distribution capabilities. For example, recruiting and training a sales force is expensive and time-consuming and could delay any future product launch. Further, we may underestimate the size of the sales force required for a successful product launch, including with respect to the launch of EYSUVIS, and may need to expand our sales force earlier and at a higher cost than we anticipated. If the commercial launch of any of our product candidates for which we establish additional commercial infrastructure is delayed or does not occur for any reason, we would have prematurely or unnecessarily incurred these commercialization expenses. This may be costly, and our investment would be lost if we cannot retain or reposition our sales and marketing personnel.

Factors that may inhibit our efforts to commercialize EYSUVIS, INVELTYS or any product candidates for which we receive marketing approval on our own include:

- our inability to recruit, train and retain adequate numbers of effective sales and marketing personnel;
- our inability to obtain and maintain coverage, adequate pricing, and adequate reimbursement from third-party payors, including government payors;
- the inability of sales personnel to obtain access to clinicians, including as a result of limitation on office visits as a result of COVID-19 or other health concerns, or persuade adequate numbers of clinicians to prescribe our products;
- the lack of complementary products to be offered by sales personnel, which may put us at a competitive disadvantage relative to companies with more extensive product lines; and
- unforeseen costs and expenses associated with maintaining and expanding an independent sales, marketing and distribution organization.

While we cannot be certain when, if ever, we will seek and/or receive marketing approval to commercialize any of our product candidates outside the United States, we may seek marketing approval and explore commercialization of EYSUVIS in certain markets outside the United States, including the European Union, utilizing a variety of collaboration, distribution and other marketing arrangements with one or more third parties. Our product revenues and our profitability, if any, under any such third-party collaboration, distribution or other marketing arrangements are likely to be lower than if we were to market, sell and distribute EYSUVIS ourselves. We may also consider seeking marketing approval outside the United States for other product candidates in the future. If we decide to seek regulatory approval for any of our product candidates outside the United States, we may need to seek additional patent approvals, seek licenses to patents held by third parties and/or face claims of infringing third-party patent rights.

In addition, we may not be successful in entering into arrangements with third parties to sell, market and distribute EYSUVIS, INVELTYS or any of our product candidates or we may be unable to do so on terms that are favorable to us. We likely will have little control over such third parties, and any of them may fail to devote the necessary resources and attention to sell and market effectively EYSUVIS, INVELTYS or any of our product candidates for which we obtain marketing approval. If we do not maintain our sales, marketing and distribution capabilities successfully, or do not establish additional capabilities if and when needed successfully, either on our own or in collaboration with third parties, we will not be successful in commercializing EYSUVIS, INVELTYS or any of our product candidates for which we obtain marketing approval.

We face substantial competition, which may result in others discovering, developing or commercializing products before or more successfully than we do. Our competitors include major pharmaceutical companies with significantly greater financial resources. EYSUVIS, INVELTYS and our product candidates will also compete with existing branded, generic and off-label products.

The development and commercialization of new drug products is highly competitive. We face competition with respect to EYSUVIS, INVELTYS and will face competition with respect to any product candidates that we may seek to develop or commercialize in the future, from major pharmaceutical companies, specialty pharmaceutical companies and biotechnology companies worldwide. Potential competitors also include academic institutions, government agencies and other public and private research organizations that conduct research, seek patent protection and establish collaborative arrangements for research, development, manufacturing and commercialization.

Our products and product candidates target markets that are already served by a variety of competing products. Many of these existing products have achieved widespread acceptance among clinicians, patients and payors. In addition, many of these products are available on a generic basis, and our products or our product candidates may not demonstrate sufficient additional clinical benefits to clinicians, patients or payors to justify a higher price compared to generic products. In many cases, insurers or other third-party payors, particularly Medicare, seek to encourage the use of generic products.

The current disease management approaches for dry eye disease in the United States includes non-pharmaceutical therapies and pharmaceutical therapies. Non-pharmaceutical therapies include over the counter artificial tear eye drops, which are palliative and used on an intermittent or chronic basis to provide short-term symptomatic relief of dryness and irritation; hot compresses for the eye and lid hygiene management; and devices, such as punctal plugs that are inserted into the tear ducts to inhibit tear drainage, resulting in more moisture on the surface of the eye.

Pharmaceutical therapies for dry eye disease include on label prescription drugs, including Restasis, Xiidra, and Cequa, which are the only prescription pharmaceutical products other than EYSUVIS that are approved in the United States for use in patients with dry eye disease; and off label prescription drugs, including topical steroid drops and/or other similar products, which are sometimes prescribed for treatment of dry eye disease. Generic versions of Restasis are expected to become available in the United States in the near future. Restasis and Cequa are both topical cyclosporine formulations that are approved for increasing tear production in patients whose tear production is presumed to be suppressed due to ocular keratoconjunctivitis sicca. Xiidra is a topical anti-inflammatory therapy approved for treatment of the signs and symptoms of dry eye disease.

EYSUVIS is indicated for the short-term (up to two weeks) treatment of the signs and symptoms of dry eye disease, which includes dry eye flares. Any product that is developed for the treatment of the signs and/or symptoms of dry eye disease could directly compete with EYSUVIS. There are several product candidates in preclinical and clinical development in the United States for the treatment of dry eye disease. If any of these product candidates is approved and such product candidate either treats the signs and/or symptoms of dry eye disease or reduces the frequency of flares in dry eye patients, it could reduce the overall market opportunity for EYSUVIS. These product candidates are being developed by pharmaceutical, biotechnology, specialty pharmaceutical and generic drug companies of various sizes, such as Oyster Point Pharma's OC-01 nasal spray, for which an NDA was submitted in December 2020 and, if approved, could be launched as early as late 2021, Aldeyra Therapeutics' reproxalap ophthalmic solution, Novaliq's CyclAsol and NOV03, which have been licensed to Bausch Health Companies Inc., and others.

Following ocular surgery, topical steroids are commonly prescribed to manage and prevent complications from post-operative inflammation. Topical steroid drops are the main competition to INVELTYS for the treatment of inflammation and pain following ocular surgery. The current branded market leaders for topical steroids in the United States, based on revenue, are Lotemax® products and Durezol®. Generic topical steroid formulations consist mainly of products containing prednisolone, fluorometholone or dexamethasone. In addition, the first generic formulations of loteprednol suspension 0.5% (Lotemax suspension) and loteprednol ophthalmic gel 0.5% (Lotemax Gel) were launched in May 2019 and February 2021, respectively, and Durezol lost its patent exclusivity in 2019, which could result in a potential generic launch of this product at any time.

There are also non-topical formulations of ocular steroids that have been recently approved and/or marketed. Eyepoint Pharmaceutical launched Dexycu®, an intraocular suspension of dexamethasone for the treatment of post-operative inflammation, in July 2019. Also in July 2019, Ocular Therapeutix launched Dextenza®, an intracanalicular insert of dexamethasone, for the treatment of ocular pain following ophthalmic surgery. There are also a number of companies in the United States developing products and therapies in preclinical research and clinical development for the treatment of inflammation and pain following ocular surgery. In addition, there are various formulations of steroids that are produced by compounding pharmacies and that are in drop form or are injected into the eye following ocular surgery.

Our commercial opportunity could be reduced or eliminated if our competitors develop and commercialize products that are safer, more effective, have fewer or less severe side effects, are more convenient or are less expensive than our products. Our competitors also may obtain FDA or other regulatory approval for their products more rapidly than we may obtain approval for ours, which could result in our competitors establishing a strong market position before we are able to enter the market.

In addition, our ability to compete may be affected in many cases by insurers or other third-party payors, particularly Medicare, seeking to encourage the use of generic products. Generic products are currently being used for certain of the indications that we are pursuing, and additional products are expected to become available on a generic basis over the coming years.

Many of the companies against which we are competing or which we may compete against in the future have significantly greater financial resources and expertise in research and development, manufacturing, preclinical testing, conducting clinical trials, obtaining regulatory approvals and marketing approved products than we do. Mergers and acquisitions in the pharmaceutical and biotechnology industries may result in even more resources being concentrated among a smaller number of our competitors. Smaller and other early stage companies may also prove to be significant competitors, particularly through collaborative arrangements with large and established companies. These third parties compete with us in recruiting and retaining qualified scientific and management personnel, establishing clinical trial sites and patient registration for clinical trials, as well as in acquiring technologies complementary to, or necessary for, our programs. Given that EYSUVIS and INVELTYS utilize a known FDA-approved corticosteroid, these products and any similar product candidates, if approved, may face competition from generic and branded versions of existing drugs based on corticosteroids that are administered in a different manner.

If our contracted manufacturing facilities experience production issues for any reason, we may be unable to manufacture commercial quantities of our products or product candidates for a substantial amount of time, which could have a material adverse effect on our business.

We rely on third-party contract manufacturers to manufacture commercial supplies of EYSUVIS and INVELTYS. Specifically, we rely on the following: Woodstock Sterile Solutions, Inc. (formerly known as Catalent Pharma Solutions, LLC), or Woodstock, to manufacture and supply to us a minimum amount of EYSUVIS and INVELTYS bottles for commercial use; Altasciences company, or Altasciences, for manufacturing bulk intermediates; and Chemo Iberica SA, or Chemo Iberica, to manufacture and supply to us a bulk supply of loteprednol etabonate, or LE. We expect to rely on third parties to manufacture clinical supplies of any other product candidates and commercial supplies of any other products, if and when approved for marketing by applicable regulatory authorities, as well as for packaging, serialization, storage, distribution and other production logistics. If these third parties do not successfully carry out their contractual duties, meet expected deadlines or manufacture our products or our product candidates in accordance with regulatory requirements, if there are disagreements between us and such parties, or if such parties are unable to expand capacities to support commercialization of our products or any of our product candidates for which we obtain marketing approval, we may not be able to compete, or may be delayed in producing sufficient product or product candidates to meet our supply requirements. These facilities, and suppliers to these facilities, may also be affected by natural disasters, such as floods or fire, epidemics or pandemics, such as COVID-19, or such facilities could face manufacturing issues, such as contamination or regulatory concerns following a regulatory inspection of such facility. In such instances, we may need to locate an appropriate replacement third-party relationship, which may not be readily available or on acceptable terms, or at all, which would cause additional delay and increased expense, including as a result of additional required FDA approvals, and may have a material adverse effect on our business.

Our third-party manufacturers are subject to inspection and approval by the FDA before we can commence the manufacture and sale of any of our products or product candidates, and thereafter subject to FDA inspection from time to time. Failure by our third-party manufacturers to pass such inspections and otherwise satisfactorily complete the FDA approval regimen with respect to our products or product candidates may result in regulatory actions such as the issuance of FDA Form 483 notices of observations, warning letters or injunctions or the loss of operating licenses. Depending on the severity of any potential regulatory action, our clinical or commercial supply could be interrupted or limited, which could have a material adverse effect on our business.

We or our third-party manufacturers may also encounter shortages in the raw materials or active pharmaceutical ingredient, or API, necessary to produce our product candidates in the quantities needed for our clinical trials, or our products or our product candidates, if approved, in sufficient quantities for commercialization or to meet an increase in demand, as a result of capacity constraints or delays or disruptions in the market for the raw materials or API, including shortages caused by the purchase of such raw materials or API by our competitors or others and shortages related to epidemics or pandemics, such as the COVID-19 pandemic. The failure of us or our third-party manufacturers to obtain the raw materials or API necessary to manufacture sufficient quantities of our products or product candidates, may have a material adverse effect on our business.

Product liability lawsuits against us could divert our resources and could cause us to incur substantial liabilities and limit commercialization of EYSUVIS, INVELTYS and any other products that we may develop.

We face an inherent risk of product liability exposure related to the use of our product candidates that we develop in human clinical trials. We face an even greater risk as we commercialize EYSUVIS, INVELTYS or any other products that we may develop. If we cannot successfully defend ourselves against claims that our product candidates or products caused injuries, we will incur substantial liabilities. Regardless of merit or eventual outcome, liability claims may result in:

- decreased demand for EYSUVIS, INVELTYS and any other products that we may develop;
- injury to our reputation and significant negative media attention;
- withdrawal of clinical trial participants;
- significant costs to defend the related litigation;
- substantial monetary awards to trial participants or patients;
- loss of revenue;
- reduced time and attention of our management to pursue our business strategy; and
- the inability to successfully commercialize EYSUVIS, INVELTYS and any other products that we may develop.

We currently hold \$15 million in product liability insurance coverage in the aggregate, with a per incident limit of \$15 million, which may not be adequate to cover all liabilities that we may incur. We may need to increase our insurance coverage if and as we commence commercialization of EYSUVIS or any product candidates for which we obtain marketing approval. Insurance coverage is increasingly expensive. We may not be able to maintain insurance coverage at a reasonable cost or in an amount adequate to satisfy any liability that may arise.

### **Risks Related to Product Development**

We are dependent on the success of EYSUVIS, INVELTYS and any product candidates for which we receive marketing approval. If we are unable to successfully commercialize our products and product candidates, our business will be materially harmed.

We have devoted a significant portion of our financial resources and business efforts to the development of INVELTYS for the post-operative treatment of inflammation and pain following ocular surgery and EYSUVIS for the short-term (up to two weeks) treatment of the signs and symptoms of dry eye disease. There is a significant risk that we will fail to successfully commercialize EYSUVIS and INVELTYS. Our ability to generate meaningful product revenues will depend on our successful commercialization of EYSUVIS and INVELTYS.

The success of our products EYSUVIS and INVELTYS and any product candidates for which we receive marketing approval will depend on many factors, including the following:

- successful commercialization of EYSUVIS and INVELTYS in the United States, including maintaining sales, marketing, manufacturing and distribution capabilities for EYSUVIS and INVELTYS;
- acceptance of EYSUVIS and INVELTYS and any product candidates we develop by patients, the medical community and third-party payors;

### **Table of Contents**

- obtaining and maintaining coverage, adequate pricing, and adequate reimbursement from third-party payors, including government payors;
- successfully developing and applying for and receiving marketing approvals from applicable regulatory authorities for any product candidates;
- maintaining regulatory approval of our manufacturing processes and our third-party manufacturers' facilities from applicable regulatory authorities and maintaining adequate supply of our products;
- maintaining a workforce of experienced scientists and others with experience in eye diseases to continue to develop our product candidates;
- leveraging our sales, marketing and distribution capabilities for our current products and expanding upon these capabilities if and when appropriate;
- establishing additional sales, marketing and distribution capabilities for, and successfully launching commercial sales of any other product candidates for which we obtain marketing approval, whether alone or in collaboration with others;
- effectively competing with other therapies;
- maintaining an acceptable safety profile of our products following approval;
- obtaining and maintaining patent and trade secret protection and regulatory exclusivity for our product candidates;
- protecting our rights in our intellectual property portfolio; and
- not infringing, misappropriating or otherwise violating others' intellectual property rights.

If we do not achieve one or more of these factors in a timely manner or at all, we could experience significant delays or an inability to successfully commercialize EYSUVIS, INVELTYS or our product candidates, which would materially harm our business. In addition, other than our approved products, EYSUVIS and INVELTYS, all of our other development efforts are in the early stages of preclinical development. We may never identify any product candidates or advance any product candidates to clinical-stage development from these preclinical development programs. Therefore, our ability to generate product revenue will depend heavily on the successful commercialization of EYSUVIS and INVELTYS, as the development and eventual commercialization of product candidates from our preclinical development programs may never occur.

If clinical trials of any product candidate that we develop fail to demonstrate safety and efficacy to the satisfaction of the FDA or other regulatory authorities or do not otherwise produce favorable results, we may incur additional costs or experience delays in completing, or ultimately be unable to complete, the development and commercialization of such product candidate.

All of our current development efforts are in the early stages of preclinical development. The risk of failure in developing product candidates is high. It is impossible to predict when or if any product candidate would prove effective or safe in humans or will receive regulatory approval. Before obtaining marketing approval from regulatory authorities for the sale of any product candidate, we must complete preclinical development and then conduct extensive clinical trials to demonstrate the safety and efficacy of our product candidates in humans. Clinical testing is expensive, difficult to design and implement, can take many years to complete and is uncertain as to outcome. A failure of one or more clinical trials can occur at any stage of testing. The outcome of preclinical testing and early clinical trials may not be predictive of the success of later stage clinical trials, and interim results of a clinical trial do not necessarily predict final results. Moreover, preclinical and clinical data are often susceptible to varying interpretations and analyses, and many companies that have believed their product candidates performed satisfactorily in preclinical studies and clinical trials have nonetheless failed to obtain marketing approval of their product candidates. Furthermore, the failure of any product candidates to demonstrate safety and efficacy in any clinical trial could negatively impact the perception of our other product candidates and/or cause the FDA or other regulatory authorities to require additional testing before approving any of our product candidates. For example, we previously conducted a Phase 2 clinical trial of EYSUVIS for the treatment of meibomian gland dysfunction which did not achieve its primary endpoint. The failure of this trial may have an adverse impact on the perceived safety or efficacy of EYSUVIS in treating dry eye disease or other indications or of INVELTYS.

In January 2018, we announced that we had completed two Phase 3 clinical trials evaluating EYSUVIS, STRIDE 1 and STRIDE 2, evaluating the safety and efficacy of EYSUVIS versus placebo in patients with dry eye disease. In STRIDE 1, statistical significance was achieved for both primary endpoints. However, in STRIDE 2 we did not achieve statistical significance for the primary symptom endpoint of ocular discomfort severity. In August 2019, we announced that we received a complete response letter, or CRL, from the FDA indicating that positive efficacy data from an additional clinical trial will be needed to support a resubmission of our new drug application, or NDA. On March 9, 2020, we announced that our Phase 3 clinical trial of EYSUVIS, which we refer to as STRIDE 3, met both of its primary symptom endpoints and its key secondary sign endpoint, and on April 30, 2020, we resubmitted our NDA with the positive data from STRIDE 3. On October 26, 2020, we received approval from the FDA to market EYSUVIS in the United States. Our Phase 3 clinical trials of EYSUVIS may not be sufficient to support an application for marketing approval outside the United States. Further, if regulatory authorities outside the United States do not accept the data from any trial we conduct in the United States, in particular if the European Union does not allow us to utilize the results from our Phase 3 clinical trials of EYSUVIS pursuant to the Article 10(3) submission pathway or otherwise, we will likely need to conduct additional trials to obtain marketing approval in such jurisdiction, which would be costly and time-consuming and could delay or permanently halt our ability to commercialize the applicable product candidates in the applicable jurisdictions.

If we are required to conduct additional clinical trials or other testing beyond those that we currently expect, if we are unable to successfully complete clinical trials of our product candidates or other testing, if the results of these trials or tests are not positive or are only modestly positive or if there are safety concerns, we may:

- be delayed in obtaining marketing approval for our product candidates;
- not obtain marketing approval at all;
- obtain approval for indications or patient populations that are not as broad as intended or desired;
- obtain approval with labeling that includes significant use or distribution restrictions or safety warnings, including boxed warnings;
- be subject to additional post-marketing testing requirements; or
- have the product removed from the market after obtaining marketing approval.

If we experience any of a number of possible unforeseen events in connection with our clinical trials, potential marketing approval or commercialization of our product candidates could be delayed or prevented, and our competitors could bring products to market before we do.

We may experience numerous unforeseen events during, or as a result of, clinical trials that could delay or prevent our ability to receive marketing approval or commercialize any product candidates that we may develop, including:

- clinical trials of our product candidates may produce negative or inconclusive results, and we may decide, or regulators may recommend or require us, to conduct additional clinical trials or abandon product development programs;
- the number of patients required for clinical trials of our product candidates may be larger than we anticipate, enrollment in these clinical trials may be slower than we anticipate, or participants may drop out of these clinical trials at a higher rate than we anticipate;
- our third-party contractors may fail to comply with regulatory requirements or meet their obligations to us in a timely manner, or at all;
- regulators or institutional review boards may not authorize us or our investigators to commence a clinical trial or conduct a clinical trial at a prospective trial site;
- we may experience delays in reaching, or fail to reach, agreement on acceptable clinical trial contracts or clinical trial protocols with prospective trial sites;
- we may decide, or regulators or institutional review boards may require us, to suspend or terminate clinical
  research for various reasons, including noncompliance with regulatory requirements or a finding that the
  participants are being exposed to unacceptable health risks;
- we may be subject to additional post-marketing testing requirements to maintain regulatory approval;
- regulators may revise the requirements for approving our product candidates, or such requirements may not be as we anticipate;
- the cost of clinical trials of our product candidates may be greater than we anticipate;
- the supply or quality of our product candidates or other materials necessary to conduct clinical trials of our product candidates may be insufficient or inadequate or may be delayed;
- our product candidates may have undesirable side effects or other unexpected characteristics, causing us or our investigators, regulators or institutional review boards to suspend or terminate trials;
- ongoing or future restrictions resulting from the COVID-19 pandemic and its collateral consequences may result in internal and external operational delays and limitations; and
- regulatory authorities may withdraw their approval of a product or impose restrictions on its distribution, such as in the form of a modified Risk Evaluation and Mitigation Strategy.

Our product development costs will also increase if we experience delays in testing or marketing approvals. We do not know whether any of our preclinical studies or clinical trials will begin as planned, will need to be restructured or will be completed on schedule, or at all. Significant preclinical or clinical trial delays also could shorten any periods during which we may have the exclusive right to commercialize our product candidates or allow our competitors, such as those developing treatments for dry eye disease, to bring products to market before we do and impair our ability to successfully commercialize our product candidates.

If we experience delays or difficulties in the enrollment of patients in clinical trials, our receipt of necessary regulatory approvals could be delayed or prevented.

We may not be able to initiate or continue clinical trials for product candidates we develop if we are unable to locate and enroll a sufficient number of eligible patients to participate in these trials as required by the FDA or similar regulatory authorities outside the United States.

Patient enrollment is affected by a variety of factors, including:

- the prevalence and severity of the disease or condition under investigation;
- the patient eligibility criteria for the trial in question;
- the perceived risks and benefits of the product candidate under study;
- the existence of existing treatments for the indications for which we are conducting clinical trials;
- the efforts to facilitate timely enrollment in clinical trials;
- the patient referral practices of clinicians;
- the ability to monitor patients adequately during and after treatment;
- the proximity and availability of clinical trial sites for prospective patients;
- the conducting of clinical trials by competitors for product candidates that treat the same indications as our product candidates;
- the impact of public health epidemics, such as the ongoing COVID-19 pandemic; and
- the lack of adequate compensation for prospective patients.

For example, we experienced a delay in patient enrollment for STRIDE 3, which evaluated EYSUVIS for the short-term treatment of the signs and symptoms of dry eye disease. There were a number of factors that may have impacted the delay, including increased competition for eligible patients from competitors that were developing product candidates to treat similar indications and the limited number of patients who fit the eligibility criteria for STRIDE 3. Our inability to locate and enroll a sufficient number of patients for our clinical trials could result in significant delays, could require us to abandon one or more clinical trials altogether and could delay or prevent our receipt of necessary regulatory approvals. Enrollment delays in our clinical trials may result in increased development costs for our product candidates, which would cause the value of our company to decline and limit our ability to obtain additional financing.

If serious adverse or unacceptable side effects are identified during the development or commercialization of our products or product candidates, we may need to abandon or limit our commercialization efforts for our products or development of such product candidates.

If EYSUVIS, INVELTYS or any of our product candidates are associated with serious adverse events or undesirable side effects in clinical trials or following approval and/or commercialization, or if our products or product candidates have characteristics that are unexpected, we may need to abandon their development or limit development or marketing to narrower uses or subpopulations in which the serious adverse events, undesirable side effects or other characteristics are less prevalent, less severe or more acceptable from a risk-benefit perspective. The most common adverse effects to date in trials evaluating the safety and efficacy of EYSUVIS and INVELTYS have been eye pain, instillation site pain, blurred vision and photophobia, which is discomfort or pain due to exposure to light. There have been no serious adverse events related to the administration of EYSUVIS and INVELTYS reported in any of our clinical

trials and serious adverse events reported to date following approval and commercialization have been very rare. Increases in IOP and cataract formation are additional adverse effects associated with the use of corticosteroids in general. We have no clinical safety data on or patient exposure to either EYSUVIS or INVELTYS for longer than 28 days. Our understanding of the relationship between our products and these adverse effects may change as we gather more information, and additional unexpected adverse effects may occur. Compounds that initially show promise in clinical or earlier stage testing for treating ophthalmic disease or other diseases may later be found to cause side effects that prevent further development and commercialization of the compound. In addition, adverse events which had initially been considered unrelated to the study treatment may later, even following approval and/or commercialization, be found to be caused by the study treatment. Moreover, incorrect or improper use of our products or our product candidates (including use of EYSUVIS or INVELTYS more frequently than is prescribed) by patients could cause increases in IOP and may result in additional unexpected side effects or adverse events. There can be no assurance that our products or our product candidates will be used correctly, and if used incorrectly, such misuse could hamper commercial adoption or market acceptance of our products or product candidates, if approved, at the rate we currently expect.

# We may not be successful in our efforts to develop new product candidates based on our AMPPLIFY technology or expand the use of our AMPPLIFY technology for treating additional diseases and conditions.

We are currently directing a portion of our development efforts towards applying our AMPPLIFY technology to develop new product candidates that are designed to diffuse through the mucus layer and enable the active drug substance to reach cells in the underlying target tissue. We have product candidates at various stages of development for treatment of eye diseases and may explore the potential use of our AMPPLIFY technology in other diseases. Our existing product candidates and any other potential product candidates that we identify may not be suitable for continued preclinical or clinical development, including as a result of being shown to have harmful side effects or other characteristics that indicate that they are unlikely to be products that will receive marketing approval and achieve market acceptance. If we do not successfully develop and commercialize our product candidates that we develop based upon our AMPPLIFY technology, we will not be able to obtain substantial product revenues in future periods.

We may expend our limited resources to pursue a particular product candidate or indication and fail to capitalize on product candidates or indications that may be more profitable or for which there is a greater likelihood of success.

Because we have limited financial and managerial resources, we focus on research programs and product candidates that we identify for specific indications. As a result, we may forego or delay pursuit of opportunities with other product candidates or for other indications that later prove to have greater commercial potential. Our resource allocation decisions may cause us to fail to capitalize on viable commercial products or profitable market opportunities. Our spending on current and future research and development programs and product candidates for specific indications may not yield any commercially viable products. If we do not accurately evaluate the commercial potential or target market for a particular product candidate, we may relinquish valuable rights to that product candidate through collaboration, licensing or other royalty arrangements in cases in which it would have been more advantageous for us to retain sole development and commercialization rights to such product candidate.

We may in the future conduct clinical trials for product candidates at sites outside the United States, and the FDA may not accept data from trials conducted in such locations.

We may in the future choose to conduct one or more of our clinical trials outside the United States. Although the FDA may accept data from clinical trials conducted outside the United States, acceptance of these data is subject to conditions imposed by the FDA. For example, the clinical trial must be well designed and conducted and be performed by qualified investigators in accordance with ethical principles. The trial population must also adequately represent the U.S. population, and the data must be applicable to the U.S. population and U.S. medical practice in ways that the FDA deems clinically meaningful. In addition, while these clinical trials are subject to the applicable local laws, FDA acceptance of the data will depend on its determination that the trials also complied with all applicable U.S. laws and regulations. If the FDA does not accept the data from any trial that we conduct outside the United States, it would likely result in the need for additional trials, which would be costly and time-consuming and could delay or permanently halt our development of the applicable product candidates.

## Risks Related to Our Dependence on Third Parties

We contract with third parties for the manufacture of EYSUVIS and INVELTYS and plan to contract with third parties for clinical and commercial supply of any future product candidates. This reliance on third parties increases the risk that we will not have sufficient quantities of our products and product candidates or such quantities at an acceptable cost, which could delay, prevent or impair our development or commercialization efforts.

We do not own or operate manufacturing facilities for the production of commercial quantities of EYSUVIS, INVELTYS or any product candidates. We rely on Woodstock to manufacture and supply to us a minimum amount of EYSUVIS and INVELTYS bottles. We also rely on Altasciences for manufacturing bulk intermediates, and Chemo Iberica to manufacture and supply to us a bulk supply of LE. We expect to rely on third-party manufacturers to manufacture commercial supplies of all of our products and clinical supplies of any other product candidates if and when approved for marketing by applicable regulatory authorities. Our current and anticipated future dependence upon others for the manufacture of EYSUVIS, INVELTYS and any other product or product candidate that we develop may adversely affect our future profit margins and our ability to commercialize any products that receive marketing approval on a timely and competitive basis. In addition, certain of our third-party manufacturers have in the past, and may in the future, experience performance issues that result in lower than expected yields. Any performance failure on the part of our existing or future manufacturers could delay clinical development, marketing approval or the supply and sale of any product of ours that has been approved for commercial use.

To date, we have obtained materials for our clinical trials and the commercialization of EYSUVIS and INVELTYS from third-party manufacturers, including Woodstock and Altasciences. We have supply agreements in place with these contract manufacturers to provide commercial supply. We obtain the API for EYSUVIS and INVELTYS from Chemo Iberica, a third-party API manufacturer. While we have long-term commercial supply agreements with these third-party manufacturers, if these suppliers do not perform as we expect, we may be required to replace one or more suppliers. Although we believe that there are a number of potential long-term replacements to our suppliers, we may incur added costs and delays in identifying and qualifying any such replacements.

The FDA maintains strict requirements governing the manufacturing process. When a manufacturer seeks to modify or make even seemingly minor changes to that process, the FDA may require the applicant to conduct a comparability study that evaluates the potential differences in the product resulting from the change in the manufacturing process. The FDA has issued several rounds of guidance on this point. In connection with any application for approval to market product candidates in the United States, we may be required to conduct a comparability study if the product we intend to market is supplied by a manufacturer different from the one who supplied the product evaluated in our clinical studies. Delays in designing and completing this study to the satisfaction of the FDA could delay or preclude our development and commercialization plans and thereby limit our revenues and growth.

Reliance on third-party manufacturers entails additional risks, including:

- EYSUVIS, INVELTYS and any other product that we develop may compete with other product candidates
  and products for access to a limited number of suitable manufacturing facilities that operate under current
  good manufacturing practices, or cGMP, regulations;
- reliance on the third-party for regulatory compliance and quality assurance;
- the possible breach of the manufacturing agreement by the third-party;
- the possible misappropriation of our proprietary information, including our trade secrets and know-how; and
- the possible termination or nonrenewal of the agreement by the third-party at a time that is costly or inconvenient for us.

Third-party manufacturers may not be able to comply with cGMP regulations or similar regulatory requirements outside the United States. Our failure, or the failure of our third-party manufacturers, to comply with applicable regulations could result in sanctions being imposed on us, including clinical holds, fines, injunctions, civil penalties, delays, suspension or withdrawal of approvals, license revocation, seizures or recalls of product candidates or products, operating restrictions and criminal prosecutions, any of which could significantly and adversely affect supplies of our products and harm our business and results of operations.

Any products that we may develop may compete with other product candidates and products for access to manufacturing facilities. There are a limited number of manufacturers that operate under cGMP regulations and that might be capable of manufacturing for us. We were previously required to change our third-party manufacturer when the manufacturer was purchased by a third-party and exited the contract manufacturing business. The process of changing manufacturers can cause substantial time delays, and if we are required to change our manufacturer again in the future, it may delay our planned clinical trials or development timeline.

Any performance failure on the part of our existing or future manufacturers could delay clinical development or marketing approval. We do not currently have arrangements in place for redundant supply for bulk drug substances. If any one of our current contract manufacturers cannot perform as agreed, we may be required to replace that manufacturer. Although we believe that there are several potential alternative manufacturers who could manufacture our product candidates, we may incur added costs and delays in identifying and qualifying any such replacement. Additionally, while we currently are not experiencing interruptions in our manufacturing of EYSUVIS or INVELTYS, quarantines, travel restrictions and bans and other governmental restrictions related to COVID-19 may significantly impact the ability of employees of our third-party suppliers to get to their places of work to manufacture and deliver future supply if and when needed.

Our current and anticipated future dependence upon others for the manufacture of EYSUVIS, INVELTYS or our product candidates may adversely affect our future profit margins and our ability to commercialize any medicines that receive marketing approval on a timely and competitive basis.

We may enter into collaborations with third parties for the development or commercialization of our products and product candidates. If our collaborations are not successful, we may not be able to capitalize on the market potential of these products and product candidates.

We expect to utilize a variety of types of collaboration, distribution and other marketing arrangements with third parties to develop and commercialize EYSUVIS, INVELTYS or any of our product candidates for which we seek or obtain marketing approval in markets outside the United States. We also may enter into arrangements with third parties to perform these services in the United States to enhance our own sales, marketing and distribution capabilities in the United States or if we determine that such third-party arrangements are otherwise beneficial. We also may seek third-party collaborators for development and commercialization of our product candidates. For example, we may consider potential collaborative partnership opportunities prior to initiating IND-enabling studies on K006, KPI-333 or any other product candidates we develop, including our SEGRMs. Our likely collaborators for any sales, marketing, distribution, development, licensing or broader collaboration arrangements include large and mid-size pharmaceutical companies, regional and national pharmaceutical companies and biotechnology companies. We are not currently party to any such arrangement. However, if we do enter into any such arrangements with any third parties in the future, we will likely have limited control over the amount and timing of resources that our collaborators dedicate to the development or commercialization of our products and product candidates. Our ability to generate revenues from these arrangements will depend on our collaborators' abilities and efforts to successfully perform the functions assigned to them in these arrangements.

Collaborations that we enter into may pose a number of risks, including the following:

- collaborators have significant discretion in determining the amount and timing of efforts and resources that
  they will apply to these collaborations;
- collaborators may not perform their obligations as expected;

- collaborators may not pursue development of our product candidates or may elect not to continue or renew
  development programs based on results of clinical trials or other studies, changes in the collaborators'
  strategic focus or available funding, or external factors, such as an acquisition, that divert resources or create
  competing priorities;
- collaborators may not pursue commercialization of our product candidates that receive marketing approval or
  may elect not to continue or renew commercialization programs based on changes in the collaborators'
  strategic focus or available funding, or external factors, such as an acquisition, that divert resources or create
  competing priorities;
- collaborators may delay clinical trials, provide insufficient funding for a clinical trial program, stop a clinical trial or abandon a product candidate, repeat or conduct new clinical trials or require a new formulation of a product candidate for clinical testing;
- collaborators could independently develop, or develop with third parties, products that compete directly or
  indirectly with our products or product candidates if the collaborators believe that competitive products are
  more likely to be successfully developed or can be commercialized under terms that are more economically
  attractive than ours;
- product candidates discovered in collaboration with us may be viewed by our collaborators as competitive
  with their own products or product candidates, which may cause collaborators to cease to devote resources to
  the commercialization of our product candidates;
- a collaborator with marketing and distribution rights to one or more of our products or product candidates
  that achieve regulatory approval may not commit sufficient resources to the marketing and distribution of
  such product or products;
- disagreements with collaborators, including disagreements over proprietary rights, contract interpretation or
  the preferred course of development, might cause delays or termination of the research, development or
  commercialization of product candidates, might lead to additional responsibilities for us with respect to
  product candidates, or might result in litigation or arbitration, any of which would divert management
  attention and resources, be time-consuming and expensive;
- collaborators may not properly maintain or defend our intellectual property rights or may use our proprietary
  information in such a way as to invite litigation that could jeopardize or invalidate our intellectual property or
  proprietary information or expose us to potential litigation;
- collaborators may infringe, misappropriate or otherwise violate the intellectual property rights of third
  parties, which may expose us to litigation and potential liability; and
- collaborations may be terminated for the convenience of the collaborator and, if terminated, we could be
  required to raise additional capital to pursue further development or commercialization of the applicable
  product candidates.

Collaboration agreements may not lead to development or commercialization of products or product candidates in the most efficient manner, or at all. If any collaborations that we enter into do not result in the successful development and commercialization of products or if one of our collaborators terminates its agreement with us, we may not receive any future research funding or milestone or royalty payments under the collaboration. If we do not receive the funding we expect under these agreements, our development of our product candidates could be delayed, and we may need additional resources to develop our product candidates. All of the risks relating to product development, regulatory approval and commercialization described herein also apply to the activities of our collaborators.

Additionally, subject to its contractual obligations to us, if a collaborator of ours were to be involved in a business combination, it might de-emphasize or terminate the development or commercialization of any product or product candidate licensed to it by us. If one of our collaborators terminates its agreement with us, we may find it more difficult to attract new collaborators and our perception in the business and financial communities could be harmed.

We rely, and expect to continue to rely, on third parties to conduct our clinical trials, and those third parties may not perform satisfactorily, including failing to meet deadlines for the completion of such trials.

We rely on third parties, such as clinical research organizations, or CROs, clinical data management organizations, medical institutions and clinical investigators, in conducting our clinical trials and expect to continue to rely on such parties to conduct clinical trials of any product candidate that we develop. We or these third parties may terminate their engagements with us at any time for a variety of reasons, including a failure to perform by the third parties. If we need to enter into alternative arrangements, that could delay our product development activities.

Our reliance on these third parties for clinical development activities reduces our control over these activities but does not relieve us of our responsibilities. For example, we remain responsible for ensuring that each of our clinical trials is conducted in accordance with the general investigational plan and protocols for the trial. Moreover, the FDA requires us to comply with standards, commonly referred to as Good Clinical Practices, for conducting, recording and reporting the results of clinical trials to assure that data and reported results are credible and accurate and that the rights, integrity and confidentiality of trial participants are protected. We also are required to register ongoing clinical trials and post the results of completed clinical trials on a government-sponsored database, ClinicalTrials.gov, within specified timeframes. Failure to do so can result in fines, adverse publicity and civil and criminal sanctions.

If these third parties do not successfully carry out their contractual duties, meet expected deadlines or conduct our clinical trials in accordance with regulatory requirements or our stated protocols, we will not be able to obtain, or may be delayed in obtaining, marketing approvals for our product candidates and will not be able to, or may be delayed in our efforts to, successfully commercialize our product candidates. Furthermore, these third parties may also have relationships with other entities, some of which may be our competitors.

We also rely on other third parties to store and distribute drug supplies for our clinical trials. Any performance failure on the part of our distributors could delay clinical development or marketing approval of our product candidates or commercialization of our products, producing additional losses and depriving us of potential product revenue.

If we are not able to establish collaborations, we may have to alter our development and commercialization plans and our business could be adversely affected.

For some of our product candidates, we may decide to collaborate with pharmaceutical or biotechnology companies for the development of our product candidates and the commercialization of our products or the potential commercialization of our product candidates. We face significant competition in seeking appropriate collaborators. Whether we reach a definitive agreement for a collaboration will depend, among other things, upon our assessment of the collaborator's resources and expertise, the terms and conditions of the proposed collaboration and the proposed collaborator's evaluation of a number of factors. Those factors may include the design or results of clinical trials, the likelihood of approval by the FDA or similar regulatory authorities outside the United States, the potential market for the subject product candidate, the costs and complexities of manufacturing and delivering such product candidate to patients, the potential of competing products, the existence of uncertainty with respect to our ownership of technology, which can exist if there is a challenge to such ownership without regard to the merits of the challenge, and industry and market conditions generally. The collaborator may also consider alternative product candidates or technologies for similar indications that may be available to collaborate on and whether such a collaboration could be more attractive than the one with us for our product or product candidate. We may also be restricted under future license agreements from entering into agreements on certain terms with potential collaborators. Collaborations are complex and time-consuming to negotiate and document. In addition, there have been a significant number of recent business combinations among large pharmaceutical companies that have resulted in a reduced number of potential future collaborators.

If we are unable to reach agreements with suitable collaborators on a timely basis, on acceptable terms, or at all, we may have to curtail the development of a product candidate, reduce or delay its development program or one or more of our other development programs, delay the commercialization of a product or a product candidate or reduce the scope of any sales or marketing activities, or increase our expenditures and undertake development or commercialization activities at our own expense. If we elect to fund and undertake development or commercialization activities on our own, we may need to obtain additional expertise and additional capital, which may not be available to us on acceptable terms or at all. If we fail to enter into collaborations and do not have sufficient funds or expertise to undertake the necessary development and commercialization activities, we may not be able to further develop our product candidates or bring them to market or continue to develop our product platform.

### **Risks Related to Our Intellectual Property**

We may be unable to obtain and maintain patent protection for our technology, products and product candidates, or the scope of the patent protection obtained may not be sufficiently broad or enforceable, such that our competitors could develop and commercialize technology, products and product candidates similar or identical to ours, and our ability to successfully commercialize our technology, products and product candidates may be impaired.

Our success depends in large part on our ability to obtain and maintain patent protection in the United States and other countries with respect to our proprietary technology, products and product candidates. We have sought to protect our proprietary position by filing in the United States and in certain foreign jurisdictions patent applications related to our novel technologies, products and product candidates.

The patent prosecution process is expensive and time-consuming, and we may not have filed, maintained, or prosecuted and may not be able to file, maintain and prosecute all necessary or desirable patents or patent applications at a reasonable cost or in a timely manner. We may also fail to identify patentable aspects of our research and development output before it is too late to obtain patent protection.

The patent position of pharmaceutical, biotechnology, and medical device companies generally is highly uncertain, involves complex legal and factual questions and has in recent years been the subject of much litigation. As a result, the issuance, scope, validity, enforceability and commercial value of our patent rights are highly uncertain. Our pending and future patent applications may fail to result in issued patents in the United States or in other foreign countries which protect our technology, products or product candidates, or which effectively prevent others from commercializing competitive technologies and products. In addition, the laws of foreign countries may not protect our rights to the same extent as the laws of the United States, and the standards applied by the U.S. Patent and Trademark Office and foreign patent offices in granting patents are not always applied uniformly or predictably. For example, unlike patent law in the United States, European patent law precludes the patentability of methods of treatment of the human body and imposes substantial restrictions on the scope of claims it will grant if broader than specifically disclosed embodiments. Publications of discoveries in the scientific literature often lag behind the actual discoveries, and patent applications in the United States and other jurisdictions are typically not published until 18 months after filing, or in some cases not at all. Therefore, we cannot be certain whether we or our licensors were the first to make the inventions claimed in our owned or licensed patents or pending patent applications, or that we or our licensors were the first to file for patent protection of such inventions. Databases for patents and publications, and methods for searching them, are inherently limited so we may not know the full scope of all issued and pending patent applications. As a result, the issuance, scope, validity, enforceability, and commercial value of our patent rights are uncertain. Our pending and future patent applications may not result in patents being issued which protect our technology, products or product candidates, in whole or in part, or which effectively prevent others from commercializing competitive technologies, products and product candidates. In particular, during prosecution of any patent application, the issuance of any patents based on the application may depend upon our ability to generate additional preclinical or clinical data that support the patentability of our proposed claims. We may not be able to generate sufficient additional data on a timely basis, or at all. Moreover, changes in either the patent laws or interpretation of the patent laws in the United States and other countries may diminish the value of our patents or narrow the scope of our patent protection.

Even if our owned and licensed patent applications issue as patents, they may not issue in a form that will provide us with any meaningful protection for our proprietary technology, products and product candidates, prevent competitors from competing with us, or otherwise provide us with any competitive advantage. Our competitors may be able to circumvent our owned or licensed patents by developing similar or alternative technologies, products or product candidates in a non-infringing manner. In particular, a competitor may develop an approach to deliver drugs through the mucus layer to the underlying target tissue that uses a different approach than our AMPPLIFY technology, and therefore may not infringe on our patent rights.

The issuance of a patent is not conclusive as to its inventorship, ownership, scope, validity, or enforceability, and our owned and licensed patents may be challenged in the courts or patent offices in the United States and abroad. Such challenges may result in loss of exclusivity or in patent claims being narrowed, invalidated or held unenforceable, in whole or in part, which could limit our ability to stop others from using or commercializing similar or identical technology, products or product candidates, or limit the duration of the patent protection of our technology, products and product candidates. Given the amount of time required for the development, testing, and regulatory review of new product candidates, patents protecting such candidates might expire before or shortly after such candidates are commercialized. As a result, our patent portfolio may not provide us with sufficient rights to exclude others from commercializing products similar or identical to ours.

If we are not able to obtain patent term extension in the United States under the Hatch-Waxman Act and in foreign countries under similar legislation, thereby potentially extending the term of our marketing exclusivity for our products or product candidates, our business may be materially harmed.

Depending upon the timing, duration, and specifics of FDA marketing approval of our product candidates, one of the U.S. patents covering each of such product candidates or the use thereof may be eligible for up to five years of patent term extension under the Hatch-Waxman Act. The Hatch-Waxman Act allows a maximum of one patent to be extended per FDA approved product as compensation for the patent term lost during the FDA regulatory review process. A patent term extension cannot extend the remaining term of a patent beyond a total of 14 years from the date of product approval and only those claims covering such approved drug product, a method for using it or a method for manufacturing it may be extended. Also, the regulatory review period of an FDA-approved product may not serve as a basis for a patent term extension if the active ingredient of such product was subject to regulatory review and approval in an earlier product approved by the FDA. We do not expect the U.S. patents covering EYSUVIS and INVELTYS to be eligible for patent term extension due to this limitation. Patent term extension also may be available in certain foreign countries upon regulatory approval of our product candidates. Nevertheless, we may not be able to seek or be granted patent term extension either in the United States or in any foreign country because of, for example, failing to exercise due diligence during the testing phase or regulatory review process, failing to apply within applicable deadlines, failing to apply prior to expiration of relevant patents, or otherwise failing to satisfy applicable requirements. Moreover, the term of extension, as well as the scope of patent protection during any such extension, afforded by the governmental authority could be less than we request.

If we are unable to obtain patent term extension or restoration, or the term of any such extension is less than we request, the period during which we will have the right to exclusively market our product may be shortened and our competitors may obtain approval of competing products following our patent expiration sooner, and our revenue could be reduced, possibly materially.

It is possible that we will not obtain patent term extension under the Hatch-Waxman Act for a U.S. patent covering products or one of our product candidates even where that patent is eligible for patent term extension, or if we obtain such an extension, it may be for a shorter period than we had sought. Further, for our licensed patents, we may not have the right to control prosecution, including filing with the U.S. Patent and Trademark Office, a petition for patent term extension under the Hatch-Waxman Act. Thus, if one of our licensed patents is eligible for patent term extension under the Hatch-Waxman Act, we may not be able to control whether a petition to obtain a patent term extension is filed, or obtained, from the U.S. Patent and Trademark Office.

Also, there are detailed rules and requirements regarding the patents that may be submitted to the FDA for listing in the Approved Drug Products with Therapeutic Equivalence Evaluations, or the Orange Book. We may be unable to obtain patents covering our product candidates that contain one or more claims that satisfy the requirements for listing in the Orange Book. Even if we submit a patent for listing in the Orange Book, the FDA may decline to list the patent, or a manufacturer of generic drugs may challenge the listing. If one of our product candidates is approved and a patent covering that product candidate is not listed in the Orange Book, with respect to the patent, a manufacturer of generic drugs would not have to provide advance notice to us of any Abbreviated New Drug Application filed with the FDA to obtain permission to sell a generic version of such product candidate.

We may become involved in lawsuits to protect or enforce our patents or other intellectual property, which could be expensive, time-consuming and unsuccessful.

Competitors and other third parties may infringe, misappropriate or otherwise violate our owned and licensed patents, trade secrets, or other intellectual property. As a result, to counter infringement, misappropriation or unauthorized use, we may be required to file infringement or misappropriation claims or other intellectual property related proceedings, which can be expensive and time-consuming. Any claims we assert against perceived infringers could provoke these parties to assert counterclaims against us alleging that we infringe their patents or that our asserted patents are invalid. In addition, in a patent infringement or other intellectual property related proceeding, a court may decide that a patent of ours is invalid or unenforceable, in whole or in part, construe the patent's claims narrowly or refuse to stop the other party from using the technology at issue on the grounds that our patents do not cover the technology in question. An adverse result in any litigation proceeding could put one or more of our patents at risk of being invalidated, held unenforceable or interpreted narrowly, and could put any of our patent applications at risk of not yielding an issued patent. Furthermore, because of the substantial amount of discovery required in connection with intellectual property litigation, there is a risk that some of our confidential information or trade secrets could be compromised by disclosure during this type of litigation.

We may be subject to a third-party preissuance submission of prior art to the U.S. Patent and Trademark Office, or become involved in other contested proceedings such as opposition, derivation, reexamination, *inter partes* review, postgrant review, or interference proceedings in the United States or elsewhere, challenging our patent rights or the patent rights of others. An adverse determination in any such submission, proceeding or litigation could reduce the scope of, or invalidate, our patent rights, allow third parties to commercialize our technology, products or product candidates and compete directly with us, without payment to us, or result in our inability to manufacture or commercialize products without infringing third-party patent rights. In addition, if the breadth or strength of protection provided by our patents and patent applications is threatened, it could dissuade companies from collaborating with us to license, develop or commercialize current or future product candidates.

In the United States, the FDA does not prohibit clinicians from prescribing an approved product for uses that are not described in the product's labeling. Although use of a product directed by off-label prescriptions may infringe our method-of-treatment patents, the practice is common across medical specialties, particularly in the United States, and such infringement is difficult to detect, prevent, or prosecute.

Third parties may initiate legal proceedings alleging that we are infringing, misappropriating or otherwise violating their intellectual property rights, the outcome of which would be uncertain and could have a material adverse effect on the success of our business.

Our commercial success depends upon our ability to develop, manufacture, market, and sell EYSUVIS, INVELTYS and our product candidates and use our proprietary technologies without infringing, misappropriating or otherwise violating the intellectual property and other proprietary rights of third parties. There is a considerable amount of intellectual property litigation in the biotechnology and pharmaceutical industries. We may become party to, or threatened with, infringement litigation claims regarding our products, product candidates and technology, including claims from competitors or from non-practicing entities that have no relevant product revenue and against whom our own patent portfolio may have no deterrent effect. Moreover, we may become party to future adversarial proceedings or litigation regarding our patent portfolio or the patents of third parties. Such proceedings could also include contested post-grant proceedings such as oppositions, *inter partes* review, reexamination, interference, or derivation proceedings before the U.S. Patent and Trademark Office or foreign patent offices.

The legal threshold for initiating litigation or contested proceedings is low, so that even lawsuits or proceedings with a low probability of success might be initiated and require significant resources to defend. Litigation and contested proceedings can also be expensive and time-consuming, and our adversaries in these proceedings may have the ability to dedicate substantially greater resources to prosecuting these legal actions than we can. The risks of being involved in such litigation and proceedings may increase as our product candidates commence commercialization and as we gain the greater visibility associated with being a public company. Third parties may assert infringement claims against us based on existing patents or patents that may be granted in the future. We may not be aware of all such intellectual property rights potentially relating to our products or product candidates and their uses. Thus, we do not know with certainty that EYSUVIS, INVELTYS or any of our product candidates or our development and commercialization thereof, do not and will not infringe or otherwise violate any third-party's intellectual property.

If we are found to infringe, misappropriate or otherwise violate a third-party's intellectual property rights, we could be required to obtain a license from such third-party to continue developing, manufacturing, marketing and selling our products, product candidates and technology. However, we may not be able to obtain any required license on commercially reasonable terms or at all. Even if we were able to obtain a license, it could be non-exclusive, thereby giving our competitors access to the same technologies licensed to us and could require us to make substantial licensing and royalty payments. We could be forced, including by court order, to cease commercializing the infringing technology, products or product candidates. In addition, we could be found liable for monetary damages, including treble damages and attorneys' fees if we are found to have willfully infringed a patent and could be forced to indemnify our customers or collaborators. A finding of infringement could also result in an injunction that prevents us from commercializing our products or product candidates or forces us to cease some of our business operations, which could materially harm our business. In addition, we may be forced to redesign our product candidates, seek new regulatory approvals and indemnify third parties pursuant to contractual agreements. Claims that we have misappropriated the confidential information or trade secrets of third parties could have a similar negative impact on our business.

Obtaining and maintaining patent protection depends on compliance with various procedural, document submission, fee payment and other requirements imposed by governmental patent agencies, and our patent protection could be reduced or eliminated for non-compliance with these requirements.

Periodic maintenance, renewal and annuity fees on any issued patent must be paid to the U.S. Patent and Trademark Office and foreign patent agencies in several stages or annually over the lifetime of our owned and licensed patents and patent applications. The U.S. Patent and Trademark Office and various foreign governmental patent agencies require compliance with a number of procedural, documentary, fee payment and other similar provisions during the patent application process. In certain circumstances, we rely on our licensing partners to pay these fees to, or comply with the procedural and documentary rules of, the relevant patent agency. While an inadvertent lapse can in many cases be cured by payment of a late fee or by other means in accordance with the applicable rules, there are situations in which noncompliance can result in abandonment or lapse of the patent or patent application, resulting in partial or complete loss of patent rights in the relevant jurisdiction. Non-compliance events that could result in abandonment or lapse of a patent or patent application include failure to respond to official actions within prescribed time limits, non-payment of fees and failure to properly legalize and submit formal documents. If we or our licensors fail to maintain the patents and patent applications covering our product candidates, it would have a material adverse effect on our business.

EYSUVIS, INVELTYS and certain aspects of our AMPPLIFY technology are protected by patents exclusively licensed from other companies or institutions. If these third parties terminate their agreements with us or fail to maintain or enforce the underlying patents, or we otherwise lose our rights to these patents, our competitive position and our market share in the markets for any of our approved products will be harmed.

A substantial portion of our patent portfolio is in-licensed. As such, we are a party to license agreements and certain aspects of our business depend on patents and/or patent applications owned by other companies or institutions. In particular, we hold exclusive licenses for patent families relating to EYSUVIS, INVELTYS and our product candidates and some aspects of our AMPPLIFY technology. While we control patent prosecution of the licensed patent families relating to EYSUVIS and INVELTYS, for the remainder of the patent families subject to our exclusive license agreement with The Johns Hopkins University, or JHU, that relate to our AMPPLIFY technology, JHU retains control of patent prosecution. Our rights with respect to in-licensed patents and patent applications may be lost if the applicable

license agreement expires or is terminated. We are likely to enter into additional license agreements to in-license patents and patent applications as part of the development of our business in the future, under which we may not retain control of the preparation, filing, prosecution, maintenance, enforcement and defense of such patents. If we are unable to maintain these patent rights for any reason, our ability to develop and commercialize our products or product candidates could be materially harmed.

Our licensors may not successfully prosecute certain patent applications, the prosecution of which they control, under which we are licensed and on which our business depends. Even if patents issue from these applications, our licensors may fail to maintain these patents, may decide not to pursue litigation against third-party infringers, may fail to prove infringement, or may fail to defend against counterclaims of patent invalidity or unenforceability.

Risks with respect to parties from whom we have obtained intellectual property rights may also arise out of circumstances beyond our control. In spite of our best efforts, our licensors might conclude that we have materially breached our intellectual property agreements and might therefore terminate the intellectual property agreements, thereby removing our ability to market products covered by these intellectual property agreements. If our intellectual property agreements are terminated, or if the underlying patents fail to provide the intended market exclusivity, competitors would have the freedom to seek regulatory approval of, and to market, products similar or identical to ours. Moreover, if our intellectual property agreements are terminated, our former licensors and/or assignors may be able to prevent us from utilizing the technology covered by the licensed or assigned patents and patent applications. This could have a material adverse effect on our competitive business position and our financial condition, results of operations and our business prospects.

Some intellectual property which we own or have licensed may have been discovered through government funded programs and thus may be subject to federal regulations such as "march-in" rights, certain reporting requirements, and a preference for United States industry. Compliance with such regulations may limit our exclusive rights, subject us to expenditure of resources with respect to reporting requirements, and limit our ability to contract with non-U.S. manufacturers.

Some of the intellectual property rights we own or have licensed have been generated through the use of United States government funding and may therefore be subject to certain federal regulations. For example, certain aspects of our AMPPLIFY technology as well as certain aspects of our patents that use LE as an active ingredient were developed using United States government funds. As a result, the United States government may have certain rights to intellectual property embodied in our current or future products and product candidates based on our AMPPLIFY technology or that use LE as an active ingredient pursuant to the Bayh-Dole Act of 1980. These United States government rights in certain inventions developed under a government-funded program include a non-exclusive, non-transferable, irrevocable worldwide license to use inventions for any governmental purpose. In addition, the United States government has the right to require us to grant exclusive, partially exclusive, or non-exclusive licenses to any of these inventions to a third-party if it determines that: (i) adequate steps have not been taken to commercialize the invention; (ii) government action is necessary to meet public health or safety needs; or (iii) government action is necessary to meet requirements for public use under federal regulations (also referred to as "march-in rights"). The United States government also has the right to take title to these inventions if we fail to disclose the invention to the government and fail to file an application to register the intellectual property within specified time limits. In addition, the United States government may acquire title to these inventions in any country in which a patent application is not filed within specified time limits. Intellectual property generated under a government funded program is also subject to certain reporting requirements, compliance with which may require us to expend substantial resources. In addition, the United States government requires that any products embodying the subject invention or produced through the use of the subject invention be manufactured substantially in the United States. The manufacturing preference requirement can be waived if the owner of the intellectual property can show that reasonable but unsuccessful efforts have been made to grant licenses on similar terms to potential licensees that would be likely to manufacture substantially in the United States or that under the circumstances domestic manufacture is not commercially feasible. This preference for United States manufacturers may limit our ability to contract with non-U.S. product manufacturers for products covered by such intellectual property. Any exercise by the government of any of the foregoing rights could harm our competitive position, business, financial condition, results of operations and prospects.

If we fail to comply with our obligations in our intellectual property licenses and funding arrangements with third parties, we could lose rights that are important to our business.

Our license agreement with JHU, under which we license certain of our patent rights and a significant portion of the technology for EYSUVIS, INVELTYS and our product candidates imposes royalty and other financial obligations on us and other substantial performance obligations. We also may enter into additional licensing and funding arrangements with third parties that may impose diligence, development and commercialization timelines and milestone payment, royalty, insurance and other obligations on us. If we fail to comply with our obligations under current or future license and collaboration agreements, our counterparties may have the right to terminate these agreements, in which event we might not be able to develop, manufacture or market any product or product candidate that is covered by these agreements or may face other penalties under the agreements. Such an occurrence could diminish the value of our products or product candidates. Termination of these agreements or reduction or elimination of our rights under these agreements may result in our having to negotiate new or reinstated agreements with less favorable terms, or cause us to lose our rights under these agreements, including our rights to important intellectual property or technology.

In addition, it is possible that JHU may conclude that we have materially breached the JHU licensing agreement and might therefore terminate the agreement, thereby removing our ability to market products covered by our license agreement with JHU. If the JHU licensing agreement is terminated, or if the underlying patents fail to provide the intended market exclusivity, competitors would have the freedom to seek regulatory approval of, and to market, products similar or identical to ours. Moreover, if our license agreement with JHU is terminated, JHU and/or its assignors may be able to prevent us from utilizing the technology covered by the licensed or assigned patents and patent applications. If we breach the agreement (including by failing to meet our payment obligations) and do not adequately cure such breach, the rights in the technology licensed to us under the JHU license agreement will revert to JHU at no cost to JHU. This could have a material adverse effect on our competitive business position, our financial condition, our results of operations and our business prospects.

In addition, the agreements under which we currently license intellectual property or technology from third parties are complex, and certain provisions in such agreements may be susceptible to multiple interpretations. The resolution of any contract interpretation disagreement that may arise could narrow what we believe to be the scope of our rights to the relevant intellectual property or technology, or increase what we believe to be our financial or other obligations under the relevant agreement, either of which could have a material adverse effect on our business, financial condition, results of operations, and prospects. Moreover, if disputes over intellectual property that we have licensed prevent or impair our ability to maintain our current licensing arrangements on commercially acceptable terms, we may be unable to successfully develop and commercialize the affected products or product candidates, which could have a material adverse effect on our business, financial conditions, results of operations, and prospects.

# We may not be able to protect our intellectual property and proprietary rights throughout the world.

Filing, prosecuting, and defending patents on our products and product candidates in all countries throughout the world would be prohibitively expensive, and the laws of foreign countries may not protect our rights to the same extent as the laws of the United States. Consequently, we may not be able to prevent third parties from practicing our inventions in all countries outside the United States, or from selling or importing products made using our inventions in and into the United States or other jurisdictions. Competitors may use our technologies in jurisdictions where we have not obtained patent protection to develop their own products and, further, may export otherwise infringing products to territories where we have patent protection or licenses, but enforcement is not as strong as that in the United States. These products may compete with our products, and our patents or other intellectual property rights may not be effective or sufficient to prevent them from competing.

Many companies have encountered significant problems in protecting and defending intellectual property rights in foreign jurisdictions. The legal systems of certain countries, particularly certain developing countries, do not favor the enforcement of patents, trade secrets, and other intellectual property protection, particularly those relating to biotechnology products, which could make it difficult for us to stop the infringement of our patents or marketing of competing products in violation of our intellectual property and proprietary rights generally. Proceedings to enforce our intellectual property and proprietary rights in foreign jurisdictions could result in substantial costs and divert our efforts

and attention from other aspects of our business, could put our patents at risk of being invalidated or interpreted narrowly, could put our patent applications at risk of not issuing, and could provoke third parties to assert claims against us. We may not prevail in any lawsuits that we initiate, and the damages or other remedies awarded, if any, may not be commercially meaningful. Accordingly, our efforts to enforce our intellectual property and proprietary rights around the world may be inadequate to obtain a significant commercial advantage from the intellectual property that we develop or license.

Many countries have compulsory licensing laws under which a patent owner may be compelled to grant licenses to third parties. In addition, many countries limit the enforceability of patents against government agencies or government contractors. In these countries, the patent owner may have limited remedies, which could materially diminish the value of such patent. If we or any of our licensors is forced to grant a license to third parties with respect to any patents relevant to our business, our competitive position may be impaired, and our business, financial condition, results of operations, and prospects may be adversely affected.

We may be subject to claims by third parties asserting that our employees or we have misappropriated their intellectual property, or claiming ownership of what we regard as our own intellectual property.

Many of our and our licensors' employees and contractors were previously employed at other biotechnology, medical device or pharmaceutical companies, including our competitors or potential competitors. Although we try to ensure that our employees and contractors do not use the proprietary information or know-how of others in their work for us, we may be subject to claims that these individuals have used or disclosed intellectual property, including trade secrets or other proprietary information, of any such employee's former employer. Litigation may be necessary to defend against these claims

In addition, while it is our policy to require our employees and contractors who may be involved in the development of intellectual property to execute agreements assigning such intellectual property to us, we may be unsuccessful in executing such an agreement with each party who in fact develops intellectual property that we regard as our own. Furthermore, we are unable to control whether our licensors have obtained similar assignment agreements from their own employees and contractors. Our and their assignment agreements may not be self-executing or may be breached, and we or our licensors may be forced to bring claims against third parties, or defend claims they may bring against us, to determine the ownership of what we regard as our intellectual property.

If we or our licensors fail in prosecuting or defending any such claims, in addition to paying monetary damages, we may lose valuable intellectual property rights or personnel which could have a material adverse effect on our competitive business position and prospects. Such intellectual property rights could be awarded to a third-party, and we could be required to obtain a license from such third-party to commercialize our technology or products, which may not be available on commercially reasonable terms or at all. Even if we are successful in prosecuting or defending against such claims, litigation could result in substantial costs and be a distraction to management.

Intellectual property litigation or other legal proceedings relating to intellectual property could cause us to spend substantial resources and distract our personnel from their normal responsibilities.

Even if resolved in our favor, litigation or other legal proceedings relating to intellectual property claims may cause us to incur significant expenses, and could distract our technical and management personnel from their normal responsibilities. In addition, there could be public announcements of the results of hearings, motions or other interim proceedings or developments and if securities analysts or investors perceive these results to be negative, it could have a substantial adverse effect on the price of our common stock. Such litigation or proceedings could substantially increase our operating losses and reduce the resources available for development activities or any future sales, marketing or distribution activities. We may not have sufficient financial or other resources to conduct such litigation or proceedings adequately. Some of our competitors may be able to sustain the costs of such litigation or proceedings more effectively than we can because of their greater financial resources and may also have an advantage in such proceedings due to their more mature and developed intellectual property portfolios. Uncertainties resulting from the initiation and continuation of patent litigation or other proceedings could have an adverse effect on our ability to compete in the marketplace.

# If we are unable to protect the confidentiality of our trade secrets, our business and competitive position would be harmed.

In addition to seeking patents for our technology, our products and product candidates, we also rely on trade secrets, including unpatented know-how, technology and other proprietary information, to maintain our competitive position. We seek to protect these trade secrets, in part, by entering into non-disclosure and confidentiality agreements with parties who have access to them, such as our employees, corporate collaborators, outside scientific collaborators, contract manufacturers, consultants, advisors and other third parties. We also enter into confidentiality and invention or patent assignment agreements with our employees and consultants. Despite these efforts, any of these parties may breach the agreements and disclose our proprietary information, including our trade secrets, and we may not be able to obtain adequate remedies for such breaches. Detecting the disclosure or misappropriation of a trade secret and enforcing a claim that a party illegally disclosed or misappropriated a trade secret is difficult, expensive and time-consuming, and the outcome is unpredictable. In addition, some courts inside and outside the United States are less willing or unwilling to protect trade secrets. If any of our trade secrets were to be lawfully obtained or independently developed by a competitor, we would have no right to prevent them, or those to whom they communicate it, from using that technology or information to compete with us. If any of our trade secrets were to be disclosed to or independently developed by a competitor, our competitive position would be harmed.

# Risks Related to Regulatory Approval of Our Product Candidates and Other Legal Compliance Matters

If we are not able to obtain required regulatory approvals, we will not be able to commercialize our product candidates, and our ability to generate significant revenue will be materially impaired.

Our product candidates and the activities associated with their development and commercialization, including their design, testing, manufacture, safety, efficacy, recordkeeping, labeling, storage, approval, advertising, promotion, sale and distribution, are subject to comprehensive regulation by the FDA and other regulatory agencies in the United States and by comparable authorities in other countries. Failure to obtain marketing approval for a product candidate will prevent us from commercializing the product candidate.

Other than EYSUVIS and INVELTYS, we have not received approval to market any product candidate from regulatory authorities in any jurisdiction. We may never generate the necessary data or results required to obtain regulatory approval of any other products with the market potential sufficient to enable us to achieve profitability. We have only limited experience in submitting and supporting the applications necessary to gain marketing approvals and have relied on, and expect to continue to rely on, third-party consultants and vendors to assist us in this process. Securing marketing approval requires the submission of extensive preclinical and clinical data and supporting information to regulatory authorities for each therapeutic indication to establish the product candidate's safety and efficacy. Securing marketing approval also requires the submission of information about the product manufacturing process to, and inspection of manufacturing facilities by, the regulatory authorities. The FDA or other regulatory authorities may determine that any product candidate that we develop is not effective, is only moderately effective, is not safe or has undesirable or unintended side effects, toxicities or other characteristics that preclude our obtaining marketing approval or prevent or limit commercial use.

The process of obtaining marketing approvals, both in the United States and abroad, is expensive, may take many years, if approval is obtained at all, and can vary substantially based upon a variety of factors, including the type, complexity and novelty of the product candidates involved. Changes in marketing approval policies during the development period, changes in or the enactment of additional statutes or regulations, or changes in regulatory review for each submitted product application, may cause delays in the approval or rejection of an application. Regulatory authorities have substantial discretion in the approval process and may refuse to accept any application or may decide that our data are insufficient for approval and require additional preclinical, clinical or other studies. In addition, varying interpretations of the data obtained from preclinical and clinical testing could delay, limit or prevent marketing approval of a product candidate.

In addition, disruptions at the FDA and other agencies may prolong the time necessary for new drugs to be reviewed and/or approved by necessary government agencies, which would adversely affect our business. For example, over the last several years, the U.S. government has shut down several times and certain regulatory agencies, such as the FDA, have had to furlough critical employees and stop critical activities. If a prolonged government shutdown occurs, it could significantly impact the ability of the FDA to timely review and process our regulatory submissions, which could have a material adverse effect on our business.

If we experience delays in obtaining approval or if we fail to obtain approval of any product candidate that we develop, the commercial prospects for such product candidate may be harmed and our ability to generate revenues will be materially impaired.

# Failure to obtain marketing approval in foreign jurisdictions would prevent our product candidates from being marketed abroad.

In order to market and sell EYSUVIS, INVELTYS or our product candidates in the European Union and many other jurisdictions, we or our potential third-party collaborators, must obtain separate marketing approvals and comply with numerous and varying regulatory requirements. The approval procedure varies among countries and can involve additional testing. Our Phase 3 clinical trials of EYSUVIS, INVELTYS or any product candidate may not be sufficient to support an application for marketing approval outside the United States.

The time required to obtain approval outside of the United States may differ substantially from that required to obtain FDA approval. The regulatory approval process outside the United States generally includes all of the risks associated with obtaining FDA approval. In addition, in many countries outside the United States, it is required that the product be approved for reimbursement before the product can be sold in that country. We or our potential collaborators may not obtain approvals from regulatory authorities outside the United States on a timely basis, if at all. Approval by the FDA does not ensure approval by regulatory authorities in other countries or jurisdictions, and approval by one regulatory authority outside the United States does not ensure approval by regulatory authorities in other countries or jurisdictions or by the FDA. However, a failure or delay in obtaining regulatory approval in one country may have a negative effect on the regulatory process in other countries. We may not be able to file for marketing approvals and may not receive necessary approvals to commercialize our products in any market, which could significantly and materially harm our business.

Additionally, we could face heightened risks with respect to seeking marketing approval in the United Kingdom as a result of the recent withdrawal of the United Kingdom from the European Union, commonly referred to as Brexit. Pursuant to the formal withdrawal arrangements agreed between the United Kingdom and the European Union, the United Kingdom withdrew from the European Union, effective December 31, 2020. On December 24, 2020, the United Kingdom and European Union entered into a trade and cooperation agreement. The agreement sets out certain procedures for approval and recognition of medical products in each jurisdiction. Any delay in obtaining, or an inability to obtain, any marketing approvals, as a result of Brexit or otherwise, may force us to restrict or delay efforts to seek regulatory approval in the United Kingdom and/or European Union for EYSUVIS, INVELTYS or our product candidates, which could significantly and materially harm our business.

The terms of approvals, ongoing regulations and post-marketing restrictions for our products may limit how we manufacture and market our products, which could materially impair our ability to generate revenue.

Once marketing approval has been granted, an approved product and its manufacturer and marketer are subject to ongoing review and extensive regulation. We, and any potential collaborators we may have in the future, must therefore comply with requirements concerning advertising and promotion for EYSUVIS, INVELTYS or for any of our products for which we obtain marketing approval. Promotional communications with respect to drug products and medical devices are subject to a variety of legal and regulatory restrictions and must be consistent with the information in the product's approved labeling. Thus, we are limited to promoting EYSUVIS and INVELTYS in accordance with their approved labels and the accompanying label may limit the approved use of any other product for which we obtain marketing approval, which could limit sales of such product.

The FDA may also impose requirements for costly post-marketing testing and surveillance to monitor the safety or efficacy of the product, including the adoption and implementation of risk evaluation and mitigation strategies. The FDA closely regulates the post-approval marketing and promotion of drugs to ensure drugs are marketed only for the approved indications and in accordance with the provisions of the approved labeling and regulatory requirements. The FDA imposes stringent restrictions on manufacturers' communications regarding off-label use and if we do not restrict the marketing of our products only to their approved indications, we may be subject to enforcement action for off-label marketing. Violations of the Federal Food, Drug, and Cosmetic Act relating to the promotion of prescription drugs or the promotion or manufacturing of drug products or medical devices may lead to investigations by the FDA, Department of Justice and state Attorneys General alleging violations of federal and state healthcare fraud and abuse laws, as well as state consumer protection laws.

In addition, later discovery of previously unknown adverse events or other problems with our products, manufacturers or manufacturing processes, or failure to comply with regulatory requirements, may have various consequences, including:

- restrictions on such products, manufacturers or manufacturing processes;
- restrictions and warnings in the labeling and marketing of a product;
- restrictions on product distribution or use;
- requirements to conduct post-marketing clinical trials;
- warning or untitled letters;
- withdrawal of the products from the market;
- refusal to approve pending applications or supplements to approved applications that we submit;
- recall of products;
- fines, restitution or disgorgement of profits or revenue;
- suspension or withdrawal of marketing approvals;
- refusal to permit the import or export of our products;
- product seizure;
- exclusion and debarment from federal healthcare reimbursement programs; or
- injunctions or the imposition of civil or criminal penalties.

Non-compliance with European Union requirements or laws of other countries regarding safety monitoring or pharmacovigilance can also result in significant financial penalties. Similarly, failure to comply with the European Union's or other countries' requirements regarding the protection of personal information can lead to significant penalties and sanctions.

In addition, manufacturers of approved products and those manufacturers' facilities are required to comply with extensive FDA requirements, including ensuring that quality control and manufacturing procedures conform to cGMPs applicable to drug manufacturers or quality assurance standards applicable to medical device manufacturers, which include requirements relating to quality control and quality assurance as well as the corresponding maintenance of records and documentation and reporting requirements. We, our contract manufacturers, any contract manufacturers we

may engage in the future, our future collaborators and their contract manufacturers will also be subject to other regulatory requirements, including submissions of safety and other post-marketing information and reports, registration and listing requirements, requirements regarding the distribution of samples to clinicians, recordkeeping, and costly post-marketing studies or clinical trials and surveillance to monitor the safety or efficacy of the product such as the requirement to implement a risk evaluation and mitigation strategy.

We may be subject to substantial penalties if we fail to comply with regulatory requirements or if we experience unanticipated problems with our products.

Our relationships with customers and third-party payors may be subject, directly or indirectly, to applicable anti-kickback, fraud and abuse, false claims, transparency, health information privacy and security, and other healthcare laws and regulations, which could expose us to criminal sanctions, civil penalties, contractual damages, reputational harm, administrative burdens and diminished profits and future earnings.

Healthcare providers, clinicians and third-party payors in the United States and elsewhere play a primary role in the recommendation and prescription and use of EYSUVIS and INVELTYS, and will play a primary role in the recommendation and prescription and use of any product candidates for which we obtain marketing approval. Our arrangements with third-party payors and customers may expose us to broadly applicable fraud and abuse and other healthcare laws and regulations that may constrain the business or financial arrangements and relationships through which we market, sell and distribute EYSUVIS and INVELTYS and any products for which we obtain marketing approval. The applicable federal, state and foreign healthcare laws and regulations that may affect our ability to operate include:

- the federal Anti-Kickback Statute, which prohibits, among other things, persons from knowingly and
  willfully soliciting, offering, receiving or providing remuneration, directly or indirectly, in cash or in kind, to
  induce or reward, or in return for, either the referral of an individual for, or the purchase, order or
  recommendation of, any good or service, for which payment may be made under a federal healthcare
  program such as Medicare and Medicaid;
- federal civil and criminal false claims laws and civil monetary penalty laws, including the federal False
  Claims Act, which impose criminal and civil penalties, including civil whistleblower or *qui tam* actions,
  against individuals or entities for knowingly presenting, or causing to be presented, to the federal
  government, including the Medicare and Medicaid programs, claims for payment that are false or fraudulent
  or making a false statement to avoid, decrease or conceal an obligation to pay money to the federal
  government;
- the federal Health Insurance Portability and Accountability Act of 1996, or HIPAA, which imposes criminal
  and civil liability for executing a scheme to defraud any healthcare benefit program or making false
  statements relating to healthcare matters;
- HIPAA, as amended by the Health Information Technology for Economic and Clinical Health Act of 2009, and their respective implementing regulations, which imposes obligations, including mandatory contractual terms, on covered healthcare providers, health plans and healthcare clearinghouses, as well as their business associates, with respect to safeguarding the privacy, security and transmission of individually identifiable health information; and
- analogous state and foreign laws and regulations, such as state anti-kickback and false claims laws, which may apply to sales or marketing arrangements and claims involving healthcare items or services reimbursed by non-governmental third-party payors, including private insurers, state and foreign laws that require pharmaceutical companies to comply with the pharmaceutical industry's voluntary compliance guidelines and the relevant compliance guidance promulgated by the federal government or otherwise restrict payments that may be made to healthcare providers, state and foreign laws that require drug manufacturers to report information related to payments and other transfers of value to clinicians and other healthcare providers or marketing expenditures, and state and foreign laws governing the privacy and security of health information in certain circumstances, many of which differ from each other in significant ways and often are not preempted by HIPAA, thus complicating compliance efforts.

If our operations are found to be in violation of any of the laws described above or any governmental regulations that apply to us, we may be subject to penalties, including civil and criminal penalties, damages, fines, individual imprisonment, integrity obligations, and the curtailment or restructuring of our operations. Any penalties, damages, fines, individual imprisonment, integrity obligations, exclusion from funded healthcare programs, or curtailment or restructuring of our operations could adversely affect our financial results. Our corporate compliance program is designed to ensure that we will develop, market and sell our products and product candidates in compliance with all applicable laws and regulations, but we cannot guarantee that this program will protect us from governmental investigations or other actions or lawsuits stemming from a failure to be in compliance with such laws or regulations. If any such actions are instituted against us and we are not successful in defending ourselves or asserting our rights, those actions could have a significant impact on our business, including the imposition of significant fines or other sanctions.

Efforts to ensure that our business arrangements with third parties will comply with applicable healthcare laws and regulations may involve substantial costs. It is possible that governmental authorities will conclude that our business practices may not comply with current or future statutes, regulations or case law involving applicable fraud and abuse or other healthcare laws and regulations. If our operations are found to be in violation of any of these laws or any other governmental regulations that may apply to us, we may be subject to significant civil, criminal and administrative penalties, including, without limitation, damages, fines, imprisonment, exclusion from participation in government funded healthcare programs, such as Medicare and Medicaid, and the curtailment or restructuring of our operations. If any of the clinicians or other healthcare providers or entities with whom we do or expect to do business is found to be not in compliance with applicable laws, it may be subject to criminal, civil or administrative sanctions, including exclusions from participation in government funded healthcare programs.

Recently enacted and future legislation may affect our ability to commercialize our products and product candidates and the prices we obtain for any products that are approved in the United States or foreign jurisdictions.

In the United States and some foreign jurisdictions, there have been a number of legislative and regulatory changes and proposed changes regarding the healthcare system that could affect our ability to profitably sell or commercialize EYSUVIS, INVELTYS or any product candidate for which we obtain marketing approval. The pharmaceutical industry has been a particular focus of these efforts and have been significantly affected by legislative initiatives. Current laws, as well as other healthcare reform measures that may be adopted in the future, may result in more rigorous coverage criteria and in additional downward pressure on the price that we receive for any FDA approved product.

In the United States, the Medicare Prescription Drug, Improvement, and Modernization Act of 2003, or the Medicare Modernization Act, changed the way Medicare covers and pays for pharmaceutical products. The legislation expanded Medicare coverage for drug purchases by the elderly and introduced a new reimbursement methodology based on average sales prices for clinician administered drugs. In addition, this legislation provided authority for limiting the number of drugs that will be covered in any therapeutic class. Cost reduction initiatives and other provisions of this legislation could decrease the coverage and price that we receive for any approved products. While the Medicare Modernization Act applies only to drug benefits for Medicare beneficiaries, private payors often follow Medicare coverage policy and payment limitations in setting their own reimbursement rates. Therefore, any reduction in reimbursement that results from the Medicare Modernization Act may result in a similar reduction in payments from private payors.

In March 2010, President Obama signed into law the Patient Protection and Affordable Care Act, as amended by the Health Care and Education Affordability Reconciliation Act, or collectively the ACA. The Budget Control Act of 2011, among other things, led to aggregate reductions to Medicare payments to providers of up to 2% per fiscal year that started in 2013 and, due to subsequent legislative amendments to the statute, will stay in effect through 2029 unless additional congressional action is taken. The Coronavirus Aid, Relief, and Economic Security Act, or the CARES Act, which was enacted on March 27, 2020, suspended the 2% Medicare sequester from May 1, 2020 through December 31, 2020, and extended the sequester by one year, through 2030. The American Taxpayer Relief Act of 2012, among other things, reduced Medicare payments to several types of providers and increased the statute of limitations period for the government to recover overpayments to providers from three to five years. These laws may result in additional reductions in Medicare and other healthcare funding and otherwise affect the prices we may obtain for EYSUVIS, INVELTYS and for any of our product candidates for which we may obtain regulatory approval or the frequency with which EYSUVIS, INVELTYS or any product candidate is prescribed or used.

We expect that additional healthcare reforms may result in additional reductions in Medicare and other healthcare funding, more rigorous coverage criteria, new payment methodologies and additional downward pressure on the price that we receive for EYSUVIS, INVELTYS or any other approved product and/or the level of reimbursement physicians receive for administering any approved product we might bring to market. Reductions in reimbursement levels may negatively impact the prices we receive or the frequency with which our products are prescribed or administered. Any reduction in reimbursement from Medicare or other government programs may result in a similar reduction in payments from private payors.

Since enactment of the ACA, there have been numerous legal challenges and Congressional actions to repeal and replace provisions of the law. For example, with enactment of the Tax Cuts and Jobs Act of 2017, or the 2017 Tax Act, which was signed by President Trump on December 22, 2017, Congress repealed the "individual mandate." The repeal of this provision, which required most Americans to carry a minimal level of health insurance, became effective in 2019. According to the Congressional Budget Office, the repeal of the individual mandate will cause 13 million fewer Americans to be insured in 2027 and premiums in insurance markets may rise. The Trump Administration also took executive actions to undermine or delay implementation of the ACA, but those were rescinded by the Biden Administration. In addition, the Centers for Medicare & Medicaid Services has proposed regulations that would give states greater flexibility in setting benchmarks for insurers in the individual and small group marketplaces, which may have the effect of relaxing the essential health benefits required under the ACA for plans sold through such marketplaces.

The costs of prescription pharmaceuticals in the United States has also been the subject of considerable discussion in the United States, and members of Congress and the Biden Administration have stated that they will address such costs through new legislative and administrative measures. The pricing of prescription pharmaceuticals is also subject to governmental control outside the United States. In these countries, pricing negotiations with governmental authorities can take considerable time after the receipt of marketing approval for a product. To obtain reimbursement or pricing approval in some countries, we may be required to conduct a clinical trial that compares the cost-effectiveness of our product candidates to other available therapies. If reimbursement of our products is unavailable or limited in scope or amount, or if pricing is set at unsatisfactory levels, our ability to generate revenues and become profitable could be impaired.

At the state level, individual states are increasingly aggressive in passing legislation and implementing regulations designed to control pharmaceutical and biological product pricing, including price or patient reimbursement constraints, discounts, restrictions on certain product access and marketing cost disclosure and transparency measures, and, in some cases, designed to encourage importation from other countries and bulk purchasing. In addition, regional health care authorities and individual hospitals are increasingly using bidding procedures to determine what pharmaceutical products and which suppliers will be included in their prescription drug and other health care programs. These measures could reduce the ultimate demand for our products, once approved, or put pressure on our product pricing. We expect that additional state and federal healthcare reform measures will be adopted in the future, any of which could limit the amounts that federal and state governments will pay for healthcare products and services, which could result in reduced demand for our product candidates or additional pricing pressures.

If we or any third-party manufacturers we engage fail to comply with environmental, health and safety laws and regulations, we could become subject to fines or penalties or incur significant costs.

We and any third-party manufacturers we engage or may engage are subject to numerous environmental, health and safety laws and regulations, including those governing laboratory procedures and the handling, use, storage, treatment and disposal of hazardous materials and wastes. From time to time and in the future, our operations may involve the use of hazardous materials, including chemicals and biological materials, and produce hazardous waste products. We generally contract with third parties for the disposal of these materials and wastes. We cannot eliminate the risk of contamination or injury from these materials. In the event of contamination or injury resulting from our use of hazardous materials, we could be held liable for any resulting damages, and any liability could exceed our resources. We also could incur significant costs associated with civil or criminal fines and penalties for failure to comply with such laws and regulations.

Although we maintain general liability insurance as well as workers' compensation insurance to cover us for costs and expenses we may incur due to injuries to our employees resulting from the use of hazardous materials, this insurance may not provide adequate coverage against potential liabilities. We hold \$3.0 million of environmental liability insurance for claims that may be asserted against us in connection with our storage or disposal of biological, hazardous or radioactive materials. These limits, both in the aggregate and per incident, may not be adequate to cover all liabilities that we may incur.

In addition, we may incur substantial costs in order to comply with current or future environmental, health and safety laws and regulations. These current or future laws and regulations may impair our research, development or production efforts. Our failure to comply with these laws and regulations also may result in substantial fines, penalties or other sanctions.

Further, with respect to the operations of any future third-party contract manufacturers, it is possible that if they fail to operate in compliance with applicable environmental, health and safety laws and regulations or properly dispose of wastes associated with our products, we could be held liable for any resulting damages, suffer reputational harm or experience a disruption in the manufacture and supply of our product candidates or products.

We are subject to anti-corruption laws, as well as export control laws, customs laws, sanctions laws and other laws governing our operations. If we fail to comply with these laws, we could be subject to civil or criminal penalties, other remedial measures and legal expenses, be precluded from developing manufacturing and selling certain products outside the United States or be required to develop and implement costly compliance programs, which could adversely affect our business, results of operations and financial condition.

Our operations are subject to anti-corruption laws, including the U.S. Foreign Corrupt Practices Act, or FCPA, the U.K. Bribery Act 2010, or Bribery Act, and other anti-corruption laws that apply in countries where we do business and may do business in the future. The FCPA, Bribery Act and these other laws generally prohibit us, our officers, and our employees and intermediaries from bribing, being bribed or making other prohibited payments to government officials or other persons to obtain or retain business or gain some other business advantage. Compliance with the FCPA, in particular, is expensive and difficult, particularly in countries in which corruption is a recognized problem. In addition, the FCPA presents particular challenges in the pharmaceutical industry, because, in many countries, hospitals are operated by the government, and doctors and other hospital employees are considered foreign officials. Certain payments to hospitals in connection with clinical trials and other work have been deemed to be improper payments to government officials and have led to FCPA enforcement actions.

We may in the future operate in jurisdictions that pose a high risk of potential FCPA or Bribery Act violations, and we may participate in collaborations and relationships with third parties whose actions could potentially subject us to liability under the FCPA, Bribery Act or local anti-corruption laws. In addition, we cannot predict the nature, scope or effect of future regulatory requirements to which our international operations might be subject or the manner in which existing laws might be administered or interpreted. If we expand our operations outside of the United States, we will need to dedicate additional resources to comply with numerous laws and regulations in each jurisdiction in which we plan to operate.

We are also subject to other laws and regulations governing our international operations, including regulations administered by the governments of the United Kingdom and the United States, and authorities in the European Union, including applicable export control regulations, economic sanctions on countries and persons, customs requirements and currency exchange regulations, collectively referred to as the Trade Control laws. In addition, various laws, regulations and executive orders also restrict the use and dissemination outside of the United States, or the sharing with certain non-U.S. nationals, of information classified for national security purposes, as well as certain products and technical data relating to those products. If we expand our presence outside of the United States, it will require us to dedicate additional resources to comply with these laws, and these laws may preclude us from developing, manufacturing, or selling certain products and product candidates outside of the United States, which could limit our growth potential and increase our development

There is no assurance that we will be completely effective in ensuring our compliance with all applicable anti-corruption laws, including the FCPA, the Bribery Act or other legal requirements, including Trade Control laws. If we are not in compliance with the FCPA, Bribery Act and other anti-corruption laws or Trade Control laws, we may be subject to criminal and civil penalties, disgorgement and other sanctions and remedial measures, and legal expenses, which could have an adverse impact on our business, financial condition, results of operations and liquidity. The SEC also may suspend or bar issuers from trading securities on U.S. exchanges for violations of the FCPA's accounting provisions. Any investigation of any potential violations of the FCPA, the Bribery Act, other anti-corruption laws or Trade Control laws by U.S., U.K. or other authorities could also have an adverse impact on our reputation, our business, results of operations and financial condition.

## We might not be able to utilize a significant portion of our net operating loss carryforwards and research and development tax credit carryforwards.

As of December 31, 2020, we had federal net operating loss, or NOL, carryforwards of \$243.2 million, which may be available to offset future federal tax liabilities and expire at various dates beginning in 2030. As of December 31, 2020, we also had state NOL carryforwards of \$215.0 million, which may be available to offset future state income tax liabilities and expire at various dates beginning in 2030, and federal and state research and development credit carryforwards of approximately \$2.4 million, which begin to expire in 2039 (federal) and 2034 (state). These NOL carryforwards could expire unused and be unavailable to offset our future income tax liabilities.

In general, under Sections 382 and 383 of the Code, the amount of benefits from our NOL and research and development tax credit carryforwards, respectively, may be impaired or limited if we incur an "ownership change," generally defined as a greater than 50% change (by value) in our equity ownership by certain stockholders, over a three-year period. We previously completed an analysis and determined that an ownership change has materially limited our net operating loss carryforwards and research and development tax credits available to offset future tax liabilities. We may be further limited by any changes that may have occurred or may occur subsequent to December 31, 2020. Any such limitations may result in greater tax liabilities than we would incur in the absence of such limitations and increased liabilities could adversely affect our business, results of operations, financial position and cash flows. If our ability to use our historical NOL and research and development tax credit carryforwards is materially limited, it would harm our future operating results by effectively increasing our future tax obligations.

There is also a risk that due to regulatory changes, such as suspensions on the use of NOLs, or other unforeseen reasons, our existing NOLs and research and development tax credit carryforwards could expire or otherwise become unavailable to offset future income tax liabilities. As described below in "Changes in tax laws or in their implementation or interpretation could adversely affect our business and financial condition," the 2017 Tax Act, as amended by the CARES Act, includes changes to U.S. federal tax rates and the rules governing NOL carryforwards that may significantly impact our ability to utilize our NOLs to offset taxable income in the future. In addition, state NOLs generated in one state cannot be used to offset income generated in another state. For these reasons, even if we attain profitability, we may be unable to use a material portion of our NOLs and other tax attributes.

#### Risks Related to Employee Matters and Managing Growth

## Our future success depends on our ability to retain key executives and to attract, retain and motivate qualified personnel.

We are highly dependent on the research and development, clinical, business development and commercialization expertise of Mark Iwicki, our President and Chief Executive Officer, Todd Bazemore, our Chief Operating Officer, Mary Reumuth, our Chief Financial Officer, Kim Brazzell, Ph.D., our Chief Medical Officer, Hongming Chen, Sc.D., our Chief Scientific Officer, and Eric Trachtenberg, our General Counsel, Chief Compliance Officer and Corporate Secretary, as well as the other principal members of our management, scientific, clinical and commercial teams. Although we have entered into employment agreements with our executive officers, each of them may terminate their employment with us at any time. We do not maintain "key person" insurance for any of our executives or other employees.

Recruiting and retaining qualified scientific, clinical, manufacturing, accounting, legal and sales and marketing personnel will also be critical to our success. The loss of the services of our executive officers or other key employees could impede the achievement of our research, development and commercialization objectives and seriously harm our ability to successfully implement our business strategy. Furthermore, replacing executive officers and key employees may be difficult and may take an extended period of time because of the limited number of individuals in our industry with the breadth of skills and experience required to successfully develop, gain regulatory approval of and commercialize products. Competition to hire from this limited pool is intense, and we may be unable to hire, train, retain or motivate these key personnel on acceptable terms given the competition among numerous pharmaceutical and biotechnology companies for similar personnel. We also experience competition for the hiring of scientific and clinical personnel from universities and research institutions. In addition, we rely on consultants and advisors, including scientific and clinical advisors, to assist us in formulating our research and development and commercialization strategy. Our consultants and advisors may be employed by employers other than us and may have commitments under consulting or advisory contracts with other entities that may limit their availability to us. If we are unable to continue to attract and retain high quality personnel, our ability to pursue our growth strategy will be limited.

We have expanded and may continue to expand our development, regulatory, commercial and manufacturing capabilities and are continuing to implement sales, marketing and distribution capabilities, and as a result, we may encounter difficulties in managing our growth, which could disrupt our operations.

We have experienced and expect to continue experiencing significant growth in the number of our employees and the scope of our operations, particularly in the areas of drug development, clinical, regulatory affairs, manufacturing, sales, marketing and distribution. For example, in the fourth quarter of 2020, we increased our sales force from 56 TSMs to 91 TSMs, from seven RSLs to 14 RSLs and added two new area sales leaders. In 2021, we increased our sales force from 91 TSMs to 105 TSMs around the start of the third quarter, with a subsequent expansion up to 125 TSMs expected by yearend, subject to continued growth in payor coverage and the status of the COVID-19. To manage our recent, planned and potential future growth, we must continue to implement and improve our managerial, operational and financial systems, and may further expand our facilities and continue to recruit and train additional qualified personnel. Due to our limited financial resources and our limited experience in managing such growth, we may not be able to effectively manage our recently expanded operations, planned sales force expansion or any future expansion of our operations or recruit and train additional qualified personnel. The expansion of our operations may lead to significant costs and may divert our management and business development resources. Furthermore, operational and other restrictions related to COVID-19 may further hamper our ability to grow as needed, including our planned sales force expansion, and/or to manage our growth. Any inability to manage growth could delay the execution of our business plans or disrupt our operations.

Our internal computer systems, or those of our vendors, contractors or consultants, may fail or suffer security breaches, which could result in a material disruption of our product development programs and commercialization of our products.

Despite the implementation of security measures, our internal computer systems and those of our current and any future vendors, contractors or consultants, including any collaborator, are vulnerable to damage from cyber-attacks, computer viruses, worms and other destructive or disruptive software, unauthorized access, natural disasters, terrorism, war and telecommunication and electrical failures. Cyber incidents or attacks could include the deployment of harmful malware, ransomware, denial-of-service attacks, unauthorized access to or deletion of files, social engineering and other means to affect service reliability and threaten the confidentiality, integrity and availability of information. Cyber incidents also could include phishing attempts or e-mail fraud to cause payments or information to be transmitted to an unintended recipient. System failures, accidents, cyberattacks or security breaches could cause interruptions in our operations, it could result in a material disruption of our development programs, the commercialization of our products and our business operations, whether due to a loss of our trade secrets or other proprietary information or other similar disruptions, in addition to possibly requiring substantial expenditures of resources to remedy. The loss of clinical trial data from completed or future clinical trials could result in delays in our regulatory approval efforts and significantly increase our costs to recover or reproduce the data. To the extent that any disruption or security breach were to result in a loss of, or damage to, our data or applications, or inappropriate disclosure of confidential, personal or proprietary information, we could incur liability, including civil fines and penalties under the General Data Protection Regulation

(EU) 2016/679, HIPAA and other relevant state and federal privacy laws in the United States and abroad, our competitive position could be harmed and the further development and commercialization of our product candidates could be delayed. In addition, we may not have adequate insurance coverage to provide compensation for any losses associated with such events.

While we have not experienced any material losses relating to cyber-attacks, we have been the subject of a successful phishing attempt. We could be subject to risks caused by misappropriation, misuse, leakage, falsification or intentional or accidental release or loss of information maintained in the information systems and networks of our company, including personal information of our employees. In addition, outside parties may attempt to penetrate our systems or those of our vendors, contractors or consultants or fraudulently induce our employees or employees of our vendors, contractors or consultants to disclose sensitive information in order to gain access to our data. Like other companies, we may experience threats to our data and systems, including malicious codes and viruses, and other cyber-attacks. The number and complexity of these threats continue to increase over time. If a material breach of our security or that of our vendors, contractors or consultants occurs, the market perception of the effectiveness of our security measures could be harmed, we could lose business and our reputation and credibility could be damaged. We could be required to expend significant amounts of money and other resources to repair or replace information systems or networks. Although we develop and maintain systems and controls designed to prevent these events from occurring, and we have a process to identify and mitigate threats, the development and maintenance of these systems, controls and processes is costly and requires ongoing monitoring and updating as technologies change and efforts to overcome security measures become more sophisticated. Moreover, despite our efforts, the possibility of these events occurring cannot be eliminated entirely.

#### **Risks Related to Our Common Stock**

Our executive officers and directors and their affiliates, if they choose to act together, will continue to have the ability to significantly influence all matters submitted to stockholders for approval.

As of June 30, 2021, our executive officers and directors and their affiliates in the aggregate, owned shares representing approximately 25.92% of our capital stock, based on the most recent institutional stockholder ownership filings with the SEC. As a result, if these stockholders were to choose to act together, they may be able to significantly influence all matters submitted to our stockholders for approval, as well as our management and affairs. For example, these persons, if they choose to act together, could significantly influence the election of directors and approval of any merger, consolidation or sale of all or substantially all of our assets.

This concentration of voting power may:

- delay, defer or prevent a change in control;
- entrench our management and our board of directors; or
- delay or prevent a merger, consolidation, takeover or other business combination involving us on terms that
  other stockholders may desire.

Provisions in our corporate charter documents and under Delaware law could make an acquisition of our company, which may be beneficial to our stockholders, more difficult and may prevent attempts by our stockholders to replace or remove our current management.

Provisions in our certificate of incorporation and our bylaws may discourage, delay or prevent a merger, acquisition or other change in control of our company that stockholders may consider favorable, including transactions in which you might otherwise receive a premium for your shares. These provisions could also limit the price that investors might be willing to pay in the future for shares of our common stock, thereby depressing the market price of our common stock. In addition, because our board of directors are responsible for appointing the members of our management team, these provisions may frustrate or prevent any attempts by our stockholders to replace or remove our

current management by making it more difficult for stockholders to replace members of our board of directors. Among other things, these provisions:

- provide for a classified board of directors such that only one of three classes of directors are elected each
  year;
- allow the authorized number of our directors to be changed only by resolution of our board of directors;
- limit the manner in which stockholders can remove directors from our board of directors;
- provide for advance notice requirements for stockholder proposals that can be acted on at stockholder meetings and nominations to our board of directors;
- require that stockholder actions must be effected at a duly called stockholder meeting and prohibit actions by our stockholders by written consent;
- limit who may call stockholder meetings;
- authorize our board of directors to issue preferred stock without stockholder approval, which could be used
  to institute a "poison pill" that would work to dilute the stock ownership of a potential hostile acquirer,
  effectively preventing acquisitions that have not been approved by our board of directors; and
- require the approval of the holders of at least 75% of the votes that all our stockholders would be entitled to cast to amend or repeal specified provisions of our certificate of incorporation or bylaws.

Moreover, because we are incorporated in Delaware, we are governed by the provisions of Section 203 of the Delaware General Corporation Law, which prohibits a person who owns in excess of 15% of our outstanding voting stock from merging or combining with us for a period of three-years after the date of the transaction in which the person acquired in excess of 15% of our outstanding voting stock, unless the merger or combination is approved in a prescribed manner.

#### An active trading market for our common stock may not be sustained.

Our shares of common stock began trading on The Nasdaq Global Select Market on July 20, 2017. Given the limited trading history of our common stock, there is a risk that an active trading market for our shares will not be sustained, which could put downward pressure on the market price for our common stock and thereby affect your ability to sell your shares. An inactive trading market may also impair our ability to raise capital to continue to fund operations by selling shares and may impair our ability to acquire other companies or technologies by using our shares as consideration.

The price of our common stock may be volatile and fluctuate substantially, which could result in substantial losses for purchasers of our common stock.

Our stock price is likely to be volatile. The stock market in general and the market for smaller biopharmaceutical companies in particular have experienced extreme volatility that has often been unrelated to the operating performance of particular companies. As a result of this volatility, you may not be able to sell your common stock at or above the price you paid for such common stock. The market price for our common stock may be influenced by many factors, including:

- our success in commercializing EYSUVIS, INVELTYS and other product candidates;
- results of clinical trials of any of our product candidates;
- results of clinical trials of product candidates of our competitors;
- changes in the structure of healthcare payment systems;
- the success of competitive products or technologies;
- regulatory or legal developments in the United States and other countries;
- developments or disputes concerning patent applications, issued patents or other proprietary rights;
- the recruitment or departure of key scientific, commercial or management personnel;
- the level of expenses related to the commercialization of EYSUVIS, INVELTYS and clinical development programs for any of our product candidates;
- the results of our efforts to discover, develop, acquire or in-license additional products, product candidates or technologies for the treatment of diseases or conditions, the costs of commercializing any such products and the costs of development of any such product candidates or technologies;
- actual or anticipated changes in estimates as to financial results, development timelines or recommendations by securities analysts;
- variations in our financial results or those of companies that are perceived to be similar to us;
- market conditions in the pharmaceutical and biotechnology sectors;
- the societal and economic impact of public health epidemics, such as the ongoing COVID-19 pandemic;
- general economic, industry and market conditions; and
- the other factors described in this "Risk Factors" section.

In the past, following periods of volatility in the market price of a company's securities, securities class-action litigation has often been instituted against that company. We also may face securities class-action litigation if we fail to successfully commercialize EYSUVIS, INVELTYS or our product candidates. Such litigation, if instituted against us, could cause us to incur substantial costs to defend such claims and divert management's attention and resources.

Sale of a substantial number of shares of our common stock into the market could cause the market price of our common stock to drop significantly, even if our business is doing well.

Sales of a substantial number of shares of our common stock in the public market, or the perception in the market that the holders of a large number of shares intend to sell shares, could reduce the market price of our common stock. As of August 4, 2021, we had outstanding 64,971,096 shares of common stock. Shares of our common stock may be freely sold in the public market at any time to the extent permitted by Rules 144 and 701 under the Securities Act of 1933, as amended, or the Securities Act, or to the extent such shares have already been registered under the Securities Act and are held by non-affiliates of ours. Persons who were our stockholders prior to our initial public offering continue to hold a substantial number of shares of our common stock. If such persons sell, or indicate an intention to sell, substantial amounts of our common stock in the public market, the trading price of our common stock could decline. Moreover, holders of a substantial number of shares of our common stock, including shares of our common stock issuable upon exercise of outstanding warrants and options, have rights, subject to specified conditions, to require us to file registration statements covering their shares or to include their shares in registration statements that we may file for ourselves or other stockholders. We have filed or intend to file registration statements registering all shares of common stock that we may issue under our equity compensation plans or pursuant to equity awards made to newly hired employees outside of equity compensation plans. These shares can be freely sold in the public market upon issuance, subject to volume limitations applicable to affiliates.

We are an "emerging growth company" and a "smaller reporting company", and the reduced disclosure requirements applicable to emerging growth companies and smaller reporting companies may make our common stock less attractive to investors

We are an "emerging growth company," as defined in the Jumpstart Our Business Startups Act of 2012, or the JOBS Act, and may remain an emerging growth company until December 31, 2022, although if the market value of our common stock that is held by non-affiliates exceeds \$700 million as of the prior June 30th or if we have annual gross revenues of \$1.07 billion or more in any fiscal year, we would cease to be an emerging growth company as of December 31 of the applicable year. We also would cease to be an emerging growth company if we issue more than \$1 billion of non-convertible debt over a three-year period.

We are also a "smaller reporting company," as defined in Rule 12b-2 under the Securities Exchange Act of 1934, as amended. We would cease to be a smaller reporting company if we have a public float in excess of \$250 million or have annual revenues in excess of \$100 million and a public float in excess of \$700 million, determined on an annual basis.

As an emerging growth company, we are permitted and intend to rely on exemptions from certain disclosure requirements that are applicable to other public companies that are not emerging growth companies. These exemptions include:

- not being required to comply with the auditor attestation requirements in the assessment of our internal control over financial reporting;
- not being required to comply with any requirement that may be adopted by the Public Company Accounting
  Oversight Board regarding mandatory audit firm rotation or a supplement to the auditor's report providing
  additional information about the audit and the financial statements;
- reduced disclosure obligations regarding executive compensation; and
- exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved.

In addition to the above reduced disclosure requirements applicable to emerging growth companies, as a smaller reporting company, we are permitted and intend to rely on exemptions from certain disclosure requirements that are applicable to other public companies that are not smaller reporting companies. These exemptions include:

- being permitted to provide only two years of audited financial statements in our annual report on Form 10-K, with correspondingly reduced "Management's Discussion and Analysis of Financial Condition and Results of Operations" disclosure;
- not being required to furnish a contractual obligations table in "Management's Discussion and Analysis of Financial Condition and Results of Operations"; and
- not being required to furnish a stock performance graph in our annual report.

We cannot predict whether investors will find our common stock less attractive as a result of our reliance on these exemptions. If some investors find our common stock less attractive as a result, there may be a less active trading market for our common stock and our stock price may be more volatile.

We have incurred and will continue to incur increased costs as a result of operating as a public company, and our management is required to devote substantial time to compliance initiatives and corporate governance practices.

As a public company, and particularly after we are no longer an emerging growth company, we will incur significant legal, accounting and other expenses that we did not incur as a private company. The Sarbanes-Oxley Act of 2002, the Dodd-Frank Wall Street Reform and Consumer Protection Act, the listing requirements of The Nasdaq Global Select Market and other applicable securities rules and regulations impose various requirements on public companies, including establishment and maintenance of effective disclosure and financial controls and corporate governance practices. Our management and other personnel devote a substantial amount of time to these compliance initiatives. Moreover, these rules and regulations have increased our legal and financial compliance costs relative to prior years and will make some activities more time-consuming and costly.

For as long as we remain an emerging growth company or a smaller reporting company, we may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies or smaller reporting companies as described in the preceding risk factor.

Pursuant to Section 404 of the Sarbanes-Oxley Act of 2002, or Section 404, we are required to furnish a report by our management on our internal control over financial reporting. However, while we remain an emerging growth company, we will not be required to include an attestation report on internal control over financial reporting issued by our independent registered public accounting firm. To achieve compliance with Section 404 within the prescribed period, we engaged in a process to document and evaluate our internal control over financial reporting, which is both costly and challenging. In this regard, we will need to continue to dedicate internal resources, potentially engage outside consultants and adopt a detailed work plan to assess and document the adequacy of internal control over financial reporting, continue steps to improve control processes as appropriate, validate through testing that controls are functioning as documented and implement a continuous reporting and improvement process for internal control over financial reporting. Despite our efforts, there is a risk that we will not be able to conclude, within the prescribed timeframe or at all, that our internal control over financial reporting is effective as required by Section 404. If we identify one or more material weaknesses in our internal control over financial reporting, it could result in an adverse reaction in the financial markets due to a loss of confidence in the reliability of our financial statements.

Because we do not anticipate paying any cash dividends on our capital stock in the foreseeable future, capital appreciation, if any, will be your sole source of gain.

We have never declared or paid cash dividends on our capital stock. We currently intend to retain all of our future earnings, if any, to finance the growth and development of our business. In addition, the terms of our Loan Agreement preclude us from paying dividends without the lenders' consent, and any future debt agreements that we may enter into may preclude us from paying dividends without the lenders' consent or at all. As a result, capital appreciation, if any, of our common stock will be your sole source of gain for the foreseeable future.

Our certificate of incorporation designates the state courts in the State of Delaware as the sole and exclusive forum for certain types of actions and proceedings that may be initiated by our stockholders, which could discourage lawsuits against the company and our directors, officers and employees.

Our certificate of incorporation provides that, unless we consent in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware will be the sole and exclusive forum for any derivative action or proceeding brought on our behalf, any action asserting a claim of breach of a fiduciary duty owed by any of our directors, officers or employees to our company or our stockholders, any action asserting a claim against us arising pursuant to any provision of the General Corporation Law of the State of Delaware or our certificate of incorporation or bylaws or as to which the General Corporation Law of the State of Delaware confers jurisdiction on the Court of Chancery of the State of Delaware, or any action asserting a claim against us governed by the internal affairs doctrine. We do not expect this choice of forum provision will apply to suits brought to enforce a duty or liability created by the Securities Act, the Exchange Act, or any other claim for which federal courts have exclusive jurisdiction.

This exclusive forum provision may limit the ability of our stockholders to bring a claim in a judicial forum that such stockholders find favorable for disputes with us or our directors, officers or employees, which may discourage such lawsuits against us and our directors, officers and employees. Alternatively, if a court were to find the choice of forum provision contained in our certificate of incorporation to be inapplicable or unenforceable in an action, we may incur additional costs associated with resolving such action in other jurisdictions, which could materially adversely affect our business, financial condition and operating results.

#### **General Risk Factors**

Changes in tax laws or in their implementation or interpretation could adversely affect our business and financial condition.

Changes in tax law may adversely affect our business or financial condition. On December 22, 2017, President Trump signed into law the 2017 Tax Act, which significantly revised the Internal Revenue Code of 1986, as amended, or the Code. The 2017 Tax Act, among other things, contained significant changes to corporate taxation, including a reduction of the corporate tax rate from a top marginal rate of 35% to a flat rate of 21%, the limitation of the tax deduction for net interest expense to 30% of adjusted earnings (except for certain small businesses), the limitation of the deduction for net operating losses arising in taxable years ending after December 31, 2017 to 80% of current year taxable income and elimination of net operating loss carrybacks for losses arising in taxable years ending after December 31, 2017 (though any such net operating losses may be carried forward indefinitely), the imposition of a one-time taxation of offshore earnings at reduced rates regardless of whether they are repatriated, the elimination of U.S. tax on foreign earnings (subject to certain important exceptions), the allowance of immediate deductions for certain new investments instead of deductions for depreciation expense over time, and the modification or repeal of many business deductions and credits.

As part of Congress's response to the COVID-19 pandemic, the Families First Coronavirus Response Act, or FFCR Act, was enacted on March 18, 2020, the CARES Act was enacted on March 27, 2020 and COVID relief provisions were included in the Consolidated Appropriations Act, 2021, or CAA, which was enacted on December 27, 2020. All contain numerous tax provisions. In particular, the CARES Act retroactively and temporarily (for taxable years beginning before January 1, 2021) suspends application of the 80%-of-income limitation on the use of net operating losses, which was enacted as part of the 2017 Tax Act. It also provides that net operating losses arising in any taxable

year beginning after December 31, 2017, and before January 1, 2021 are generally eligible to be carried back up to five years. The CARES Act also temporarily (for taxable years beginning in 2019 or 2020) relaxes the limitation of the tax deductibility for net interest expense by increasing the limitation from 30% to 50% of adjusted taxable income.

Regulatory guidance under the 2017 Tax Act, the FFCR Act, the CARES Act and the CAA is and continues to be forthcoming, and such guidance could ultimately increase or lessen impact of these laws on our business and financial condition. It is also possible that Congress will enact additional legislation in connection with the COVID-19 pandemic, some of which could have an impact on our company. In addition, it is uncertain if and to what extent various states will conform to the 2017 Tax Act, the FFCR Act, the CARES Act or the CAA.

Patent reform legislation under Leahy-Smith America Invents Act could increase the uncertainties and costs surrounding the prosecution of our patent applications and the enforcement or defense of our issued patents.

On September 16, 2011, Leahy-Smith America Invents Act, or the Leahy-Smith Act, was signed into law. The Leahy-Smith Act includes a number of significant changes to United States patent law. These include provisions that affect the way patent applications are prosecuted and may also affect patent litigation. The United States Patent Office has been developing new regulations and procedures to govern administration of the Leahy-Smith Act, and many of the substantive changes to patent law associated with the Leahy-Smith Act, and in particular, the first to file provisions, only became effective on March 16, 2013. The first to file provisions limit the rights of an inventor to patent an invention if not the first to file an application for patenting that invention, even if such invention was the first invention. Although it is not clear what, if any, impact the Leahy-Smith Act will have on the operation of our business, the Leahy-Smith Act and its implementation could increase the uncertainties and costs surrounding the prosecution of our patent applications and the enforcement or defense of our issued patents, which could have a material adverse effect on our business, financial condition, results of operations and prospects. For example, the Leahy-Smith Act provides a new administrative tribunal known as the Patent Trial and Appeals Board, or PTAB, that provides a venue for companies to challenge the validity of competitor patents at a cost that is much lower than district court litigation and on timelines that are much faster. Although it is not clear what, if any, long term impact the PTAB proceedings will have on the operation of our business, the initial results of patent challenge proceedings before the PTAB since its inception in 2013 have resulted in the invalidation of many U.S. patent claims. The availability of the PTAB as a lower-cost, faster and potentially more potent tribunal for challenging patents could therefore increase the likelihood that our own patents will be challenged, thereby increasing the uncertainties and costs of maintaining, defending and enforcing them.

#### Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

#### **Sales of Unregistered Securities**

On May 14, 2021, we granted stock options to one new employee to purchase 27,500 shares of our common stock at an exercise price of \$5.06 per share and on June 15, 2021, we granted stock options to five new employees to purchase 64,500 shares of our common stock at an exercise price of \$5.70 per share. These options were inducement grants made outside of our 2017 Equity Incentive Plan in accordance with Nasdaq Listing Rule 5635(c)(4) and Section 4(a)(2) of the Securities Act of 1933, as amended. The options have a ten-year term and vest over four years, with 25% of the shares underlying the option award vesting on the one-year anniversary of the applicable employee's new hire date and the remaining 75% of the shares underlying the award vesting monthly thereafter for three-years. Vesting of the options is subject to the employee's continued service with our company through the applicable vesting date. We intend to file a registration statement on a Form S-8 to register the shares of common stock underlying these options prior to the time at which these options become exercisable.

Other than as stated above, we did not sell any shares of our common stock, shares of our preferred stock or warrants to purchase shares of our stock, or restricted stock awards, during the period covered by this Quarterly Report on Form 10-Q that were not registered under the Securities Act of 1933, as amended.

I ise n	f Proceeds	from our	Public	Offering	of Common	Stock
USC U	n riuceeus	II VIII VUI	EUDIC	Officials	OI CAUIIIIIIOII	DIUCK

None.

**Repurchase of Shares or of Company Equity Securities** 

None.

### Item 6. Exhibits

### **Exhibit Index**

EXHIBIT 10.1†	<ul> <li>Loan and Security Agreement, dated May 4, 2021, by and among the Registrant and Oxford Finance LLC, as collateral agent and lender (incorporated by reference to Exhibit 10.1 to the Registrant's Current Report on Form 8-K (File No. 001-38150) filed on May 5, 2021).</li> </ul>
EXHIBIT 10.2†	<ul> <li>Exclusive License Agreement, dated November 10, 2009, by and between the Registrant and The Johns Hopkins University, as amended by the First Amendment dated November 19, 2012, the Second Amendment dated May 23, 2014 and the Third Amendment dated August 26, 2014.</li> </ul>
EXHIBIT 10.3†	- <u>Settlement and License Agreement, dated October 24, 2014, by and between the Registrant and GrayBug, LLC.</u>
EXHIBIT 31.1+	- Certification of Chief Executive Officer pursuant to Rules 13a-14(a) or 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
EXHIBIT 31.2+	- Certification of Chief Financial Officer pursuant to Rules 13a-14(a) or 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
EXHIBIT 32.1++	- <u>Certifications pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of The Sarbanes-Oxley Act of 2002, by Mark Iwicki, President and Chief Executive Officer of the Company.</u>
EXHIBIT 32.2++	<ul> <li>Certifications pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of The Sarbanes-Oxley Act of 2002, by Mary Reumuth, Chief Financial Officer of the Company.</li> </ul>
EXHIBIT 101.INS	- Inline XBRL Instance Document. (the instance document does not appear in the Interactive Data File because XBRL tags are embedded within the Inline XBRL document)
EXHIBIT 101.SCH	- Inline XBRL Taxonomy Extension Schema Document.
EXHIBIT 101.CAL	- Inline XBRL Taxonomy Extension Calculation Linkbase Document.
EXHIBIT 101.DEF	- Inline XBRL Taxonomy Extension Definition Linkbase Document.
EXHIBIT 101.LAB	- Inline XBRL Taxonomy Extension Label Linkbase Document.
EXHIBIT 101.PRE	- Inline XBRL Taxonomy Extension Presentation Linkbase Document.
EXHIBIT 104	- Cover Page Interactive Data File (formatted as Inline XBRL with applicable taxonomy extension information contained in Exhibits 101).

<sup>+</sup> Filed herewith

<sup>++</sup> Furnished herewith † Portions of this exhibit have been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

### KALA PHARMACEUTICALS, INC.

Dated: August 5, 2021 By: /s/ Mark Iwicki

Mark Iwicki

Chairman of the Board, Chief Executive Officer and

President (Principal Executive Officer)

Dated: August 5, 2021 By: /s/ Mary Reumuth

Mary Reumuth

Chief Financial Officer (Principal Financial and

Accounting Officer)

Certain identified information has been excluded from the exhibit because it is both (i) not material and (ii) is the type of information that the registrant treats as private or confidential. Double asterisks denote omissions.

#### **EXCLUSIVE LICENSE AGREEMENT**

#### **BETWEEN**

#### THE JOHNS HOPKINS UNIVERSITY

&

#### HANES NEWCO, INC.

JHU Agreement: [\*\*]

#### LICENSE AGREEMENT

THIS LICENSE AGREEMENT (the "Agreement") is entered into by and between THE JOHNS HOPKINS UNIVERSITY, a Maryland corporation having an address at 3400 N. Charles Street, Baltimore, Maryland, 21218-2695 ("JHU") and Hanes NewCo, Inc., a Delaware corporation having an address at c/o Jeff Wiesen, Mintz Levin Cohn Ferris Glovsky and Popeo PC, One Financial Center, Boston, MA 02111 ("Company"), with respect to the following:

#### **RECITALS**

WHEREAS, as a center for research and education, JHU is interested in licensing PATENT RIGHTS (hereinafter defined) in a manner that will benefit the public by facilitating the distribution of useful products and the utilization of new processes, but is without capacity to commercially develop, manufacture, and distribute any such products or processes; and

WHEREAS, a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*], and a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*], and a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*], and a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*], and a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*] (all hereinafter, "Inventors"); and

WHEREAS, JHU has acquired through assignment all rights, title and interest, with the exception of certain retained rights by the United States Government, in its interest in said valuable inventions; and

WHEREAS, Company desires obtain certain rights in such inventions as herein provided, and to commercially develop, manufacture, use and distribute products and processes based upon or embodying said valuable inventions throughout the world;

NOW THEREFORE, in consideration of the premises and the mutual promises and covenants contained in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto agree as follows:

## ARTICLE 1 DEFINITIONS

All references to particular Exhibits, Articles or Paragraphs shall mean the Exhibits to, and Paragraphs and Articles of, this Agreement, unless otherwise specified. For the purposes of this Agreement and the Exhibits hereto, the following words and phrases shall have the following meanings:

**1.1** "AFFILIATED COMPANY" as used herein in either singular or plural shall mean any corporation, company, partnership, joint venture or other entity, which controls, is controlled by or is under common control with Company. For purposes of this Paragraph 1.1, control shall mean the direct or indirect ownership of at least fifty- percent (50%).

- **1.2 "EFFECTIVE DATE"** of this License Agreement shall mean the date the last party hereto has executed this Agreement.
- 1.3 "EXCLUSIVE LICENSE" shall mean a grant by JHU to Company of its entire right and interest in the PATENT RIGHTS subject to rights retained by the United States Government, if any, in accordance with the Bayh-Dole Act of 1980 (established by P.L. 96-517 and amended by P.L. 98-620, codified at 35 USC § 200 et. seq. and implemented according to 37 CFR Part 401), and subject to the retained right of JHU to make, have made, provide and use for its and The Johns Hopkins Health Systems' non-commercial academic research and teaching purposes LICENSED PRODUCT(S) and LICENSED SERVICE(S), including the ability to distribute any biological material disclosed and/or claimed in PATENT RIGHTS for nonprofit non-commercial academic research use to non-commercial entities as is customary in the scientific community.
- **1.4 "KNOW-HOW AND MATERIALS"** shall mean JHU's interest in proprietary materials, information, records, and data developed by [\*\*] (including incidental information developed in his laboratory) and at his direction by his laboratory personnel, fellows, and students and that are directly related to the use of and practice of the PATENT RIGHTS, and are (a) subject to the Johns Hopkins Intellectual Property Policy and (b) owned by JHU as of the EFFECTIVE DATE or developed within [\*\*] years from the EFFECTIVE DATE and not subject to any third party rights, including without limitation the materials listed on Exhibit C. Know-How specifically excludes patentable inventions.
  - **1.5 "LICENSED FIELD"** shall mean all fields of use.
- **1.6** "LICENSED PRODUCT(S)" as used herein in either singular or plural shall mean any process or method, material, compositions, drug, or other product, the manufacture, use or sale of which would constitute, but for the license granted to Company pursuant to this Agreement, an infringement of a VALID CLAIM of PATENT RIGHTS (infringement shall include, but is not limited to, direct, contributory, or inducement to infringe) in the country of sale.
- 1.7 "LICENSED SERVICE(S)" as used herein in either singular or plural shall mean the performance on behalf of a third party of any method or the manufacture of any product or the use of any product or composition which would constitute, but for the license granted to Company pursuant to this Agreement, an infringement of a VALID CLAIM of the PATENT RIGHTS, (infringement shall include, but not be limited to, direct, contributory or inducement to infringe) in the country of performance of the services.
- **1.8** "NET SALES" shall mean gross sales revenues and fees billed by Company and AFFILIATED COMPANY from the sale of LICENSED PRODUCT(S) less (i) trade, quantity or cash discounts allowed, (ii) refunds, credits or allowances for returns, rejections and recalls; (iii) rebates and chargebacks, (iv) sales, use or other taxes and tariffs, duties or other charges levied by a governmental entity on the production, sale, delivery or use of LICENSED PRODUCT(S), and (iv) packing, freight, shipping and insurance charges.

In the event that Company or AFFILIATED COMPANY sells a LICENSED PRODUCT(S) as part of a combination, then:

- (i) in the event that Company or AFFILIATED COMPANY sells in a particular country during a particular year a LICENSED PRODUCT(S) together with other non-therapeutic ingredients or substances or as part of a kit, and Company or AFFILIATED COMPANY also sells such LICENSED PRODUCT(S) in such country in such year separately the NET SALES for purposes of royalty payments shall be based on the sales revenues and fees that would be received from the separate sale of the same quantity of LICENSED PRODUCT as is contained in the combination.
- (ii) in the event that Company or an AFFILIATED COMPANY sells, in a particular country during a particular year, a LICENSED PRODUCT for therapeutic purposes in combination with a therapeutically active ingredient which is not a LICENSED PRODUCT ("Other Items"), the NET SALES for purposes of royalty payments shall be calculated as follows:
  - (a) If all LICENSED PRODUCTS and Other Items contained in the combination are available separately in the particular country during such year, the NET SALES for purposes of royalty payments will be calculated by multiplying the NET SALES of the combination by the fraction A/A+B, where A is the separately available price of all LICENSED PRODUCTS in the combination in the particular country during such year, and B is the separately available price for all Other Items in the combination in the particular country during such year.
  - (b) If the combination includes Other Items which are not sold separately in the particular country during such year (but all LICENSED PRODUCTS contained in the combination are available separately in the particular country during such year), the NET SALES for purposes of royalty payments will be calculated by multiplying the NET SALES of the combination by A/C, where A is as defined above and C is the invoiced price of the combination.
  - (c) If the LICENSED PRODUCTS contained in the combination are not sold separately, the parties agree to negotiate a reduction in the royalty rate to reflect the fair value that the LICENSED PRODUCT attributed to the overall product sold.

The term "Other Items" does not include solvents, diluents, carriers, excipients, buffers or the like used in formulating a product.

- (iii) In no event shall: (1) Company apply the credit in both paragraphs (i) and (ii) above to the same sale of a LICENSED PRODUCT, and (2) the royalty rates be reduced by greater than [\*\*] percent ([\*\*]%) when collectively applying the credits of paragraphs (i) above, (ii) above, and the offset of Paragraph 3 in EXHIBIT A.
- 1.9 "NET SERVICE REVENUES" shall mean gross service revenues and fees billed by Company and AFFILIATED COMPANY for the performance of LICENSED SERVICE(S) less (i) trade, quantity or cash discounts allowed, (ii) refunds, rebates, and chargebacks, (iii) sales or other taxes or charges levied by a governmental entity upon and with specific reference to the LICENSED SERVICE(S). In the event that Company or AFFILIATED COMPANY sells a LICENSED SERVICE(S) that is also sold separately in combination with other services, the NET SERVICE REVENUES for purposes of royalty payments shall be based on the sales revenues that

- 1.10 "PATENT RIGHTS" shall mean the patents and applications listed in EXHIBIT D for JHU Ref #[\*\*] and JHU Ref # [\*\*], any patent applications hereafter filed for JHU Ref # [\*\*] and JHU Ref # [\*\*] (all such patent applications for JHU Ref # [\*\*], JHU Ref # [\*\*], JHU Ref # [\*\*] and JHU Ref # [\*\*] hereinafter referred to as "Patent Applications") and all continuations, divisions, claims of continuations-in-part applications directed to subject matter specifically described in the Patent Applications, continued prosecution applications and reissues, reexaminations, extensions and supplemental protection certificates thereof, and any corresponding foreign patent applications, and any patents, or other equivalent foreign patent rights issuing, granted or registered thereon. "OPTION PATENT RIGHTS" shall mean the patents and applications listed in EXHIBIT D for JHU Ref # [\*\*] (hereinafter referred to as "Option Patent Applications") and all continuations, divisions, claims of continuations-in-part applications directed to subject matter specifically described in the Option Patent Applications, continued prosecution applications and reissues, reexaminations, extensions and supplemental protection certificates thereof, and any corresponding foreign patent applications, and any patents, or other equivalent foreign patent rights issuing, granted or registered thereon. In the event of the exercise by Company of the option set forth in Section 2.5, the OPTION PATENT RIGHTS shall become PATENT RIGHTS for all purposes of this Agreement, except as set forth in Section 2.4.
- "SUBLICENSE CONSIDERATION" shall mean consideration of any kind received by the Company or AFFILIATED COMPANIES from a SUBLICENSEE(S) for the grant of a sublicense under this Agreement, including upfront fees, running royalties on LICENSED PRODUCT(S) and LICENSED SERVICE(S), milestone fees based on SUBLICENSEE achievements, and any premium paid by the SUBLICENSEE(S) over Fair Market Value for stock of the Company or an AFFILIATED COMPANY in consideration for such sublicense; provided that SUBLICENSE CONSIDERATION shall not include amounts paid to the Company or an AFFILIATED COMPANY by the SUBLICENSEE(S) for milestone payments based on Company achievements (but only if (i) Company owes a milestone payment to JHU for that same achievement or (ii) the Company did not receive payments (other than the milestone payment) from the SUBLICENSEE for product development, research work, clinical studies or regulatory approvals performed by or for the Company or AFFILIATED COMPANIES to achieve the event resulting in the milestone payment), loans, equity investments at Fair Market Value, payments for product development, research work, clinical studies and regulatory approvals performed by or for the Company or AFFILIATED COMPANIES (including third parties on their behalf), each pursuant to a specific agreement providing for a work plan and reasonable budget. The term "Fair Market Value" shall mean (i) if the stock is publicly traded, the average price that the stock in question is publicly trading at for [\*\*] days prior to the announcement of its purchase by the SUBLICENSEE(S), (ii) if the stock is not publicly traded, and the Company or AFFILIATED COMPANY has had a private equity financing (including a financing with debt securities convertible to equity) within [\*\*], the value of such stock as determined by the most recent such private financing through a financial investor (an entity whose sole interest in the Company or AFFILIATED COMPANY is financial, including an investment entity owned or controlled by a pharmaceutical company or other operating company that makes an investment separate from a sublicense granted to its parent company) of the Company or AFFILIATED COMPANY that issued the shares, or (iii) if the stock is not publicly traded, and the

Company or AFFILIATED COMPANY has not had a private equity financing (including a financing with debt securities convertible to equity) through a financial investor (as defined above) within [\*\*], a value to be agreed upon by the Company and JHU, or if they fail to agree within [\*\*] days after the sale of the stock to the SUBLICENSEE, by an independent appraisal firm selected by the Company.

- **1.12 "SUBLICENSEE(S)"** as used herein in either singular or plural shall mean any person or entity other than an AFFILIATED COMPANY to which Company or an AFFILIATED COMPANY has granted a sublicense under this Agreement.
- 1.13 "VALID CLAIM" shall mean either: (a) a claim of an issued and unexpired patent included within the PATENT RIGHTS which has not been revoked or held unenforceable, unpatentable or invalid by a decision of a court or other governmental agency of competent jurisdiction, unappealable or unappealed within the time allowed for appeal, and which has not been disclaimed, denied or admitted to be invalid or unenforceable through reexamination, reissue, disclaimer or otherwise; or (b) a claim of a pending patent application included within the PATENT RIGHTS, which claim has not been abandoned or finally disallowed without the possibility of appeal or refiling of such application, and has been pending for less than [\*\*] years from the date such claim was filed in a first national filing non-provisional patent application in the country of interest and has not been (i) canceled, (ii) withdrawn from consideration, (iii) finally determined to be unallowable by the applicable governmental authority (and from which no appeal is or can be taken), or (iv) abandoned.

#### ARTICLE 2 LICENSE GRANT

- **2.1 Grant.** Subject to the terms and conditions of this Agreement, JHU hereby grants to Company
- (i) a world-wide EXCLUSIVE LICENSE to research, develop, make, have made, use, have used, import, have imported, offer for sale, have offered for sale, sell and have sold the LICENSED PRODUCT(S) and to provide and have provided the LICENSED SERVICE(S) in the United States and worldwide under the PATENT RIGHTS in the LICENSED FIELD, and
- (ii) a world-wide nonexclusive license to use the KNOW HOW AND MATERIALS in the LICENSED FIELD, provided that JHU will not grant a license to KNOW HOW AND MATERIALS to any other commercial entity.

This Grant shall apply to the Company and any AFFILIATED COMPANY. If any AFFILIATED COMPANY exercises rights under this Agreement, such AFFILIATED COMPANY shall be bound by all terms and conditions of this Agreement, including but not limited to indemnity and insurance provisions and royalty payments, which shall apply to the exercise of the rights, to the same extent as would apply had this Agreement been directly between JHU and the AFFILIATED COMPANY. In addition, Company shall remain fully liable to JHU for all acts and obligations of AFFILIATED COMPANY such that acts of the AFFILIATED COMPANY shall be considered acts of the Company.

KNOW HOW AND MATERIALS will be transferred from time-to-time by JHU's Inventor, [\*\*], at his discretion, and shall not be sold, invoiced, provided, or otherwise transferred, separately from PATENT RIGHTS.

- **2.2 Sublicense.** Company may sublicense to others under this Agreement, subject to the terms and conditions of this Paragraph 2.2. As a condition to its validity and enforceability, each sublicense agreement shall: (a) incorporate by reference the terms and conditions of this Agreement, (b) be consistent with the terms, conditions and limitations of this Agreement, (c) name JHU as an intended third party beneficiary of the obligations of SUBLICENSEE without imposition of obligation or liability on the part of JHU or its Inventors to the SUBLICENSEE, and (d) specifically incorporate Paragraphs 6.2 "Representations by JHU", 7.1 "Indemnification", 10.1 "Use of Name", 10.4 "Product Liability" into the body of the sublicense agreement, and cause the terms used in therein to have the same meaning as in this Agreement.. Company shall provide to JHU a copy of each fully executed sublicense agreement, within [\*\*] days of execution by both Company and proposed SUBLICENSEE. To the extent that any terms, conditions or limitations of any sublicense agreement are inconsistent with this Agreement, those terms, conditions and limitations are null and void against JHU.
- 2.3 Government Rights. The United States Government may have acquired a nonexclusive, nontransferable, irrevocable, paid-up license to practice or have practiced for or on behalf of the United States the inventions described in PATENT RIGHTS throughout the world. To the extent that the inventions claimed in the PATENT RIGHTS were funded by grants, awards or contracts with the United States government, the rights granted herein are additionally subject to: (i) the requirement that any LICENSED PRODUCT(S) produced for use or sale within the United States shall be substantially manufactured in the United States (unless a waiver under 35 USC § 204 or equivalent is granted by the appropriate United States government agency), (ii) the right of the United States government to require JHU, or its licensees, including Company, to grant sublicenses to responsible applicants on reasonable terms when necessary to fulfill health or safety needs, and, (iii) other rights acquired by the United States government under the laws and regulations applicable to the grant/contract award under which the inventions were made.
- **2.4 Improvements.** Subject to any third party, including U.S. Government, obligations of JHU under any agreement under which any Improvement is made, JHU will notify Company in writing of any Improvement within [\*\*] days of disclosure to the JHU technology transfer office. Company will have a right of first negotiation for [\*\*] days to amend this Agreement to add the Improvement and related patent rights, know how and materials to this Agreement. JHU and Company will negotiate in good faith on reasonable terms for adding the Improvement to this Agreement. For purposes hereof, "Improvement" shall mean an invention (i) made in the laboratory of [\*\*], (ii) that would infringe the PATENT RIGHTS if made used, imported or sold without a license to PATENT RIGHTS, (iii) pertaining to mucosal delivery using mucus penetrating nanoparticles or microparticles, and (iv) reported to the JHU technology transfer office within [\*\*] from the EFFECTIVE DATE; provided, however, that if the OPTION PATENT RIGHTS become PATENT RIGHTS upon exercise of the option set forth in Section 2.5, the OPTION PATENT RIGHTS shall not be included in PATENT RIGHTS for purposes of this Section 2.4.

**2.5 Option for OPTION PATENT RIGHTS.** JHU hereby grants to Company an exclusive option for a period of [\*\*] months from the EFFECTIVE DATE to add the OPTION PATENT RIGHTS to the definition of PATENT RIGHTS hereunder. Such option may be exercised by written notice to JHU at any time during such [\*\*] month period, and upon the giving of such written notice, the OPTION PATENT RIGHTS shall become PATENT RIGHTS for all purposes of this Agreement, except as set forth in Section 2.4.

# ARTICLE 3 FEES, ROYALTIES, & PAYMENTS

- **3.1 License Fee.** Company shall pay to JHU within [\*\*] days of the EFFECTIVE DATE of this Agreement a license fee as set forth in Exhibit A. JHU will not submit an invoice for the license fee, which is nonrefundable and shall not be credited against royalties or other fees.
- **3.2 Minimum Annual Royalties.** Company shall pay to JHU minimum annual royalties as set forth in Exhibit A. These minimum annual royalties shall be due, without invoice from JHU, within [\*\*] days of [\*\*] of each year, commencing with [\*\*] Running royalties accrued under Paragraph 3.3 and percentage of SUBLICENSE CONSIDERATION accrued under Paragraph 3.4 and paid to JHU during the one year period preceding [\*\*] of each year, commencing with [\*\*] shall be credited against the minimum annual royalties due on that anniversary date.
- **3.3 Running Royalties.** Company shall pay to JHU a running royalty as set forth in Exhibit A, for each LICENSED PRODUCT(S) sold, and for each LICENSED SERVICE(S) provided, by Company, AFFILIATED COMPANIES, based on NET SALES and NET SERVICE REVENUES for the term of this Agreement. Such payments shall be made quarterly, as set forth in Section 5.1(a). All non-US taxes related to LICENSED PRODUCT(S) or LICENSED SERVICE(S) sold under this Agreement shall be paid by Company and shall not be deducted from royalty or other payments due to JHU, but shall be deducted from gross sales revenues and fees and gross service revenues and fees in the calculation of NET SALES and NET SERVICE REVENUES to the extent such taxes have been included in gross sales revenues and fees and gross service revenues and fees. JHU shall be responsible for paying any and all taxes (other than withholding taxes or deduction of tax at source required by applicable law to be paid by Company) levied on it by account of its receipt of any payments it receives under this Agreement. If applicable laws require that taxes be withheld or deducted at source from any amounts due to JHU under this Agreement, the Company shall (a) deduct these taxes from the remittable amount, (b) pay the taxes to the proper taxing authority, and (c) deliver to JHU a statement including the amount of tax withheld and justification therefor, and such other information as may be necessary for tax credit purposes. Company shall cooperate with JHU in any action by JHU for a refund of such taxes withheld.

In order to insure JHU the full royalty payments contemplated hereunder, Company agrees that in the event any LICENSED PRODUCT(S) shall be sold by the Company to an AFFILIATED COMPANY, by an AFFILIATED COMPANY to the Company, or among AFFILIATED COMPANIES the royalties to be paid hereunder for such LICENSED PRODUCT(S) shall be based upon the greater of: 1) the NET SALES at which the purchaser of LICENSED PRODUCT(S) resells such product to the end user, 2) the NET SERVICE REVENUES received

from using the LICENSED PRODUCT(S) in providing a service, or 3) the NET SALES of LICENSED PRODUCT(S) paid by the purchaser.

In the event that consideration in lieu of money is received by Company or an AFFILIATED COMPANY from the sale of LICENSED PRODUCT(S), the fair market value of such consideration shall be included in the determination of NET SALES for such sale. Such fair market value shall be determined by the Company or AFFLILIATED COMPANY, as applicable, in good faith.

- **3.4 Sublicense Consideration.** In addition to the running royalty as set forth under Paragraph 3.3, Company shall pay to JHU a percentage of SUBLICENSE CONSIDERATION as set forth in <a href="Exhibit A">Exhibit A</a>. This percentage of SUBLICENSE CONSIDERATION shall be due, without the need for invoice from JHU, within [\*\*] days after the end of each calendar quarter in which sublicense consideration is received.
- **3.5 Milestones.** Company shall pay to JHU milestones as set forth in <u>Exhibit A</u>. These milestones shall be due, without invoice from JHU, within [\*\*] days of achievement of such milestone.
- **3.6 Patent Reimbursement.** Company will reimburse JHU for the costs of preparing, filing, maintaining and prosecuting PATENT RIGHTS incurred prior to the EFFECTIVE DATE ("PRIOR PATENT COSTS") (currently estimated at about \$ [\*\*]). In the event the OPTION PATENT RIGHTS become PATENT RIGHTS upon exercise of the option set forth in Section 2.5, the sum of \$[\*\*] shall be added to the PRIOR PATENT COSTS. Company shall pay without invoice from JHU \$[\*\*] per month until the sooner of (i) all PRIOR PATENT COSTS have been paid or (ii) Company has raised a total of \$[\*\*], at which time the balance of unpaid PRIOR PATENT COSTS shall be due within [\*\*] days.

In accordance with Paragraph 4.1 below, Company will reimburse JHU, within [\*\*] days of the receipt of an invoice from JHU, for all costs associated with the preparation, filing, maintenance, and prosecution of PATENT RIGHTS and the OPTION PATENT RIGHTS incurred by JHU subsequent to the EFFECTIVE DATE of this Agreement.

- **3.7 Form of Payment.** All payments under this Agreement shall be made in U.S. Dollars by either check or wire transfer.
- **3.8 Payment Information.** All check payments from Company to JHU shall be sent to:

Director Johns Hopkins Technology Transfer The Johns Hopkins University 100 N. Charles Street, 5th Floor Baltimore, MD 21201

Attn: JHU Agrmt# [\*\*]

or such other addresses which JHU may designate in writing from time to time. Checks are to be made payable to "The Johns Hopkins University". Wire transfers may be made through:

[\*\*]

[\*\*]

Transit/Routing/ABA number: [\*\*]

SWIFT code: [\*\*]

CHIPS ABA number: [\*\*]
Account Number: [\*\*]
Type of Account: Depository

Reference: [\*\*] (JHU Agrmt. # [\*\*] Attn: Financial Manager

Company shall be responsible for any and all costs associated with wire transfers.

Via ACH

[\*\*]

Transit/routing/ABA number: [\*\*]

Account number: [\*\*]
Type of account: depository

CTX format is preferred; CCD+ is also accepted

3.9 Late Payments. In the event that any payment due hereunder is not made when due, the payment shall accrue interest beginning on the tenth day following the due date thereof, calculated at the annual rate of the sum of (a) [\*\*] percent ([\*\*]%) plus (b) the prime interest rate quoted by The Wall Street Journal on the date said payment is due, the interest being compounded on the last day of each calendar quarter, provided however, that in no event shall said annual interest rate exceed the maximum legal interest rate for corporations. Each such payment when made shall be accompanied by all interest so accrued. Said interest and the payment and acceptance thereof shall not negate or waive the right of JHU to seek any other remedy, legal or equitable, to which it may be entitled because of the delinquency of any payment including, but not limited to termination of this Agreement as set forth in Paragraph 9.2, subject to the cure provisions set forth therein.

# ARTICLE 4 PATENT PROSECUTION, MAINTENANCE, & INFRINGEMENT

**4.1 Prosecution & Maintenance.** JHU, at Company's expense, shall file, prosecute and maintain all patents and patent applications specified under PATENT RIGHTS and, subject to the terms and conditions of this Agreement, Company shall be licensed thereunder. Title to all such patents and patent applications shall reside in JHU. JHU shall have full and complete control over all patent matters in connection therewith under the PATENT RIGHTS, provided however, that JHU shall (a) cause its patent counsel to timely copy Company on all official actions and written correspondence with any patent office, and (b) allow Company an opportunity to comment and advise JHU. JHU shall consider and reasonably incorporate all comments and advice unless

detrimental to JHU's intellectual property rights. By concurrent written notification to JHU and its patent counsel at least [\*\*] days in advance (or later at JHU's discretion) of any filing or response deadline, or fee due date, Company may elect not to have a patent application filed in any particular country or not to pay expenses associated with prosecuting or maintaining any patent application or patent, provided that Company pays for all costs incurred up to JHU's receipt of such notification. Failure to provide such notification can be considered by JHU to be Company's authorization to proceed at Company's expense. Upon such notification, JHU may file, prosecute, and/or maintain such patent applications or patent at its own expense and for its own benefit, and any rights or license granted hereunder held by Company, AFFILIATED COMPANIES or SUBLICENSEE(S) relating to the PATENT RIGHTS which comprise such patent applications or patent and/or apply to the particular country, shall terminate.

- **4.2 Notification.** Each party will notify the other promptly in writing when any infringement by another is uncovered or suspected.
- 4.3 Infringement. Company shall have the first right to enforce any patent within PATENT RIGHTS against any infringement or alleged infringement thereof, and shall at all times keep JHU informed as to the status thereof. Before Company commences an action with respect to any infringement of such patents, Company shall give careful consideration to the views of JHU and to potential effects on the public interest in making its decision whether or not to sue. Thereafter, Company may, at its own expense, institute suit against any such infringer or alleged infringer and control and defend such suit in a manner consistent with the terms and provisions hereof and recover any damages, awards or settlements resulting therefrom, subject to Paragraph 4.5. If required by law, JHU shall permit action under this Section to be brought in its name, including being joined as party-plaintiff. However, no settlement, consent judgment or other voluntary final disposition of the suit that concedes the invalidity or unenforceability of any patent within PATENT RIGHTS may be entered into without the prior written consent of JHU, which consent shall not be unreasonably withheld. This right to sue for infringement shall not be used in an arbitrary or capricious manner. JHU shall reasonably cooperate in any such litigation at Company's expense.

If Company elects not to enforce any patent within the PATENT RIGHTS, then it shall so notify JHU in writing within [\*\*] days of receiving notice that an infringement exists, and JHU may, in its sole judgment and at its own expense, take steps to enforce any patent and control, settle, and defend such suit in a manner consistent with the terms and provisions hereof, and recover, for its own account, any damages, awards or settlements resulting therefrom. However, no settlement, consent judgment or other voluntary final disposition of the suit that concedes the invalidity or unenforceability of any patent within PATENT RIGHTS may be entered into without the prior written consent of Company, which consent shall not be unreasonably withheld.

- **4.4 Patent Invalidity Suit.** If a declaratory judgment action is brought naming Company as a defendant and alleging invalidity of any of the PATENT RIGHTS, JHU may elect to take over the sole defense of the action at its own expense. Company shall cooperate fully with JHU in connection with any such action.
- **4.5 Recovery.** Any recovery by Company under Paragraph 4.3 shall be deemed to reflect loss of commercial sales, and Company shall pay to JHU [\*\*] percent ([\*\*]%) of the

recovery, net of all reasonable costs and expenses associated with each suit or settlement. If the cost and expenses exceed the recovery, then [\*\*] of the excess shall be credited against royalties payable by Company to JHU hereunder in connection with sales of LICENSED PRODUCT covered in the PATENT RIGHTS which are the subject of the infringement suit, in the country of such legal proceedings, provided, however, that any such credit under this Paragraph shall not exceed [\*\*] percent ([\*\*]%) of the royalties otherwise payable to JHU with regard to sales in the country of such action in any one calendar year, with any excess credit being carried forward to future calendar years.

**4.6 Cooperation.** Each party agrees to cooperate in any action under this Article which is controlled by the other party, provided that the controlling party reimburses the cooperating party promptly for any costs and expenses incurred by the cooperating party in connection with providing such assistance.

## ARTICLE 5 OBLIGATIONS OF THE PARTIES

- **5.1 Reports.** Company shall provide to JHU the following written reports according to the following schedules.
- (a) Company shall provide quarterly Royalty Reports, substantially in the format of Exhibit B and due within [\*\*] days of the end of each calendar quarter following the first commercial sale of a LICENSED PRODUCT or LICENSED SERVICE by Company, an AFFILIATED COMPANY or a SUBLICENSEE(S). Royalty Reports shall disclose (i) the amount of LICENSED PRODUCT(S) and LICENSED SERVICE(S) sold, the total NET SALES and NET SERVICE REVENUES of such LICENSED PRODUCT(S) and LICENSED SERVICE(S), and the running royalties due to JHU as a result of NET SALES and NET SERVICE REVENUES by Company and AFFILIATED COMPANIES thereof and (ii) the amount of SUBLICENSE CONSIDERATION received and the percentage thereof payable to JHU pursuant to Section 3.4. Payment of any such royalties and percentage of SUBLICENSE CONSIDERATION due shall accompany such Royalty Reports.
- (b) Until Company, an AFFILIATED COMPANY or a SUBLICENSEE(S) has achieved a first commercial sale of a LICENSED PRODUCT or LICENSED SERVICE, or received FDA market approval, Company shall provide [\*\*] Diligence Reports, due within [\*\*] days of the end of [\*\*] following the EFFECTIVE DATE of this Agreement. These Diligence Reports shall describe Company's, AFFILIATED COMPANIES or any SUBLICENSEE(S)'s technical efforts towards meeting its obligations under the terms of this Agreement.
- (c) Company shall provide [\*\*] Reports within [\*\*] days of the end of [\*\*] following the EFFECTIVE DATE of this Agreement. [\*\*] Reports shall include:
  - (i) evidence of insurance as required under Paragraph 10.4, or, a statement of why such insurance is not currently required, and
  - (ii) identification of all AFFILIATED COMPANIES which have exercised rights pursuant to Paragraph 2.1, or, a statement that no AFFILIATED COMPANY has exercised such rights, and

- (iii) notice of all FDA approvals of any LICENSED PRODUCT(S) or LICENSED SERVICE(S) obtained by COMPANY, AFFILIATED COMPANY or SUBLICENSEE, the patent(s) or patent application(s) licensed under this Agreement upon which such product or service is based, and the commercial name of such product or service, or, in the alternative, a statement that no FDA approvals have been obtained.
- **S.2 Records.** Company shall make and retain, for a period of [\*\*] years following the period of each report required by Paragraph 5.1, true and accurate records, files and books of account containing all the data reasonably required for the full computation and verification of sales and other information required in Paragraph 5.1. Such books and records shall be in accordance with generally accepted accounting principles consistently applied. Company shall permit the inspection of such records, files and books of account by an independent certified public accountant selected by JHU and acceptable to Company in its reasonable judgment during regular business hours upon [\*\*] business days' written notice to Company. Such inspection shall not be made more than [\*\*]. All costs of such inspection shall be paid by JHU, provided that if any such inspection shall reveal that an error has been made in the amount of payments hereunder for any calendar year equal to [\*\*] percent ([\*\*]%) or more of such payments, such costs shall be borne by Company. As a condition to entering into any such agreement, Company shall include in any agreement with its AFFILIATED COMPANIES which permits such party to make, use, sell or import the LICENSED PRODUCT(S) or provide LICENSED SERVICE(S), a provision requiring such party to retain records of sales of LICENSED PRODUCT(S) and records of LICENSED SERVICE(S) and other information as required in Paragraph 5.1 and permit an independent certified public accountant selected by JHU and acceptable to Company in its reasonable judgment to inspect such records as required by this Paragraph.
- 5.3 Commercially Reasonable Efforts. Company shall exercise commercially reasonable efforts to develop and to introduce the LICENSED PRODUCT(S) and LICENSED SERVICE(S) into the commercial market, through itself, its AFFILIATED COMPANIES and/or its SUBLICENSEE(S), consistent with sound and reasonable business practice and judgment. Beginning on the EFFECTIVE DATE Company shall spend, at a minimum, a total of \$[\*\*] USD over a [\*\*] year period on research and development activities related to potential LICENSED PRODUCTS. In addition, Company shall have [\*\*] of a LICENSED PRODUCT prior to the [\*\*] of the EFFECTIVE DATE. Following the introduction of a LICENSED PRODUCT or LICENSED SERVICE into the commercial market, and until the expiration or termination of this Agreement, Company shall endeavor to keep LICENSED PRODUCT(S) and LICENSED SERVICE(S) reasonably available to the public consistent with sound and reasonable business practice and judgment. Company shall also exercise commercially reasonable efforts to develop LICENSED PRODUCT(S) suitable for different indications within the LICENSED FIELD, consistent with sound and reasonable business practice and judgment.
- **5.4 Patent Acknowledgement.** Company agrees that all packaging containing individual LICENSED PRODUCT(S) sold by Company, AFFILIATED COMPANIES and SUBLICENSEE(S) of Company will be marked with the number of the applicable patent(s) licensed hereunder in accordance with each country's patent laws to the extent reasonably practical.

## ARTICLE 6 REPRESENTATIONS

- **6.1 Duties of the Parties.** JHU is not a commercial organization. It is an institute of research and education. Therefore, JHU has no ability to evaluate the commercial potential of any PATENT RIGHTS or LICENSED PRODUCT or other license or rights granted in this Agreement. It is therefore incumbent upon Company to evaluate the rights and products in question, to examine the materials and information provided by JHU, and to determine for itself the validity of any PATENT RIGHTS, its freedom to operate, and the value of any LICENSED PRODUCTS or LICENSED SERVICES or other rights granted.
- Representations by JHU. JHU warrants that it has good and marketable title to its interest in the inventions claimed under PATENT RIGHTS with the exception of certain retained rights of the United States Government, which may apply if any part of the JHU research was funded in whole or in part by the United States Government. JHU does not warrant the validity of any patents or that practice under such patents shall be free of infringement. EXCEPT AS EXPRESSLY SET FORTH IN THIS PARAGRAPH 6.2, COMPANY, AFFILIATED COMPANIES AND SUBLICENSEE(S) AGREE THAT THE PATENT RIGHTS ARE PROVIDED "AS IS", AND THAT JHU MAKES NO REPRESENTATION OR WARRANTY WITH RESPECT TO THE PERFORMANCE OF LICENSED PRODUCT(S) AND LICENSED SERVICE(S) INCLUDING THEIR SAFETY, EFFECTIVENESS, OR COMMERCIAL VIABILITY. JHU DISCLAIMS ALL WARRANTIES WITH REGARD TO PRODUCT(S) AND SERVICE(S) LICENSED UNDER THIS AGREEMENT, INCLUDING, BUT NOT LIMITED TO, ALL WARRANTIES, EXPRESSED OR IMPLIED, OF MERCHANTABILITY AND FITNESS FOR ANY PARTICULAR PURPOSE. NOTWITHSTANDING ANY OTHER PROVISION OF THIS AGREEMENT, JHU ADDITIONALLY DISCLAIMS ALL OBLIGATIONS AND LIABILITIES ON THE PART OF JHU AND INVENTORS, FOR DAMAGES, INCLUDING, BUT NOT LIMITED TO, DIRECT, INDIRECT, SPECIAL, AND CONSEQUENTIAL DAMAGES, ATTORNEYS' AND EXPERTS' FEES, AND COURT COSTS (EVEN IF JHU HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES, FEES OR COSTS), ARISING OUT OF OR IN CONNECTION WITH THE MANUFACTURE, USE, OR SALE OF THE PRODUCT(S) AND SERVICE(S) LICENSED UNDER THIS AGREEMENT. COMPANY, AFFILIATED COMPANIES AND SUBLICENSEE(S) ASSUME ALL RESPONSIBILITY AND LIABILITY FOR LOSS OR DAMAGE CAUSED BY A PRODUCT AND/OR SERVICE MANUFACTURED, USED, OR SOLD BY COMPANY, ITS SUBLICENSEE(S) AND AFFILIATED COMPANIES WHICH IS A LICENSED PRODUCT(S) OR LICENSED SERVICE(S) AS DEFINED IN THIS AGREEMENT.
- **6.3 Prior Assignment of Rights.** The Parties acknowledge the Assignment of Rights Agreement dated December 8, 2005 among JHU, [\*\*] (the "Assignment Agreement"). JHU agrees (i) that as between JHU on the one hand and [\*\*] on the other hand, [\*\*] are the absolute owners of the patent rights that are the subject of the Assignment Agreement, and JHU will not exercise any right it may have to terminate said Assignment Agreement, (ii) JHU is not entitled to receive any part of any stock in the Company issued to [\*\*], and (iii) the Company will not be obligated to make any payment (including without limitation, royalties and milestones) to JHU with respect to the patent assigned under the Assignment Agreement.

## ARTICLE 7 INDEMNIFICATION

**Indemnification.** JHU and the Inventors would have no legal liability exposure to third parties if JHU did not license the LICENSED PRODUCT(S) and LICENSED SERVICE(S), and any royalties JHU and the Inventors may receive is not adequate compensation for such legal liability exposure. Therefore, JHU requires Company to protect JHU and Inventors from such exposure to the same manner and extent to which insurance, if available, would protect JHU and Inventors. Furthermore, JHU and the Inventors will not, under the provisions of this Agreement or otherwise, have control over the manner in which Company or its AFFILIATED COMPANIES or its SUBLICENSEE(S) or those operating for its account or third parties who purchase LICENSED PRODUCT(S) or LICENSED SERVICE(S) from any of the foregoing entities, develop, manufacture, market or practice the inventions of LICENSED PRODUCT(S) and LICENSED SERVICE(S). Therefore, Company, AFFILIATED COMPANY and SUBLICENSEE, each solely with respect to its own practice of such inventions, shall indemnify, defend with counsel reasonably acceptable to JHU, and hold JHU, The Johns Hopkins Health Systems, their present and former trustees, officers, Inventors of PATENT RIGHTS, agents, faculty, employees and students harmless as against any judgments, fees, expenses, or other costs arising from or incidental to any product liability or other lawsuit, claim, demand or other action brought as a consequence of its own practice of said inventions, whether or not JHU or said Inventors, either jointly or severally, is named as a party defendant in any such lawsuit and whether or not JHU or the Inventors are alleged to be negligent or otherwise responsible for any injuries to persons or property. Practice of the inventions covered by LICENSED PRODUCT(S) and LICENSED SERVICE(S), by an AFFILIATED COMPANY, SUBLICENSEE, or an agent or a third party on behalf of or for the account of Company or by a third party who purchases LICENSED PRODUCT(S) and LICENSED SERVICE(S) from Company, shall be considered Company's practice of said inventions for purposes of this Paragraph. The obligation of Company to defend and indemnify as set out in this Paragraph shall survive the termination of this Agreement, shall continue even after assignment of rights and responsibilities to an affiliate or sublicensee, and shall not be limited by any other limitation of liability elsewhere in this Agreement.

# ARTICLE 8 CONFIDENTIALITY

**8.1 Confidentiality.** If necessary, the parties will exchange information, which they consider to be confidential. The recipient of such information agrees to accept the disclosure of said information which is marked as confidential at the time it is sent to the recipient, and to employ all reasonable efforts to maintain the information secret and confidential, such efforts to be no less than the degree of care employed by the recipient to preserve and safeguard its own confidential information, and in any event no less than a reasonable degree of care. The information shall not be disclosed or revealed to anyone except employees, consultants, collaborators, investors and prospective investors of the recipient who have a need to know the information and who have entered into a secrecy agreement with the recipient under which such employees are required to maintain confidential the proprietary information of the recipient and such employees shall be advised by the recipient of the confidential nature of the information and that the information shall be treated accordingly.

The obligations of this Paragraph 8.1 shall also apply to AFFILIATED COMPANIES and/or SUBLICENSEE(S) provided such information of JHU by Company. JHU's, Company's, AFFILIATED COMPANIES, and SUBLICENSEES' obligations under this Paragraph 8.1 shall extend until [\*\*] years after the termination of this Agreement.

- **8.2 Exceptions.** The recipient's obligations under Paragraph 8.1 shall not extend to any part of the information:
- **a.** that can be demonstrated to have been in the public domain or publicly known and readily available to the trade or the public prior to the date of the disclosure; or
- **b.** that can be demonstrated, from written records to have been in the recipient's possession or readily available to the recipient from another source not under obligation of secrecy to the disclosing party prior to the disclosure; or
- **c.** that becomes part of the public domain or publicly known by publication or otherwise, not due to any unauthorized act by the recipient; or
- **d.** that is demonstrated from written records to have been developed by or for the receiving party without reference to confidential information disclosed by the disclosing party.
- **e.** that is required to be disclosed by law, government regulation or court order.
- **8.3 Right to Publish.** JHU may publish manuscripts, abstracts or the like describing the PATENT RIGHTS and inventions contained therein provided confidential information of Company as defined in Paragraph 8.1, is not included or without first obtaining approval from Company to include such confidential information. Otherwise, JHU and the Inventors shall be free to publish manuscripts and abstracts or the like directed to the work done at JHU related to the licensed technology without prior approval.

### ARTICLE 9 TERM & TERMINATION

- **9.1 Term.** The term of this Agreement shall commence on the EFFECTIVE DATE and shall continue, in each country, until the date of expiration of the last to expire patent included within PATENT RIGHTS in that country or if no patents issue then for a term of twenty (20) years from the EFFECTIVE DATE of this Agreement.
- **9.2 Termination By Either Party.** This Agreement may be terminated by either party, in the event that the other party (a) files or has filed against it a petition under the Bankruptcy Act that is not dismissed within [\*\*] days, makes an assignment for the benefit of creditors, has a receiver appointed for it or a substantial part of its assets and such receivership is not terminated within [\*\*] days, or otherwise takes advantage of any statute or law designed for relief of debtors or (b) fails to perform or otherwise breaches any of its obligations hereunder, if, following the giving of notice by the terminating party of its intent to terminate and stating the grounds therefor, the party receiving such notice shall not have cured the failure or breach within [\*\*] days; provided, however, that, except where the alleged breach is for failure to pay a fixed amount due

under this agreement (such as the License Fee, Minimum Annual Royalty or the \$[\*\*] monthly patent cost reimbursement payment set forth in Section 3.6), in the event the party receiving the notice disputes the alleged failure to perform or breach in good faith, such [\*\*] day cure period shall commence upon determination by a court of competent jurisdiction (or arbitrator if the parties agree to arbitrate the matter) that the alleged failure to perform or breach exists. In no event, however, shall such notice or intention to terminate be deemed to waive any rights to damages or any other remedy which the party giving notice of breach may have as a consequence of such failure or breach.

- **9.3 Termination by Company.** Company may terminate this Agreement and the license granted herein, for any reason, upon giving JHU ninety (90) days written notice. Company may terminate its license with respect to any particular patent or patent application with 60 days' notice to JHU and Exhibit D shall be considered amended accordingly. Company will not be required to reimburse JHU for patent costs incurred after the 60-day notice period for such patents or patent applications.
- 9.4 Obligations and Duties upon Termination. If this Agreement is terminated, both parties shall be released from all obligations and duties imposed or assumed hereunder to the extent so terminated, except as expressly provided to the contrary in this Agreement. Upon termination, both parties shall cease any further use of the confidential information disclosed to the receiving party by the other party. Termination of this Agreement, for whatever reason, shall not affect the obligation of either party to make any payments for which it is liable prior to or upon such termination. Termination shall not affect JHU's right to recover unpaid royalties, fees, reimbursement for patent expenses, or other forms of financial compensation incurred prior to termination. Upon termination Company shall submit a final royalty report to JHU and any royalty payments, fees, unreimbursed patent expenses and other financial compensation due JHU shall become immediately payable. Furthermore, upon termination of this Agreement, all rights in and to the licensed technology shall revert immediately to JHU at no cost to JHU. Upon termination of this Agreement, any SUBLICENSEE(S) shall become a direct licensee of JHU, provided that JHU's obligations to SUBLICENSEE(S) are no greater than JHU's obligations to Company under this Agreement. Company shall provide written notice of such to each SUBLICENSEE(S) with a copy of such notice provided to JHU.

#### ARTICLE 10 MISCELLANEOUS

10.1 Use of Name. Except as required by applicable law or regulations, Company, AFFILIATED COMPANIES and SUBLICENSEE(S) shall not use the name of The Johns Hopkins University or The Johns Hopkins Health System or any of its constituent parts, such as the Johns Hopkins Hospital or any contraction thereof or the name of Inventors in any advertising, promotional, sales literature or fundraising documents without prior written consent from an authorized representative of JHU, or, in the case of the name of an Inventor, from such Inventor; provided that the Company may disclose the existence of this Agreement and the terms hereof and the fact that [\*\*] is a founder of the Company and a faculty member of JHU without such consent. Company, AFFILIATED COMPANIES and SUBLICENSEE(S) shall allow at least [\*\*] days notice of any proposed public disclosure for JHU's review and comment or to provide written consent.

- **10.2 No Partnership.** Nothing in this Agreement shall be construed to create any agency, employment, partnership, joint venture or similar relationship between the parties other than that of a licensor/licensee. Neither party shall have any right or authority whatsoever to incur any liability or obligation (express or implied) or otherwise act in any manner in the name or on the behalf of the other, or to make any promise, warranty or representation binding on the other.
- **10.3 Notice of Claim.** Each party shall give the other or its representative immediate notice of any suit or action filed, or prompt notice of any claim made, against them arising out of the performance of this Agreement or arising out of the practice of the inventions licensed hereunder.
- 10.4 Product Liability. Prior to initial human testing or first commercial sale of any LICENSED PRODUCT(S) or LICENSED SERVICE(S) as the case may be in any particular country, Company shall establish and maintain, covering the Company's liability in each country in which Company, an AFFILIATED COMPANY or SUBLICENSEE(S) shall test or sell LICENSED PRODUCT(S) and LICENSED SERVICE(S), product liability or other appropriate insurance coverage in the minimum amount of [\*\*] dollars (\$[\*\*]) per claim and will annually present evidence to JHU that such coverage is being maintained. Upon JHU's request, Company will furnish JHU with a Certificate of Insurance of each product liability insurance policy obtained. JHU shall be listed as an additional insured in Company's said insurance policies. If such Product Liability insurance is underwritten on a 'claims made' basis, Company agrees that any change in underwriters during the term of this Agreement will require the purchase of 'prior acts' coverage to ensure that coverage will be continuous throughout the term of this Agreement.
- **10.5 Governing Law.** This Agreement shall be construed, and legal relations between the parties hereto shall be determined, in accordance with the laws of the State of Maryland applicable to contracts solely executed and wholly to be performed within the State of Maryland without giving effect to the principles of conflicts of laws. Any disputes between the parties to this Agreement shall be brought in the state or federal courts of Maryland. Both parties agree to waive their right to a jury trial.
- **10.6 Notice.** All notices or communication required or permitted to be given by either party hereunder shall be deemed sufficiently given if mailed by registered mail or certified mail, return receipt requested, or sent by overnight courier providing evidence of delivery, such as Federal Express, to the other party at its respective address set forth below or to such other address as one party shall give notice of to the other from time to time hereunder. Mailed notices shall be deemed to be received on the third business day following the date of mailing. Notices sent by overnight courier shall be deemed received the following business day.

If to Company:

Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C. One Financial Center Boston, MA 02111 Attn: Jeffrey M. Wiesen, Esq. With a Copy to: Mintz, Levin, Cohn, Ferris,

Glovsky and Popeo, P.C. One Financial Center Boston, MA 02111

Attn: Jeffrey M. Wiesen, Esq.

If to JHU: Director

Technology Transfer Johns Hopkins University 100 N. Charles Street

5th Floor

Baltimore, MD 21201 Attn: Agrmt [\*\*]

- **10.7 Compliance with All Laws.** In all activities undertaken pursuant to this Agreement, both JHU and Company covenant and agree that each will in all material respects comply with such Federal, state and local laws and statutes, as may be in effect at the time of performance and all valid rules, regulations and orders thereof regulating such activities.
- **10.8** Successors and Assigns. Neither this Agreement nor any of the rights or obligations created herein, except for the right to receive any remuneration hereunder, may be assigned by either party, in whole or in part, without the prior written consent of the other party, except that either party shall be free to assign this Agreement in connection with its merger or consolidation or any sale of substantially all of its assets without the consent of the other. This Agreement shall bind and inure to the benefit of the successors and permitted assigns of the parties hereto.
- 10.9 No Waivers; Severability. No waiver of any breach of this Agreement shall constitute a waiver of any other breach of the same or other provision of this Agreement, and no waiver shall be effective unless made in writing. Any provision hereof prohibited by or unenforceable under any applicable law of any jurisdiction shall as to such jurisdiction be deemed ineffective and deleted herefrom without affecting any other provision of this Agreement. It is the desire of the parties hereto that this Agreement be enforced to the maximum extent permitted by law, and should any provision contained herein be held by any governmental agency or court of competent jurisdiction to be void, illegal and unenforceable, the parties shall negotiate in good faith for a substitute term or provision which carries out the original intent of the parties.
- 10.10 Entire Agreement; Amendment. Company and JHU acknowledge that they have read this entire Agreement and that this Agreement, including the attached Exhibits constitutes the entire understanding and contract between the parties hereto and supersedes any and all prior or contemporaneous oral or written communications with respect to the subject matter hereof, all of which communications are merged herein. It is expressly understood and agreed that (i) there being no expectations to the contrary between the parties hereto, no usage of trade, verbal agreement or another regular practice or method dealing within any industry or between the parties hereto shall be used to modify, interpret, supplement or alter in any manner the express terms of this Agreement; and (ii) this Agreement shall not be modified, amended or in any way altered except by an instrument in writing signed by both of the parties hereto.

- **10.11 Delays or Omissions.** Except as expressly provided herein, no delay or omission to exercise any right, power or remedy accruing to any party hereto, shall impair any such right, power or remedy to such party nor shall it be construed to be a waiver of any such breach or default, or an acquiescence therein, or in any similar breach or default be deemed a waiver of any other breach or default theretofore or thereafter occurring. Any waiver, permit, consent or approval of any kind or character on the part of any party of any breach or default under this Agreement, or any waiver on the part of any party of any provisions or conditions of this Agreement, must be in writing and shall be effective only to the extent specifically set forth in such writing. All remedies either under this Agreement or by law or otherwise afforded to any party, shall be cumulative and not alternative.
- **10.12 Force Majeure.** If either party fails to fulfill its obligations hereunder (other than an obligation for the payment of money), when such failure is due to an act of God, or other circumstances beyond its reasonable control, including but not limited to fire, flood, civil commotion, riot, war (declared and undeclared), revolution, or embargoes, then said failure shall be excused for the duration of such event and for such a time thereafter as is reasonable to enable the parties to resume performance under this Agreement, provided however, that in no event shall such time extend for a period of more than [\*\*] days.
- **10.13 Further Assurances.** Each party shall, at any time, and from time to time, prior to or after the EFFECTIVE DATE of this Agreement, at reasonable request of the other party, execute and deliver to the other such instruments and documents and shall take such actions as may be required to more effectively carry out the terms of this Agreement.
- **10.14 Survival.** All representations, warranties, covenants and agreements made herein and which by their express terms or by implication are to be performed after the execution and/or termination hereof, or are prospective in nature, shall survive such execution and/or termination, as the case may be. This shall include Paragraphs 3.7 (Late Payments), 5.2 (Records), and Articles 6, 7, 8, 9, and 10.
- **10.15 No Third Party Beneficiaries.** Nothing in this Agreement shall be construed as giving any person, firm, corporation or other entity, other than the parties hereto and their successors and permitted assigns, any right, remedy or claim under or in respect of this Agreement or any provision hereof.
- **10.16 Headings.** Article headings are for convenient reference and not a part of this Agreement. All Exhibits are incorporated herein by this reference.
- **10.17 Counterparts.** This Agreement may be executed in counterparts, each of which shall be deemed an original and all of which when taken together shall be deemed but one instrument.

IN WITNESS WHEREOF, this Agreement shall take effect as of the EFFECTIVE DATE when it has been executed below by the duly authorized representatives of the parties.

THE JOHNS HOPKINS UNIVERSITY	HANES NEWCO, INC.
/s/ Wesley D. Blakeslee	/s/ Robert Paull
Wesley D . Blakeslee, J.D.	Name: Robert Paull
Executive Director	Title: CEO
Johns Hopkins Technology Transfer	
10 Nov. 2009	November 9th, 2009
(Date)	(Date)
I have read and agree to abide by the terms of this Agreement:	
/s/ [**]	11/11/09
[**]	Date

EXHIBIT A. LICENSE FEE & ROYALITIES. EXHIBIT B. SALES & ROYALTY REPORT FORM.

EXHIBIT C. MATERIALS.

EXHIBIT D. PATENT APPLICATIONS.

#### **EXHIBIT A**

## LICENSE FEE & ROYALTIES

- **1. License Fee:** The license fee due under Paragraph 3.1 is
  - (i) [\*\*] dollars (\$[\*\*]), and
  - (ii) shares of Company stock in such amounts that after issuance of those shares JHU shall own [\*\*] percent ([\*\*]%) of the total pre-financing outstanding shares of Company. After that, the JHU equity interest shall be diluted at the same rate as the founders' shares through subsequent rounds of equity financing. JHU will become a party to Company's Investor Rights Agreement on the same basis as Company's founders.
- **2. Minimum Annual Royalties:** The minimum annual royalties pursuant to Paragraph 3.2 are:

[\*\*] year: [\*\*] dollars (\$[\*\*]).

[\*\*] year: [\*\*] dollars (\$[\*\*]).

[\*\*] year and each year thereafter until Launch

Fifty thousand dollars (\$50,000).

Year:

Launch Year and each year thereafter: One hundred thousand dollars (\$100,000).

"Launch Year" shall mean the year of the first commercial sale of a LICENSED PRODUCT in the U.S., EU or Japan.

**3. Royalties:** The running royalty rate payable under Paragraph 3.3 is:

Portion of Annual Sales	Royalty Rate
0-\$[**]	[**]%
\$[**]-\$[**]	[**]%
>\$[**]	[**]%

For those annual sales exceeding \$[\*\*]: In the event that Company is required to pay running royalties on any patent rights not licensed hereunder ("Other Payments") in order to commercialize a LICENSED PRODUCT or LICENSED SERVICE, Company may offset [\*\*]% of such Other Payments actually paid against royalty payments owed to JHU, provided that that the royalty payment to JHU shall not fall below [\*\*]% of that which would otherwise be due JHU for that LICENSED PRODUCT or LICENSED SERVICE.

**4. Sublicense consideration:** The percent of SUBLICENSE CONSIDERATION payable under Paragraph 3.4 is:

[\*\*]% for any sublicense executed prior to [\*\*] months after the EFFECTIVE DATE;

[\*\*]% for any sublicense executed between [\*\*] months after the EFFECTIVE DATE and [\*\*] months after the EFFECTIVE DATE; and

[\*\*]% for any sublicense executed more than [\*\*] months after the EFFECTIVE DATE.

In the event that (i) Company sublicenses PATENT RIGHTS together with patent rights owned by Company and necessary to commercialize a LICENSED PRODUCT, the above percentages will be reduced by [\*\*]% and (ii) Company pays a portion of sublicensing revenue to a third party under a third-party license for patent rights necessary to commercialize a LICENSED PRODUCT, Company may offset [\*\*]% of such amounts actually paid against percentage of SUBLICENSE CONSIDERATION payments owed to JHU; provided that the percentage of SUBLICENSE CONSIDERATION payments to JHU shall not fall below [\*\*]%.

- **5. Milestones:** The milestones payable under Paragraph 3.5 are:
  - (i) [\*\*] upon [\*\*]
  - (ii) \$[\*\*] upon [\*\*]
  - (iii) \$[\*\*] upon [\*\*]
  - (iv) \$[\*\*] upon [\*\*]
  - (v) \$[\*\*] upon [\*\*]

Each milestone payment shall be paid for the first three LICENSED PRODUCTS to achieve such milestone and no milestone payments shall be required for subsequent LICENSED PRODUCTS. Milestone payments for the second and third LICENSED PRODUCTS shall be reduced by [\*\*]%.

# FIRST AMENDMENT TO EXCLUSIVE LICENSE AGREEMENT

This First Amendment to Exclusive License Agreement is entered into by and between The Johns Hopkins University ("JHU") and Kala Pharmaceuticals, Inc., formerly known as Hanes NewCo, Inc, a Delaware corporation ("Kala" or "Company").

**WHEREAS**, Hanes NewCo, Inc. and JHU entered into an Exclusive License Agreement dated November 10, 2009 ("Agreement") pursuant to which JHU licensed certain patent rights and know-how to Hanes NewCo, Inc.

WHEREAS, Hanes NewCo, Inc. changed its name to Kala Pharmaceuticals, Inc. on December 11, 2009.

**WHEREAS**, the parties wish to amend the Agreement on the terms set forth herein;

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth below, the parties agree as follows:

- Capitalized terms used herein and not defined herein shall have the respective meanings ascribed to such terms in the Agreement.
- 2. Each of the parties hereby agrees to amend the agreement to substitute Kala Pharmaceuticals, Inc. in place of Hanes NewCo, Inc. every place such name appears in the Agreement.
- 3. As of the First Amendment Date, Section 2.4 of the Agreement shall be superseded in its entirety by the following:

"Commencing on the First Amendment Date, Company may, within [\*\*] days following the end of a calendar quarter, by email to Director, Technology Licensing [email: [\*\*] or [\*\*]]. request information about any Improvements that were disclosed to the JHU technology transfer office during the previous calendar quarter. Within [\*\*] days after receipt of such email, JHU shall notify Company in writing of any such Improvements; provided, however, that if an invention disclosure describing an Improvement also contains disclosure for one or more additional inventions which are not Improvements ("Other Inventions"), then JHU may redact information on such Other Inventions. Subject to any third party, including U.S. Government, obligations of JHU under any agreement under which any Improvement is made, Company will have a right of first negotiation to amend this Agreement to add the Improvement and related patent rights, know how and materials to this Agreement; provided, that Company notifies JHU of Company's interest in exercising such right of first negotiation within [\*\*] days after receipt of written notice from JHU of such Improvement. JHU and Company will negotiate in good faith on reasonable terms for adding the Improvement to this Agreement. For purposes hereof, "Improvement" shall mean an invention: (i) made in the laboratory of [\*\*]; (ii) that would infringe the PATENT RIGHTS if made, used, imported or sold without a license to PATENT RIGHTS; (iii) pertaining to mucosal delivery using mucus penetrating nanoparticles or microparticles; and (iv) reported to the JHU technology transfer office within [\*\*] years from the First Amendment Date."

4. In Section 10.6, change the addresses for notices to the Company to the following:

"If to Company: Kala Pharmaceuticals, Inc.

135 Beaver Street #309 Waltham, MA 02453 Attn: Chief Executive Officer With a Copy to: General Counsel"

- 5. In all other respects the Agreement shall remain in full force and effect.
- 6. This First Amendment shall be effective as of the date the last party hereto has executed this First Amendment (the "First Amendment Date").
- 7. For the convenience of the parties hereto, this Amendment may be executed in two counterparts, each of which shall be deemed to be an original, but both of which together shall constitute one and the same instrument, without necessity of production of the others. Signatures may be exchanged by facsimile or electronic transmission and each of the parties to this Amendment agrees that it will be bound by its own facsimile signature and that it accepts the facsimile signature of the other party.

8. Amendment Fee. Company shall pay to JHU within [\*\*] days of the First Amendment Date [\*\*] dollars (\$[\*\*]) as an amendment fee. JHU will not submit an invoice for the amendment fee, which is nonrefundable and shall not be credited against royalties or other fees.

This Agreement may be executed in one or more counterparts each of which shall be deemed an original but all of which together shall constitute one and the same instrument.

# KALA PHARMAEUTICALS, INC.

# THE JOHNS HOPKINS UNIVERSITY

By: /s/ Guillaume Pfefer duly authorized By: /s/ Wesley D. Blakeslee duly authorized

Guillaume Pfefer Wesley D. Blakeslee, J.D.

CEO Executive Director

Date: Johns Hopkins Technology Transfer

Date: 11/19/2012

# SECOND AMENDMENT TO EXCLUSIVE LICENSE AGREEMENT

This Second Amendment to Exclusive License Agreement ("Second Amendment") is entered into by and between The Johns Hopkins University, a Maryland corporation having an address at 3400 N. Charles Street, Baltimore, MD 21218-2695 ("JHU"), and Kala Pharmaceuticals, Inc., formerly known as Hanes NewCo, Inc, a Delaware corporation having an address at 100 Beaver Street, Suite 201, Waltham, MA 02453 ("Kala" or "Company").

**WHEREAS,** Kala and JHU entered into an Exclusive License Agreement dated November 10, 2009, as amended by a First Amendment (the "First Amendment") dated November 19, 2012 (as so amended, the "Agreement"), pursuant to which JHU licensed certain patent rights and know-how to Kala;

**WHEREAS**, the Exclusive License Agreement provides for milestone payments to JHU upon the achievement by Company of certain clinical trial milestones with respect to Licensed Products;

**WHEREAS,** JHU, Company, and GrayBug LLC are discussing, as of the Second Amendment Date (as defined below), the proposed terms of a draft Settlement and License Agreement between JHU. Company, and GrayBug LLC (as may be modified by JHU, Company, and GrayBug, the "Proposed Settlement Terms"). (GrayBug LLC is hereinafter referred to as "GrayBug").

WHEREAS, Company and JHU wish to amend the Agreement on the terms set forth herein;

**NOW, THEREFORE,** in consideration of the mutual covenants and agreements set forth below, the Company and JHU agree as follows:

- 1. As of the Second Amendment Dale, Section 3.5 of the Agreement shall be superseded in its entirety by the following:
  - "Company shall pay to JHU milestones as set forth in Exhibit A. These milestones shall be due, without invoice from JHU, within [\*\*] days of achievement of such milestone; provided however, if any of the milestones set forth in Exhibit A are achieved prior to [\*\*], the milestone payment for such milestone shall not be due until the earlier of (i) [\*\*] days after execution by JHU, Company, and GrayBug of the Proposed Settlement Terms and (ii) [\*\*]. Notwithstanding anything to the contrary in this Agreement or its exhibits, upon execution of the Proposed Settlement Terms by JHU, Company, and GrayBug, each milestone achieved prior to the execution of the Proposed Settlement Terms shall be subject to any payment reductions set forth in the executed Proposed Settlement Terms as if such milestones were achieved by Company after the execution of the Proposed Settlement Terms."
- 2. This Second Amendment shall be effective as of May 22, 2014 (the "Second Amendment Date").
- 3. In all other respects the Agreement shall remain in full force and effect.
- 4. Neither Company nor JHU is required to pay an amendment fee in connection with the execution of this Second Amendment.

This Agreement may be executed in one or more counterparts each of which shall be deemed an original but all of which together shall constitute one and the same instrument.

## KALA PHARMACEUTICALS, INC.

## THE JOHNS HOPKINS UNIVERSITY

By: /s/ Guillaume Pfefer	By: /s/ Steven L. Kousouris
duly authorized	duly authorized
[Guillaume Pfefer]	[Wesley D. Blakeslee, J.D.] Steven L. Kousouris
President &CEO	[Executive Director] Sr. Director
Date: 5/23/2014	Johns Hopkins Technology Transfer
	Date: 5/23/2014

# THIRD AMENDMENT TO EXCLUSIVE LICENSE AGREEMENT

This Third Amendment to Exclusive License Agreement ("Third Amendment") is entered into by and between The Johns Hopkins University, a Maryland corporation having an address at 3400 N. Charles Street, Baltimore, MD 21218-2695 ("JHU"), and Kala Pharmaceuticals, Inc., formerly known as Hanes NewCo, Inc., a Delaware corporation having an address at 100 Beaver Street, Suite 201, Waltham, MA 02453 ("Kala" or "Company").

WHEREAS, Kala and JHU entered into an Exclusive License Agreement dated November 10, 2009, as amended by a First Amendment (the "First Amendment") dated November 19, 2012 and a Second Amendment (the "Second Amendment") effective as of May 22, 2014 (as so amended, the "Agreement"), pursuant to which JHU licensed certain patent rights and know-how to Kala and modified certain milestones, respectively;

**WHEREAS,** Paragraph 2.5 and certain other provisions of the Exclusive License Agreement reference "OPTION PATENT RIGHTS" and Company declined in 2010 to license such rights;

**WHEREAS**, the parties are joint owners of the patent rights relating to the invention entitled [\*\*] and the parties wish to amend the Agreement to add such jointly owned patent rights, and to amend the terms of the Agreement as set forth herein regarding each party's rights with respect to the patent rights for such invention;

**WHEREAS,** JHU, Company, and GrayBug LLC ("GrayBug") are discussing proposed terms for a Settlement and License Agreement between JHU, Company and GrayBug (as may be modified by JHU, Company and GrayBug, the "Settlement Agreement");

**WHEREAS,** the parties wish to amend the Agreement on the terms set forth herein; **NOW, THEREFORE,** in consideration of the mutual covenants and agreements set forth below, the parties agree as follows:

- 1. Capitalized terms used herein and not defined herein shall have the respective meanings ascribed to such terms in the Agreement.
- 2. The parties agree that notwithstanding that no date is associated with Kala's signature to the First Amendment, the parties agree that the "First Amendment Date" as defined in the First Amendment is November 19, 2012.
- 3. As of the Third Amendment Date, the second "Whereas" recital of the Agreement shall be superseded in its entirety by the following:

"WHEREAS, a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*], and a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*], and a valuable invention entitled [\*\*] was developed during the course of research conducted by [\*\*], and a valuable invention entitled [\*\*] was developed during the course of research conducted for

Company by [\*\*], and conducted for JHU by [\*\*] (all of the individuals in this paragraph who are officially affiliated with JHU, "Inventors"); and"

4. As of the Third Amendment Date, the third "Whereas" recital of the Agreement shall be superseded in its entirety by the following

"WHEREAS, JHU has acquired through assignment all rights, title and interest, with the exception of Kala's ownership interest in connection with JHU Ref. #[\*\*] and certain retained rights by the United States Government, in its interest in said valuable inventions; and"

- 5. As of the Third Amendment Date, a new Section 1.14 shall be added to the Agreement as follows:
  - "KALA/JHU JOINTLY OWNED PATENT RIGHTS' shall mean the issued patents and patent applications listed in EXHIBIT D for JHU Ref # [\*\*] (all such patent applications for JHU Ref # [\*\*] hereinafter referred to as "JOINT PATENT APPLICATIONS") and all continuations, divisions, claims of continuations-in-part applications directed to subject matter specifically described in the Joint Patent Applications, continued prosecution applications and reissues, reexaminations, extensions and supplemental protection certificates thereof, and any corresponding foreign patent applications, and any patents, or other equivalent foreign patent rights issuing, granted or registered thereon."
- 6. As of the Third Amendment Date, the definitions "LICENSED PRODUCTS" in Section 1.6 of the Agreement and "LICENSED SERVICES" in Section 1.7 of the Agreement are amended such that the phrase "the license granted to Company pursuant to this Agreement" in each definition shall include the addition of "or the Company's ownership interest" as follows:

"but for the license granted to Company pursuant to this Agreement or the Company's ownership interest,"

7. As of the Third Amendment Date, Section 1.10 of the Agreement shall be superseded in its entirety by the following:

""PATENT RIGHTS" shall mean the issued patents and patent applications listed in EXHIBIT D for JHU Ref # [\*\*] (including, without limitation, JHU Ref. # [\*\*]), JHU Ref # [\*\*], JHU Ref # [\*\*], JHU Ref # [\*\*], and KALA/JHU JOINTLY OWNED PATENT RIGHTS (all such patent applications for JHU Ref # [\*\*] (including, without limitation, JHU Ref. # [\*\*]), JHU Ref # [\*\*], JHU Ref # [\*\*], JHU Ref # [\*\*] and JOINT PATENT APPLICATIONS hereinafter referred to as "Patent Applications") and all continuations, divisions, claims of continuations-in-part applications directed to subject matter specifically described in the Patent Applications, continued prosecution applications and reissues, reexaminations, extensions and supplemental protection certificates thereof, and any corresponding foreign patent applications, and any patents, or other equivalent foreign patent rights issuing, granted or registered thereon."

- 8. As of the Third Amendment Date, the following sentence shall be added to the end of Section 2.2 of the Agreement:
  - "The sublicensing restrictions set forth in this Section 2.2 shall not apply to Company's right to sublicense Company's interest in KALA/JHU JOINTLY OWNED PATENT RIGHTS."
- 9. Section 4.1 of the Agreement shall not apply to KALA/JHU JOINTLY OWNED PATENT RIGHTS. Notwithstanding anything to the contrary in Section 4.1 of the Agreement, Company and JHU jointly own the KALA/JHU JOINTLY OWNED PATENT RIGHTS.
- 10. As of the Third Amendment Date, the following sentence shall be added to the end of Section 4.4 of the Agreement:
  - "Notwithstanding anything to the contrary in this Section 4.4, if a declaratory judgment action is brought naming Company as a defendant and alleging invalidity of any of the KALA/JHU JOINTLY OWNED PATENT RIGHTS, Company shall have the first right to defend such action; provided that Company elects to take over the sole defense of the action at its own expense. JHU shall cooperate fully with Company in connection with any such action."
- 11. As of the Third Amendment Date, the first sentence of Section 4.5 of the Agreement shall be amended to read:
  - "Any recovery by Company under Paragraph 4.3 shall be deemed to reflect loss of commercial sales, and Company shall pay to JHU [\*\*] percent ([\*\*]%) of the recovery, net of all reasonable costs and expenses associated with each suit or settlement; provided, however, that, with respect to any recovery by Company under Paragraph 4.3 for infringement of KALA/JHU JOINTLY OWNED PATENT RIGHTS, but no PATENT RIGHTS other than KALA/JHU JOINTLY OWNED PATENT RIGHTS, Company shall pay to JHU [\*\*] percent ([\*\*]%) of the recovery, net of all reasonable costs and expenses associated with each suit or settlement."
- 12. As of the Third Amendment Date, a new Section 4.7 shall be added to the Agreement as follows:
  - "Prosecution and Maintenance of Kala/JHU Jointly Owned Patent Rights. Notwithstanding anything to the contrary in this Agreement, except as provided below, Company, at Company's expense, shall file, prosecute and maintain all patents and patent applications specified under the KALA/JHU JOINTLY OWNED PATENT RIGHTS. Title to all such patents and patent applications shall reside in Company and JHU. Company shall have control over all patent matters in connection therewith under the KALA/JHU JOINTLY OWNED PATENT RIGHTS, provided however, that COMPANY shall (a) cause its patent counsel to timely copy JHU on all official actions and written correspondence with any patent office, and (b) allow JHU an opportunity to comment and advise Company.

Company shall consider and reasonably incorporate all comments and advice unless detrimental to Company's intellectual property rights. Upon signed written notification by Company to JHU that it will not file, prosecute, and/or maintain such patent applications or patent in a particular country, JHU shall have the first right to file, prosecute and/or maintain such patent applications or patent in such country at JHU's expense and JHU's interest in such patent applications or patent shall no longer be licensed to Company."

- 13. As of the Third Amendment Date, Section 9.4 of the Agreement is hereby amended, such that to the extent that any PATENT RIGHTS are jointly owned between the parties, such PATENT RIGHTS to the extent owned by Kala, including Kala's ownership interest in KALA/JHU JOINTLY OWNED PATENT RIGHTS, shall remain with Kala upon termination of the Agreement, and shall not transfer to JHU, and any licensee(s) under Kala's ownership interest of such PATENT RIGHTS shall remain a licensee(s) of Kala; provided that to the extent that a SUBLICENSEE was granted a license by Kala under JHU's ownership interest in such jointly owned PATENT RIGHTS, such SUBLICENSEE shall automatically become a direct licensee of JHU solely with respect to JHU's ownership interest in such jointly owned PATENT RIGHTS upon such termination, provided that JHU's obligations to such SUBLICENSEE(S) are no greater than JHU's obligations to Company under this Agreement.
- 14. As of the Third Amendment Date, the following sentence shall be added to the end of Section 10.1 of the Agreement:

"Notwithstanding anything to the contrary in this Section 10.1, Kala shall have the right to disclose the name of The Johns Hopkins University or The Johns Hopkins Health System or any of its constituent parts or any Inventors to exercise Kala's rights under Section 4.7 to file, prosecute and/or maintain the KALA/JHU JOINTLY OWNED PATENT RIGHTS."

15. As of the Third Amendment Date, the address for notices to Company in Section 10.6 shall be changed to the following:

"If to Company: Kala Pharmaceuticals, Inc.

100 Beaver Street, Suite 201 Waltham, MA 02453 Attn: Chief Executive Officer With a Copy to: General Counsel"

16. As of the Third Amendment Dale, Section 2 in Exhibit A of the Agreement shall be amended as follows:

"Launch Year and each year thereafter: One hundred thousand dollars (\$100,000)," shall be omitted and replaced with "Launch Year and each year thereafter: One hundred fifty thousand dollars (\$150,000)."

17. As of the Third Amendment Date, the following paragraph shall be added to the end of Section 3 of Exhibit A of the Agreement:

"Notwithstanding anything to the contrary in this Agreement or its exhibits, if the only VALID CLAIMS of PATENT RIGHTS that cover the LICENSED PRODUCT or LICENSED SERVICE are VALID CLAIMS of KALA/JHU JOINTLY OWNED PATENT RIGHTS, then only [\*\*]% of the NET SALES or the NET SERVICE REVENUES of such LICENSED PRODUCT or LICENSED SERVICE shall apply for purposes of the royalty calculation as set forth in this Section 3.3 of the Agreement and Section 3 of this Exhibit A."

18. As of the Third Amendment Date, the following paragraph shall be added to the end of Section 4 of Exhibit A of the Agreement:

"Notwithstanding anything to the contrary in this Agreement or its exhibits, the SUBLICENSE CONSIDERATION for any sublicense of only KALA/JHU JOINTLY OWNED PATENT RIGHTS, shall be reduced by [\*\*]% as applied to any obligations of Company under this Agreement. For clarity, the preceding paragraph regarding offsets for a third-party license shall apply after the reduction of SUBLICENSE CONSIDERATION set forth in this paragraph."

19. As of the Third Amendment Date, the following paragraph shall be added to the end of Section 5 of Exhibit A of the Agreement:

"Notwithstanding anything to the contrary in this Agreement or its exhibits, each milestone payment in this Section 5 shall be reduced by [\*\*]% if the only VALID CLAIMS of PATENT RIGHTS that cover the LICENSED PRODUCT are VALID CLAIMS of KALA/JHU JOINTLY OWNED PATENT RIGHTS; such [\*\*]% reduction is in addition to any reductions that already apply to the second and third LICENSED PRODUCTS that achieve the applicable milestone"

- 20. As of the Third Amendment Date, Exhibit D of the Agreement shall be superseded in its entirety by the updated Exhibit D ("Amended Exhibit D") attached hereto.
- 21. As of the Third Amendment Date, (a) Paragraph 2.5 shall be deleted in its entirety, (b) the clause "provided, however, that if the OPTION PATENT RIGHTS become PATENT RIGHTS upon exercise of the option set forth in Section 2.5, the OPTION PATENT RIGHTS shall not be included in PATENT RIGHTS for purposes of this Section 2.4" in the last sentence of Section 2.4 of the Agreement shall be deleted in its entirety, (c) the second sentence of Section 3.6 of the Agreement shall be deleted in its entirety, and (d) the second paragraph of Section 3.6 of the Agreement shall be amended to read as follows:

"In accordance with Paragraph 4.1 below, Company will reimburse JHU, within [\*\*] days of the receipt of an invoice from JHU for all costs associated with the preparation, filing, maintenance, and prosecution of PATENT RIGHTS incurred by JHU subsequent to the EFFECTIVE DATE of this Agreement."

- 22. In all other respects the Agreement shall remain in full force and effect.
- 23. This Third Amendment shall be effective as of the date the last party hereto has executed this Third Amendment (the "Third Amendment Date").

- 24. For the convenience of the parties hereto, this Amendment may be executed in two counterparts, each of which shall be deemed to be an original, but both of which together shall constitute one and the same instrument, without necessity of production of the others. Signatures may be exchanged by facsimile or electronic transmission and each of the parties to this Amendment agrees that it will be bound by its own facsimile signature and that it accepts the facsimile signature of the other party.
- 25. <u>Amendment Fee</u>. Company shall pay to JHU within [\*\*] days of the Third Amendment Date [\*\*] dollars (\$[\*\*]) as an amendment fee (the "Third Amendment Fee"). JHU will not submit an invoice for the Third Amendment Fee, which shall not be credited against royalties or other fees. The Third Amendment Fee shall be refunded to Company in full if this Third Amendment is terminated according to Section 26.
- 26. <u>Termination due to no Settlement</u>. In the event the Settlement Agreement is not executed by JHU, Company, GrayBug, and any other required parties, on or before January I, 2015, each section of this Third Amendment, except for the last sentence of Section 25, shall terminate and the applicable sections in the Agreement shall revert to the language as of immediately prior to the Third Amendment Date; provided that (A) Exhibit D of the Agreement shall be superseded in its entirety by the Amended Exhibit D (but excluding JHU Ref [\*\*]) attached hereto, and (B) each party shall retain its rights as a co-owner in any patent rights jointly owned by Kala and JHU relating to JHU Ref [\*\*].

This Third Amendment may be executed in one or more counterparts each of which shall be deemed an original but all of which together shall constitute one and the same instrument.

# KALA PHARMAEUTICALS, INC.

## THE JOHNS HOPKINS UNIVERSITY

Charles McDermott Director Steve Kousouris

Sr. Director, Administration
Chief Business Officer and Interim President

Johns Hopkins Technology Transfer

Date: Johns Hopkins Technology Transfer

August 26, 2014 Date: 8/26/2014

Certain identified information has been excluded from the exhibit because it is both (i) not material and (ii) is the type of information that the registrant treats as private or confidential. Double asterisks denote omissions.

SUBJECT TO FED. R. EVID. 408 CONFIDENTIAL

## **Settlement and License Agreement**

This Settlement and License Agreement (including its attachments, this "Agreement") is entered into as of October 24, 2014 ("Settlement Agreement Effective Date"), by and between GrayBug, LLC, a limited liability company organized and existing under the laws of the State of Maryland and having an address of P.O. Box 13043, Baltimore, MD 21203, on behalf of itself and its Affiliates (as defined below) (collectively, "GrayBug"), and Kala Pharmaceuticals, Inc., a corporation organized and existing under the laws of State of Delaware and having a place of business at 100 Beaver Street, Suite 201, Waltham, MA 02453, on behalf of itself and its Affiliates (collectively, "Kala").

WHEREAS, there is a dispute over the rights licensed to each Party under certain Patent Rights (as defined below) owned by JHU (as defined below);

WHEREAS, the Parties and JHU now wish to settle such dispute in an amicable manner which is intended to settle and clarify rights granted by JHU to each Party and to permit each Party to have the freedom to effectively develop pharmaceutical products that will benefit patients;

NOW, THEREFORE, in consideration of the mutual covenants contained herein, and other good and valuable consideration, the receipt of which is hereby acknowledged, upon approval of these terms by JHU (including acknowledgement by [\*\*]), GrayBug and Kala agree as follows:

## 1. Construction; Definitions.

1.A <u>Construction</u>. Each Party acknowledges that it has been advised by counsel during the course of negotiation of this Agreement, and, therefore, that this Agreement shall be interpreted without regard to any presumption or rule requiring construction against the Party causing this Agreement to be drafted. Any reference in this Agreement to a Section, subsection, paragraph or clause shall be deemed to be a reference to a Section, subsection, paragraph or clause of this Agreement. Except where the context otherwise requires, (a) wherever used, the use of any gender will be applicable to all genders; (b) the word "or" is used in the inclusive sense (and/or); (c) any definition of or reference to any agreement, instrument or other document refers to such agreement, instrument other document as from time to time amended, supplemented or otherwise modified (subject to any restriction on such amendments, supplements or modifications set forth herein or therein); (d) any reference to any law refers to such law as from time to time enacted, repealed or amended and includes any regulations under such law; (e) the words "herein", "hereof" and hereunder", and words of similar import, refer to this Agreement in its entirety and not to any particular provision hereof; and (f) the words "include", "includes" and "including" shall be deemed to be followed by the phrase "but not limited to", "without limitation" or words of similar import.

1.B <u>Definitions</u>. Unless specifically set forth to the contrary herein, the following terms, whether used in the singular or plural, shall have the respective meanings set forth below:

"Affiliate" means, with respect to a first Person, any second Person that directly or indirectly, through one or more intermediaries, is Controlled by, Controls, or is under Common Control with, such first Person, but only for so long as such Control shall continue. For purposes of this definition, "Controlling" (including, with correlative meanings, "Controlled by" and "under Common Control with") means, with respect to a Person, possession, direct or indirect, of (a) at least fifty percent (50%) of the voting securities (whether directly or pursuant to any option, warrant, or other similar arrangement) or other comparable equity interests of such Person, or (b) the power to elect or appoint at least fifty percent (50%) of the Board of Directors of such Person or otherwise to direct or cause direction of the management and policies of such Person (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise).

"<u>Business Day</u>" means a day that is not a Saturday, a Sunday, nor a day on which banking institutions in Boston, Massachusetts and Baltimore, Maryland are authorized by law to remain closed.

"Confidential Information" means any proprietary information which is disclosed by one Party or JHU to the other Party in connection with this Agreement, and including any "Confidential Information" as defined in the Prior CDA. In addition, the terms of this Agreement and the Prior CDA shall be considered Confidential Information of each Party. Notwithstanding the foregoing, Confidential Information excludes information that, in each case as demonstrated by competent written documentation:

- i. is in the public domain or publicly known prior to the disclosure to the receiving Party;
- ii. is lawfully in the receiving Party's possession prior to the disclosure;
- iii. becomes part of the public domain or publicly known by publication or otherwise through no unauthorized act or omission on the part of the receiving Party;
- iv. is provided to a receiving Party by a Third Party who has a lawful right to disclose it; or
- is independently discovered by a receiving Party without use of the disclosing Party's Confidential Information.

"Counterpart" means, with respect to a patent application, (a) any patent applications (including provisional applications) from which such patent application claims priority, (b) all continuations, continuations-in-part, continued prosecution applications and divisionals of such patent application or any application described in clause (a), (c) any issued and unexpired letters patents issuing on such patent application or any application described in clauses (a) or (b), (d) any patent extensions, pediatric extensions, supplementary protection certificates, registrations, confirmations, reissues,

re-examinations, validations, revalidations, and renewals of any patent described in clause (c), and (e) with respect to clauses (a) through (d), any foreign counterparts thereof.

"Current Patent Rights" means (a) with respect to GrayBug as the licensing party or Kala as the sublicensed party, the GrayBug Current Patent Rights and (b) with respect to Kala as the licensing party or GrayBug as the sublicensed party, the Kala Current Patent Rights.

"Exclusive Field" means (a) with respect to GrayBug, the GrayBug Field of Use, and (b) with respect to Kala, the Kala Field of Use.

"Existing JHU License Agreement" means (a) with respect to GrayBug, the GrayBug/JHU Agreement, and (b) with respect to Kala, the Kala/JHU Agreement.

"<u>Future Co-Licensed Patent Rights</u>" means any Future Patent Right that is licensed to GrayBug and Kala pursuant to Section 2(c)(iii), until such time (if any) as such license has expired or been terminated with respect to either Party.

"Future Patent Rights" means (a) any Patent Right (other than GrayBug Current Patent Rights, Kala Current Patent Rights, Kala/JHU Jointly Owned Patent Rights, Other JHU Patent Rights, Transferred Patent Rights, and Jointly Owned Transferred Patent Rights) that, (i) is in whole owned by JHU, (ii) names solely [\*\*] as inventor, or, as coinventors, limited to the following: [\*\*] and, subject to Section 2(c)(ix): [\*\*] (each of the foregoing, a "Co-Inventor"), or any other person (including laboratory personnel, fellows, and students) working, as of the relevant time, in the laboratory of [\*\*] who has authorized JHU in writing to license the relevant invention under this Settlement Agreement, (iii) is not subject to pre-existing Third Party rights granted by JHU or obligations of JHU to Third Parties (provided, however, that, to the extent such Third Party rights and obligations terminate during the Second Notice period of Section 2(c)(i), the relevant Patent Rights shall be considered Future Patent Rights), and (iv) relates to the GrayBug Field of Use or Kala Field of Use, and (b) any Counterparts of the foregoing.

"GrayBug Current Patent Rights" means [\*\*]; and any Counterparts of the foregoing.

"GrayBug Field of Use" means the use of a Particle for delivery of a biologically active material to the eye via injection, excluding such use of any Particle comprising or consisting of loteprednol etabonate. For the avoidance of doubt, GrayBug Field of Use includes without limitation injections to the eye that are intravitreal, intrastromal, intracameral, sub-tenon (including posterior and anterior), sub-retinal, retro-bulbar, peri-bulbar, suprachorodial, conjunctival, subconjunctival, episcleral, posterior juxtascleral, scleral, circumcorneal, and tear duct injections, in each case, wherein a Particle does not go through mucus, mucin, or a mucosal barrier of the eye, but the injection means can pass through mucus, mucin, or a mucosal barrier of the eye to inject the Particle to a point of delivery that is not mucus, mucin, or a mucosal barrier of the eye.

"GrayBug/JHU Agreement" means the Exclusive License Agreement between GrayBug and JHU dated [\*\*], as amended on or before the Settlement Agreement Effective Date and as may be amended after the Settlement Agreement Effective Date in accordance with this Agreement.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"JHU" means The Johns Hopkins University.

"JHU Patent Rights" means the GrayBug Current Patent Rights, the Kala Current Patent Rights, the Co-Licensed Other JHU Patent Rights, and the Future Co-Licensed Patent Rights.

"<u>Jointly Owned Transferred Patent Rights</u>" means JHU's ownership interest in the Kala/JHU Jointly Owned Patent Rights transferred to GrayBug pursuant to Section 2(d)(iv).

"<u>Kala/JHU Agreement</u>" means the Exclusive License Agreement between Kala and JHU dated November 10, 2009, as amended on or before the Settlement Agreement Effective Date and as may be amended after the Settlement Agreement Effective Date in accordance with this Agreement.

"Kala/JHU Jointly Owned Patent Rights" means [\*\*], and any Counterparts thereof.

"Kala Current Patent Rights" means [\*\*]; and any Counterparts of the foregoing.

"Kala Field of Use" means the use of a Particle for delivery of a biologically active material through mucus, mucin, or a mucosal barrier; provided, however, that such delivery does not involve administration via injection to the eye. For the avoidance of doubt, Kala Field of Use includes without limitation topical administration to the eye.

"<u>Licensed Product</u>" means any process or method, material, compositions, drug, or other product, the manufacture, use or sale of which would constitute, but for the license granted to the relevant Party pursuant to this Agreement, an infringement of a Valid Claim of the relevant Patent Rights (where infringement includes direct, contributory, or inducement to infringe) in the country of sale.

"<u>Licensed Service</u>" means the performance on behalf of a Third Party of any method or the manufacture of any product or the use of any product or composition which would constitute, but for the license granted to the relevant Party pursuant to this Agreement, an infringement of a Valid Claim of the relevant Patent Rights (where infringement includes direct, contributory, or inducement to infringe) in the country of performance of the services.

"Losses" means claims, actions, causes of action, suits, defenses, judgments, debts, offsets, accounts, covenants, contracts, agreements, torts, damages, and any and all demands and liabilities whatsoever, including costs, expenses, and attorneys' fees.

"Other JHU Patent Rights" means the Patent Rights listed in Schedule 1.24 and any Counterparts thereof, which shall be designated as (a) "Co-Licensed Other JHU Patent Rights" while both Parties retain licenses thereunder, pursuant to Section 2(b), (b) "GrayBug-Licensed Other JHU Patent Rights" while only GrayBug retains a license thereunder, pursuant to Section 2(b), and (c) "Kala-Licensed Other JHU Patent Rights" while only Kala retains a license thereunder, pursuant to Section 2(b).

"<u>Particle</u>" means a particle for delivery of a biologically active material, which particle has a [\*\*] or a [\*\*] (which may include a [\*\*] particle).

"Party" means each of GrayBug and Kala individually, and "Parties" means GrayBug and Kala collectively.

"Patent Rights" means patents, patent applications or provisional patent applications, utility models and utility model applications, petty patents, innovation patents, patents of addition, divisionals, continuations, continuation-in-part applications, continued prosecution applications, requests for continued examinations, reissues, renewals, registrations, confirmation, validations, revalidations, reexaminations, and extensions, pediatric extensions and supplementary protection certificates granted in relation thereto, in any country of the world.

"<u>Person</u>" means any natural person, corporation, general partnership, limited partnership, joint venture, proprietorship, or other business organization or a governmental agency or a political subdivision thereto.

"Potential Future Patent Right" means an invention (other than an invention covered by GrayBug Current Patent Rights, Kala Current Patent Rights, Kala Current Patent Rights, Kala/JHU Jointly-Owned Patent Rights or Future Patent Rights, as such Patent Rights exist as of the Settlement Agreement Effective Date), that (a) [\*\*] is an inventor of and (b) relates to the GrayBug Field of Use or the Kala Field of Use.

"Prior CDA" means the Confidentiality Agreement between Kala Pharmaceuticals, Inc. and GrayBug, LLC, made and entered into as of June 28, 2013.

"Prosecution and Maintenance" means, with regard to a Patent Right, the preparation, filing, and prosecution of the applicable patent application(s), the maintenance of applicable patent application(s) and patent(s), and the preparation, filing, prosecution, and control of appeals, post-grant reviews, defense of *inter partes* reexaminations, the prosecution of *ex parte* reexaminations, initiation or defense of interferences, derivation proceedings, requests for patent term adjustments, the defense of oppositions, and other similar proceedings with respect to such Patent Right, and any appeals therefrom. For clarity, "Prosecution and Maintenance" shall not include any other enforcement actions taken with respect to a Patent Right.

"Remaining Field" means any field outside the GrayBug Field of Use and the Kala Field of Use.

"Third Party," means a Person other than GrayBug, Kala, JHU and their respective Affiliates.

"Transferred Patent Rights" means (a) with respect to GrayBug, Kala Current Patent Rights transferred to GrayBug pursuant to Section 2(a)(v), and (b) with respect to Kala, GrayBug Current Patent Rights transferred to Kala pursuant to Section 2(a)(v).

"Valid Claim" means either: (a) a claim of an issued and unexpired patent included within the relevant Patent Rights which has not been revoked or held unenforceable, unpatentable or invalid by a decision of a court or other governmental agency of competent jurisdiction, unappealable or unappealed within the time allowed for appeal, and which has not been disclaimed, denied or admitted to be invalid or unenforceable through reexamination, reissue, disclaimer or otherwise; or (b) a claim of a pending patent application included within the relevant Patent Rights, which claim has not been abandoned or finally disallowed without the possibility of appeal or refiling of such application, and has been pending for less than [\*\*] years from the date such claim was filed in a first national filing non-provisional patent application in the country of interest and has not been (i) canceled, (ii) withdrawn from consideration, (iii) finally determined to be unallowable by the applicable governmental authority (and from which no appeal is or can be taken), or (iv) abandoned.

## 2. <u>License Grants</u>.

## a. Current Patent Rights.

- i. <u>GrayBug Current Patent Rights</u>. GrayBug hereby grants to Kala a worldwide, royalty-free, exclusive, sublicenseable (in accordance with Section 2(a)(iii)) sublicense, under the GrayBug Current Patent Rights, to research, develop, make, have made, use, have used, import, have imported, offer to sell, have offered for sale, sell and have sold Licensed Products in the Kala Field of Use, and to provide and have provided Licensed Services in the Kala Field of Use, in the United States and worldwide.
- ii. <u>Kala Current Patent Rights</u>. Kala hereby grants to GrayBug a worldwide, royalty-free, exclusive, sublicenseable (in accordance with Section 2(a)(iii)) sublicense, under the Kala Current Patent Rights, to research, develop, make, have made, use, have used, import, have imported, offer to sell, have offered for sale, sell and have sold Licensed Products in the GrayBug Field of Use, and to provide and have provided the Licensed Services in the GrayBug Field of Use, in the United States and worldwide.

- iii. Sublicensing. Each Party may sublicense the rights granted to it under the other Party's Current Patent Rights, including the right to further sublicense; provided that (A) any such sublicense must involve some aspect of collaboration, joint research, development, manufacture, partnership, or the like, and in any event, the grant of sublicenses beyond [\*\*] more tiers is not permitted without the original sublicensing Party's written consent, (B) all sublicensees (whether direct or through any tier) are bound by the applicable terms of this Agreement, and (C) the Party granting such sublicense (directly or through its sublicensees) provides the other Party notice of such sublicensee (1) within [\*\*] Business Days after the execution of such sublicense agreement for a direct sublicense or (2) within [\*\*] Business Days after execution of such sublicense agreement for a further sublicense granted through its sublicensee, and in any event of (C)(1) or (C)(2) of this Section 2(a)(iii), in each case to the extent permissible by law (including securities laws), prior to any public disclosure or press release for any such sublicense agreement.
- iv. Separation of Obligations. Neither Party granted a sublicense under the other Party's Current Patent Rights pursuant to Section 2(a)(i) or 2(a)(ii), or any Party's sublicensees under Section 2(a)(iii), shall be responsible for complying with the diligence, reporting, audit or other obligations under such other Party's Existing JHU License Agreement, except to the extent incorporated into this Agreement, and, except to the extent provided herein, such Party shall comply with the diligence, reporting, audit or other obligations to JHU under its own Existing JHU License Agreement with respect to such other Party's Current Patent Rights to the same extent as if such other Party's Current Patent Rights were "PATENT RIGHTS" under such Party's own Existing JHU License Agreement.
- v. <u>Termination of Direct Licensee</u>. Upon a termination, on a Current Patent Right-by-Current Patent Right and country-by-country basis, of the Existing JHU License Agreement with respect to a Party, which termination is not then being disputed by such Party, such terminated Current Patent Right shall be automatically licensed to the other Party under such other Party's Existing JHU License Agreement on a royalty-free, sublicenseable, exclusive basis (the "<u>Transferred Patent Rights</u>") with no rights thereunder retained by such terminated(ing) Party; <u>provided that</u> such license will be subject to any Third Party sublicenses that had been granted by such terminated(ing) Party that JHU continues to honor as a direct license. JHU shall provide the other Party written notice of the termination of a Party's rights with respect to such Current Patent Right under its Existing JHU License Agreement, and the other Party shall have [\*\*] Business Days from its receipt of the notice to send JHU a written notice that the Party declines to accept the automatic license described in this section.

The other Party shall not be obligated to pay any costs or fees related to the terminated(ing) Party's Current Patent Right unless and until such Party accepts the automatic license by not providing a notice to JHU declining such automatic license within the time period specified in the preceding sentence.

vi. Termination of Sublicensee. If, pursuant to Section 9(c) or in accordance with the terms of its Existing JHU License Agreement, a Party sublicensed under any of the other Party's Current Patent Rights terminates (or JHU terminates, which termination is not then being disputed by the sublicensed Party) the sublicense to such sublicensed Party with respect to any of such directly licensed Party's Current Patent Rights, any Third Party sub-sublicenses, if any, that were granted by such terminated(ing) sublicensed Party under such directly licensed Party's Current Patent Rights shall be deemed direct sublicenses (each such Third Party, a "New Sublicensee") of such directly licensed Party under such Current Patent Rights, provided that such Third Party's sub-sublicense will automatically terminate if, at such time, the Third Party is not in good standing or in breach of any term of its sub-sublicense agreement. Such New Sublicensee shall remain subject to the terms of such New Sublicensee's agreement with the terminated(ing) sublicensed Party, a copy of which shall be promptly provided to the directly licensed Party; provided, however, that such directly licensed Party shall have no obligations to such New Sublicensee beyond respecting the sublicense.

# b. Other JHU Patent Rights.

i. <u>License Grant</u>. JHU hereby grants (A) to GrayBug a worldwide, royalty-bearing (in accordance with the terms and conditions of the GrayBug/JHU Agreement, as amended herein, and specifically as amended by Section 4), sublicenseable (in accordance with Section 2(b)(ii)), exclusive license, under Other JHU Patent Rights, to research, develop, make, have made, use, have used, import, have imported, offer to sell, have offered for sale, sell and have sold Licensed Products in the GrayBug Field of Use, and to provide and have provided Licensed Services in the GrayBug Field of Use, in the United States and worldwide, and (B) to Kala a worldwide, royalty-bearing (in accordance with the terms and conditions of the Kala/JHU Agreement, as amended herein, and specifically as amended by Section 4), sublicenseable (in accordance with Section 2(b) (ii)), exclusive license, under Other JHU Patent Rights, to research, develop, make, have made, use, have used, import, have imported, offer to sell, have offered for sale, sell and have sold Licensed Products in the Kala Field of Use, and to provide and have provided Licensed Services in the Kala Field of Use, in the United States and worldwide. If, pursuant to Section 9(c) or in

accordance with the terms of its Existing JHU Agreement, a Party terminates (or JHU terminates, which termination is not then being disputed by such Party) the license to such Party in such Party's Exclusive Field under this Section 2(b)(i) with respect to an Other JHU Patent Right, such Exclusive Field with respect to such Other JHU Patent Right shall be automatically licensed to the other Party under such other Party's Existing JHU License Agreement with no rights thereunder retained by such terminated(ing) Party, and all sublicensees receiving sublicenses granted by such terminating Party in such Exclusive Field under such Other JHU Patent Right shall be deemed sublicensees of JHU. JHU shall provide the other Party written notice of the termination of a Party's license in its Exclusive Field with respect to an Other JHU Patent Right, and the other Party shall have [\*\*] Business Days from its receipt of the notice to send JHU a written notice that the Party declines to accept the automatic license described in this section. The other Party shall not be obligated to pay the portion of any costs or fees attributable to or to be paid by the terminated(ing) Party for such Other JHU Patent Rights incurred after the termination unless and until such Party accepts the automatic license by not providing a notice to JHU declining such automatic license within the time period specified in the preceding sentence.

ii. Sublicensing. Each Party may sublicense the exclusive rights granted to it under the Other JHU Patent Rights, including the right to further sublicense; provided that (A-) such further sublicense must involve some aspect of collaboration, joint research, development, manufacture, partnership, or the like, and in any event, the grant of sublicenses beyond [\*\*] more tiers is not permitted without the original sublicensing Party's written consent, (B) all sublicensees (whether direct or through any tier) are bound by the applicable terms of this Agreement, and (C) the Party granting such sublicense (directly or through its sublicensees) provides the other Party notice of such sublicensee (1) within [\*\*] Business Days after the execution of such sublicense agreement for a direct sublicense or (2) within [\*\*] Business Days after execution of such sublicense agreement for a further sublicense granted through its sublicensee, and in any event of (C)(1) or (C)(2) of this Section 2(b)(ii), in each case to the extent permissible by law (including securities laws), prior to any public disclosure or press release for any such sublicense agreement.

## c. Future Patent Rights.

 Notice. Promptly after each of the following, JHU shall provide written notice concurrently to each of GrayBug and Kala, attaching a copy of the relevant invention disclosure, report, or patent application, as applicable: (1) JHU Technology Transfer's receipt, within [\*\*] after the Settlement Agreement Effective Date, of an invention disclosure or other similar official report for an invention which, if patented, would be a Future Patent Right or the filing, within [\*\*] after the Settlement Agreement Effective Date, of the first patent application in a family for any Future Patent Right (a "First Notice"), and (2) the filing, within [\*\*] after the Settlement Agreement Effective Date, of the first non-provisional application (in any country or under the Patent Cooperation Treaty) in a family for any Future Patent Right or the filing, even if after the [\*\*] of the Settlement Agreement Effective Date, of the first non-provisional application (in any country or under the Patent Cooperation Treaty) in a family for any Future Patent Right which was, in whole or in part, the subject of an invention disclosure, report or patent application for which JHU provided, or was obligated to provide, a First Notice (a "Second Notice"). Except as provided in this Agreement, until such time as all notices under this paragraph and under Section 2(c)(vi) have been given to the Parties and the periods for the Parties to respond to such notices have expired, JHU shall not offer to license or license to any other Person any Future Patent Right or the subject matter of any invention disclosure or other similar report for an invention which, if patented, would be a Future Patent Right.

- ii. Exercise of Right. Subject to Section 2(c)(ii)(3), below, upon receiving a notice from JHU pursuant to Section 2(c)(i), each Party shall have [\*\*] Business Days thereafter to send written notice to JHU and to the other Party, to exercise its right to non-exclusively license such Future Patent Rights in its Exclusive Field in accordance with the provisions of Sections 2(c)(iii)-(v) below, as applicable.
  - 1. If, after receipt of the First Notice under Section 2(c)(i)(1) with respect to Patent Rights, only one Party exercises its right to license such Future Patent Rights (the "First Requesting Party"), then during the period between such exercise and the expiration of the period for the Parties' written notices in response to the Second Notice under Section 2(c)(i)(2) with respect to such Future Patent Rights, such Future Patent Rights shall be subject to either Section 2(c)(iv) or (v), as applicable, but the First Requesting Party may only grant sublicenses in the other Party's Exclusive Field if such sublicenses are subject to termination in such Exclusive Field if such other Party obtains an exclusive license in such other Party's Exclusive Field pursuant to Section 2(c)(vi). Thereafter, (A) if the other Party does not exercise its right to license such Future Patent Rights pursuant to the Second Notice, such Future Patent Rights shall remain subject to Section 2(c)(iv) or (v), as applicable, or (B) if the other Party, within [\*\*] Business Days after receipt of such Second

Notice, sends a written notice to JHU and the First Requesting Party requesting a license to such Future Patent Rights, then such Future Patent Rights will be subject to Section 2(c)(iii), and, per Section 2(c)(iv) or 2(c)(v), as applicable, such other Party shall reimburse the First Requesting Party for the pro rata portion of all patent costs the First Requesting Party has paid to JHU in connection with such Future Patent Rights up to that time, subject to the provision of an invoice with respect thereto and, to the extent requested by such other Party, any relevant supporting documentation.

- 2. If, after receipt of the First Notice under Section 2(c)(i)(1) with respect to Patent Rights, both Parties exercise their right to license such Future Patent Rights, then during the period between such exercise and the expiration of the period for the Parties' written notices in response to the Second Notice under Section 2(c)(i)(2) with respect to such Future Patent Rights, such Future Patent Rights shall be subject to Section 2(c)(iii). Thereafter, (A) if both Parties exercise their right to license such Future Patent Rights pursuant to the Second Notice, such Future Patent Rights shall remain subject to Section 2(c)(iii), or (B) if only one Party, within [\*\*] Business Days after receipt of such Second Notice, sends a written notice to JHU and the other Party requesting a license to such Future Patent Rights, then such Future Patent Rights will be subject to Section 2(c)(iv) or (v), as applicable, and the formerly-licensed Party shall not be entitled to any refund of any amounts it paid to JHU in connection with such Future Patent Rights up to that time and shall remain bound to pay the relevant amounts incurred by JHU in connection with such Future Patent Rights pursuant to Section 2(c) (iii) up to that time.
- 3. Notwithstanding Sections 2(c)(ii)(1) and 2(c)(ii)(2), if JHU establishes that a Future Patent Right is (A) based on research funded by a Party or (B) specifically designed for the use of one Party, then at the request of such Party, such Future Patent Right can be licensed non-exclusively to such Party in all fields and such Future Patent Right shall be added to such Party's Current Patent Rights but on a non-exclusive (but subject to Section 2(c)(iv)), sublicenseable basis, and shall otherwise be subject to the sublicenses granted by such Party under its Current Patent Rights to the other Party pursuant to this Agreement and, for the sake of clarity, shall be subject to Sections 3(a)(viii) and 4.

- iii. <u>Co-Licensed</u>. This Section 2(c)(iii) is subject to Sections 2(c)(ii)(1) and (2) and Section 2(c)(vi).
  - Rights from JHU pursuant to Section 2(c)(i) above, send a written notice to JHU and the other Party requesting a license under such Future Patent Rights, then JHU shall be deemed to have granted (A) to GrayBug a worldwide, royalty-bearing (in accordance with the terms of the GrayBug/JHU Agreement, as amended herein, and specifically as amended by Section 4), sublicenseable (in accordance with Section 2(c)(iii)(2)), non-exclusive license, under such Future Patent Rights, to research, develop, make, have made, use, have used, import, have imported, offer to sell, have offered for sale, sell and have sold Licensed Products in the GrayBug Field of Use, and to provide and have provided Licensed Services in the GrayBug Field of Use, in the United States and worldwide, and (B) to Kala a worldwide, royalty-bearing (in accordance with the terms of the Kala/JHU Agreement, as amended herein, and specifically as amended by Section 4), sublicenseable (in accordance with Section 2(c)(iii)(2)), non-exclusive license, under such Future Patent Rights, to research, develop, make, have made, use, have used, import, have imported, offer to sell, have offered for sale, sell and have sold Licensed Products in the Kala Field of Use, and to provide and have provided Licensed Services in the Kala Field of Use, in the United States and worldwide. If, pursuant to Section 9(c) or in accordance with the terms of its Existing JHU License Agreement, a Party terminates (or JHU terminates, which termination is not then being disputed by such Party) the license to such Party in its Exclusive Field under this Section 2(c)(iii)(1) with respect to a Future Patent Right, such Exclusive Field with respect to such Future Patent Right shall be automatically licensed to the other Party under such other Party's Existing JHU License Agreement with no rights thereunder retained by such terminated(ing) Party, and all sublicensees receiving sublicenses granted by such terminated(ing) Party in such Exclusive Field under such Future Patent Right shall be deemed sublicensees of JHU. JHU shall provide the other Party written notice of the termination of a Party's license in its Exclusive Field with respect to a Future Patent Right, and the other Party shall have [\*\*] Business Days from its receipt of the notice to send JHU a written notice that the Party declines to accept the automatic license described in this section. The other Party shall not be obligated to pay the portion of any costs or fees attributable to or to be paid by the terminated(ing) Party for such

If both GrayBug and Kala, within [\*\*] Business Days after receipt of a notice for Future Patent

Future Patent Rights incurred after the termination unless and until such Party accepts the automatic license by not providing a notice to JHU declining such automatic license within the time period specified in the preceding sentence.

- 2. <u>Sublicensing</u>. Each Party may sublicense the rights granted to it under the Future Co-Licensed Patent Rights, including the right to further sublicense; <u>provided that</u> (A) such further sublicense must involve some aspect of collaboration, joint research, development, manufacture, partnership, or the like, and in any event, the grant of sublicenses beyond [\*\*] more tiers is not permitted without the original sublicensing Party's written consent, (B) all sublicensees (whether direct or through any tier) are bound by the applicable terms of this Agreement, and (C) the Party granting such sublicense (directly or through its sublicensees) provides the other Party notice of such sublicensee (1) within [\*\*] Business Days after the execution of such sublicense agreement for a direct sublicense or (2) within [\*\*] Business Days after execution of such sublicense agreement for a further sublicense granted through its sublicensee, and in any event of (C)(1) or (C)(2) of this Section 2(c)(iii)(2), in each case to the extent permissible by law (including securities laws), prior to any public disclosure or press release for any such sublicense agreement.
- iv. GrayBug Only. Subject to Sections 2(c)(ii)(1) and (2) and Section 2(c)(vi), if GrayBug, but not Kala, within [\*\*] Business Days after receipt of a notice for Future Patent Rights from JHU pursuant to Section 2(c) (i) above, sends a written notice to JHU and Kala requesting a license under such Future Patent Rights, then such Future Patent Rights ("GrayBug-Licensed Future Patent Rights") shall be deemed licensed by JHU to GrayBug as if they were "PATENT RIGHTS" as defined in the GrayBug/JHU Agreement (as amended herein, and specifically as amended by Section 4), with the exception that (1) such license is non-exclusive and sublicenseable, (2) for clarity, no license, sublicense or other rights shall be deemed to have been granted to Kala in any field with respect to such Future Patent Rights pursuant to this Agreement, and (3) if JHU grants Kala or a Third Party a license under such GrayBug-Licensed Future Patent Rights, all patent costs incurred by JHU with respect to such GrayBug-Licensed Future Patent Rights, whether prior to or after the granting of such license to Kala or such Third Party(ies), shall be borne pro rata among GrayBug and all such licensees (based on the number of JHU-licensed licensees under such GrayBug-Licensed Future Patent Rights) and, with respect to any such patent costs with respect to such GrayBug-Licensed Future Patent Rights paid by GrayBug prior to JHU

- granting such a license to Kala or any such Third Party, JHU shall ensure that Kala or such Third Party reimburses such prior patent costs to GrayBug so that GrayBug will only have borne its pro rata amount of such prior patent costs.
- Kala Only. Subject to Sections 2(c)(ii)(1) and (2) and Section 2(c)(vi), if Kala, but not GrayBug, within [\*\*] Business Days after receipt of a notice for Future Patent Rights from JHU pursuant to Section 2(c)(i) above, sends a written notice to JHU and GrayBug requesting a license under such Future Patent Rights, then such Future Patent Rights ("Kala-Licensed Future Patent Rights") shall be deemed licensed by JHU to Kala as if they were "PATENT RIGHTS" as defined in the Kala/JHU Agreement (as amended herein, and specifically as amended by Section 4), with the exception that (1) such license is non-exclusive and sublicenseable, (2) for clarity, no license, sublicense or other rights shall be deemed to have been granted to GrayBug in any field with respect to such Future Patent Rights pursuant to this Agreement, and (3) if JHU grants GrayBug or a Third Party a license under such Kala-Licensed Future Patent Rights, all patent costs incurred by JHU with respect to such Kala-Licensed Future Patent Rights, whether prior to or after the granting of such license to Kala or such Third Party(ies), shall be borne pro rata among Kala and all such licensees (based on the number of JHU-licensed licensees under such Kala-Licensed Future Patent Rights) and, with respect to any such patent costs with respect to such Kala-Licensed Future Patent Rights paid by Kala prior to JHU granting such a license to GrayBug or any such Third Party, JHU shall ensure that GrayBug or such Third Party reimburses such prior patent costs to Kala so that Kala will only have borne its pro rata amount of such prior patent costs.
- vi. Right of First Negotiation. Upon a Party's exercise of its right, pursuant to this Section 2(c), to obtain a non-exclusive license in its Exclusive Field with respect to Future Patent Rights, JHU shall automatically grant to such Party an exclusive right to negotiate an exclusive license under such Future Patent Rights in its Exclusive Field, which right such Party may exercise by providing written notice to JHU within [\*\*] days after such Party exercised its right pursuant to this Section 2(c) to obtain a non-exclusive license. If such Party exercises such right of negotiation, JHU and such Party shall negotiate in good faith (for a period not to exceed [\*\*] days after notice of exercise, extendable upon mutual agreement) on reasonable terms to make the license to such Party under such Future Patent Rights exclusive in such Party's Exclusive Field. In the event a particular Licensed Product or Licensed Service is covered by both such Party's Existing JHU License Agreement and any other agreement or amendment between such Party and JHU for the Future Patent Right(s), [\*\*] for a particular milestone payment will be due to JHU and JHU will apportion such revenue

amongst the relevant agreements according to JHU's internal policies. The failure of such Party to exercise such right to negotiate, or the failure of such Party and JHU to reach agreement on the terms on which the license to such Party under such Future Patent Rights shall be made exclusive in its Exclusive Field, shall not affect such Party's non-exclusive license under such Future Patent Rights in its Exclusive Field.

vii. <u>Hart Scott Rodino Compliance</u>. If required, each Party receiving a license under Future Patent Rights shall use commercially diligent efforts to satisfy any applicable requirements under the HSR Act, and the regulations promulgated thereunder or other applicable Federal Trade Commission or Department of Justice regulations, including by making an initial HSR Act filing as soon as practicable after the exercise of the option to obtain such a license pursuant to this Section 2(c). The licenses granted under this Section 2(c) with respect to such Future Patent Rights shall not be effective until the date of either (A) satisfaction of any such requirements and the expiration or termination of all applicable HSR Act waiting periods (including any extensions thereof) or (B) the mutual written agreement of JHU and each Party receiving a license under such Future Patent Rights that no HSR Act filings are required.

## viii. Potential Future Patent Rights.

- 1. JHU hereby represents to each Party that (A) to JHU's knowledge as of the Settlement Agreement Effective Date, JHU Technology Transfer has not received an invention disclosure or other similar official report for a Potential Future Patent Right and no provisional or non-provisional patent application has been filed in any country or under the Patent Cooperation Treaty for any Potential Future Patent Right, other than those disclosed in writing by JHU Technology Transfer to GrayBug and Kala, (B) all the GrayBug Current Patent Rights existing on the Settlement Agreement Effective Date are listed in Schedule 1.21, (C) all the Kala/JHU Jointly Owned Patent Rights existing on the Settlement Agreement Effective Date are listed in Schedule 1.22, and (D) all the Kala Current Patent Rights existing on the Settlement Agreement Effective Date are listed in Schedule 1.23.
- 2. JHU shall provide written notice concurrently to each of GrayBug and Kala, promptly after JHU's receipt of an invention disclosure or other similar report, or JHU's knowledge of the filing of a provisional or non-provisional patent application, for a Potential Future Patent Right, if such receipt occurs or such knowledge is gained at any time prior to the [\*\*] of the Settlement Agreement Effective Date or thereafter with

respect to the filing of the first non-provisional application. Such notice from JHU shall initially be a non-confidential disclosure. If either GrayBug or Kala requests confidential information regarding the Potential Future Patent Right, JHU shall use reasonable efforts to comply with the request by seeking permission from any necessary Third Parties and providing such information if necessary under an appropriately tailored confidentiality agreement.

ix. JHU represents that Schedule 2(c)(ix) lists each Co-Inventor from which JHU had not, as of the Settlement Agreement Effective Date, received authorization in writing to license the relevant invention under this Settlement Agreement (each such individual, a "Pending Co-Inventor"). JHU shall use reasonable efforts to obtain authorization in writing from each such Pending Co-Inventor to license the relevant invention under this Settlement Agreement as soon as possible after the Settlement Agreement Effective Date and shall notify each Party as it obtains such authorization from a Pending Co-Inventor. Authorization shall be at each Pending Co-inventor's personal discretion. In the event JHU does not secure an individual Pending Co-inventor's authorization after using reasonable efforts to obtain such by January 1, 2015, the individual will no longer be a Co-inventor under this Agreement and be deleted from the definition of "Future Patent Rights".

## d. Kala/JHU Jointly Owned Patent Rights.

- i. <u>Kala/JHU Jointly Owned Patent Rights</u>. Kala hereby grants to GrayBug a worldwide, royalty-free, exclusive, sublicenseable (in accordance with Section 2(d)(ii)) (A) license, under Kala's ownership interest in the Kala/JHU Jointly Owned Patent Rights, and (B) sublicense, under JHU's ownership interest in the Kala/JHU Jointly Owned Patent Rights, in each case of (A) and (B) to research, develop, make, have made, use, have used, import, have imported, offer to sell, have offered for sale, sell and have sold Licensed Products in the GrayBug Field of Use, and to provide and have provided the Licensed Services in the GrayBug Field of Use, in the United States and worldwide.
- ii. <u>Sublicensing</u>. GrayBug may sublicense the rights granted to it under the Kala/JHU Jointly Owned Patent Rights, including the right to further sublicense; <u>provided that</u> (A) any such sublicense must involve some aspect of collaboration, joint research, development, manufacture, partnership, or the like, and in any event, the grant of sublicenses beyond [\*\*] more tiers is not permitted without Kala's written consent, (B) all sublicensees (whether direct or through any tier) are bound by the applicable terms of this Agreement, and (C) GrayBug (directly or through its sublicensees) provides Kala notice of such

- sublicensee (1) within [\*\*] Business Days after the execution of such sublicense agreement for a direct sublicense or (2) within [\*\*] Business Days after execution of such sublicense agreement for a further sublicense granted through its sublicensee, and in any event of (C)(1) or (C)(2) of this Section 2(d)(ii), in each case to the extent permissible by law (including securities laws), prior to any public disclosure or press release for any such sublicense agreement.
- iii. Separation of Obligations. GrayBug shall not be responsible for complying with the diligence, reporting, audit or other obligations under the Kala/JHU Agreement, except to the extent incorporated into this Agreement, and Kala shall not be required to make GrayBug comply with such obligations, and, except to the extent provided herein, GrayBug shall comply with the diligence, reporting, audit or other obligations to JHU under the GrayBug/JHU Agreement with respect to the Kala/JHU Jointly Owned Patent Rights to the same extent as if such Kala/JHU Jointly Owned Patent Rights were "PATENT RIGHTS" under the GrayBug/JHU Agreement.
- iv. Termination of Direct Licensee. Upon a termination, in accordance with the terms of the Kala/JHU Agreement, of Kala's rights under JHU's ownership interest in any Kala/JHU Jointly Owned Patent Rights (which termination is not then being disputed by Kala), (A) the license granted by Kala to GrayBug under Kala's ownership interest in such Kala/JHU Jointly Owned Patent Rights shall terminate, and (B) JHU's ownership interest in such Kala/JHU Jointly Owned Patent Rights shall be automatically licensed to GrayBug under the GrayBug/JHU Agreement on a royalty-free, sublicenseable, exclusive basis (the "Jointly Owned Transferred Patent Rights"); provided that such license will be subject to any Third Party sublicenses that had been granted by Kala under JHU's ownership interest in such Kala/JHU Jointly Owned Patent Rights that JHU or Kala continues to honor as a direct license. JHU shall provide GrayBug written notice of the termination of Kala's rights under JHU's ownership interest in such Kala/JHU Jointly Owned Patent Rights, and GrayBug shall have [\*\*] Business Days from its receipt of the notice to send JHU a written notice that GrayBug declines to accept the automatic license described in this section. GrayBug shall not be obligated to pay any costs or fees related to the Kala/JHU Jointly Owned Patent Rights incurred after the termination unless and until GrayBug accepts the automatic license by not providing a notice to JHU declining such automatic license within the time period specified in the preceding sentence.
- v. <u>Termination of Sublicensee</u>. If, pursuant to Section 9(c), GrayBug terminates the license to GrayBug under Kala's ownership interest in any Kala/JHU Jointly

Owned Patent Rights, the license granted by Kala to GrayBug under Kala's ownership interest in such Kala/JHU Jointly Owned Patent Rights shall terminate, and any Third Party sublicenses, if any, that were granted by GrayBug under Kala's ownership interest in such Kala/JHU Jointly Owned Patent Rights shall terminate. If, pursuant to Section 9(c) or in accordance with the terms of its Existing JHU License Agreement, GrayBug terminates (or JHU terminates, which termination is not then being disputed by GrayBug) the sublicense to GrayBug with respect to any Kala/JHU Jointly Owned Patent Rights, (A) the license granted by Kala to GrayBug under Kala's ownership interest in such Kala/JHU Jointly Owned Patent Rights shall terminate, and any Third Party sublicenses, if any, that were granted by GrayBug under Kala's ownership interest in such Kala/JHU Jointly Owned Patent Rights shall terminate, and (B) any Third Party sub-sublicenses, if any, that were granted by GrayBug under JHU's ownership interest in such Kala/JHU Jointly Owned Patent Rights shall be deemed direct sublicenses (each such Third Party, a "New Direct Sublicensee") of Kala under such Kala/JHU Jointly Owned Patent Rights; provided that a Third Party subsublicense will automatically terminate if the Third Party is not in good standing or in breach of any term of its sub-sublicense agreement. Each New Direct Sublicensee shall remain subject to the terms of such New Direct Sublicensee's agreement with GrayBug, a copy of which shall be promptly provided to Kala; provided, however, that Kala shall have no obligations to such New Direct Sublicensee beyond respecting the sublicense.

- vi. <u>Kala's Ownership Interest</u>. For the avoidance of doubt, the terms of this Agreement do not in any way affect Kala's ownership interest in each of the Kala/JHU Jointly Owned Patent Rights.
- e. <u>Government Rights</u>. All exclusive rights granted to a Party hereunder are subject to rights retained by the United States Government, if any, in accordance with the Bayh-Dole Act of 1980 (established by P.L. 96-517 and amended by P.L. 98-620, codified at 35 USC § 200 et. seq. and implemented according to 37 CFR Part 401) (the "<u>Bayh-Dole Act</u>"), and subject to the retained right of JHU to make, have made, provide and use for its and The Johns Hopkins Health Systems' non-commercial academic research and teaching purposes Licensed Products and Licensed Services, including the ability to distribute any biological material disclosed and/or claimed in any of the JHU Patent Rights and Kala/JHU Jointly Owned Patent Rights licensed or sublicensed hereunder for nonprofit non-commercial academic research use to non-commercial entities as is customary in the scientific community.

## f. Amendment to Know-how Grants.

- i. The GrayBug/JHU Agreement shall be amended by revising Paragraph 2.1(ii) by revising "provided that JHU will not grant a license to KNOW HOW AND MATERIALS to any other commercial entity" to read "provided that JHU will not grant a license to KNOW HOW AND MATERIALS to any commercial entity, other than Kala Pharmaceuticals, Inc. (its successors and assigns) ("Kala") and any entity which controls, is controlled by or is under common control with Kala (where "control" means the direct or indirect ownership of at least fifty percent (50%)), without Company's prior written approval; provided, however, that(i) JHU shall be free to transfer KNOW HOW AND MATERIALS to commercial entities for such entities' non-commercial research purposes, and, (ii) nothing in this Paragraph shall prevent JHU from performing sponsored research using KNOW HOW AND MATERIALS for a commercial entity". For the purpose of clarification, KNOW HOW AND MATERIALS is limited to materials, information, records and data licensed to GrayBug under the GrayBug/JHU Agreement prior to the Settlement Agreement Effective Date.
- ii. The Kala/JHU Agreement shall be amended by revising Paragraph 2.1(ii) by revising "provided that JHU will not grant a license to KNOW HOW AND MATERIALS to any other commercial entity" to read "provided that JHU will not grant a license to KNOW HOW AND MATERIALS to any commercial entity, other than GrayBug, LLC (its successors and assigns) ("GrayBug") and any entity which controls, is controlled by or is under common control with GrayBug (where "control" means the direct or indirect ownership of at least fifty percent (50%)), without Company's prior written approval; provided, however(i) JHU shall be free to transfer KNOW HOW AND MATERIALS to commercial entities for such entities' non-commercial research purposes, and, (ii) nothing in this Paragraph shall prevent JHU from performing sponsored research using KNOW HOW AND MATERIALS for a commercial entity". For the purpose of clarification, KNOW HOW AND MATERIALS is limited to materials, information, records and data licensed to Kala under the Kala/JHU Agreement prior to the Settlement Agreement Effective Date.
- g. <u>Covenant</u>. On a Patent Right-by-Patent Right and country-by-country basis, Kala agrees that (i) it shall not exercise the applicable license or sublicense granted to it in the Kala Field of Use under JHU's interest in the GrayBug Current Patent Rights, Kala Current Patent Rights, Other JHU Patent Rights, Kala/JHU Jointly Owned Patent Rights or Future Patent Rights where the use involves a Particle comprising or consisting of [\*\*], and (ii) with respect to any Kala/JHU Jointly Owned Patent Right, Kala shall not, prior to the termination of the license granted to Kala under JHU's interest in such Kala/JHU Jointly

Owned Patent Right, exploit its rights under its ownership interest in such Kala/JHU Jointly Owned Patent Right to develop and commercialize a product in the Kala Field of Use where the Particle therein comprises or consists of [\*\*].

## 3. <u>Prosecution and Enforcement.</u>

# a. Prosecution and Maintenance

- Except as set forth in this Section 3, (i) Co-Licensed Other JHU Patent Rights and Future Co-Licensed Patent Rights shall be treated as if they were "PATENT RIGHTS" as defined in each Existing JHU License Agreement for purposes of determining JHU's and each Party's respective rights and obligations with respect to the Prosecution and Maintenance thereof, and, following the termination of one Party's rights in its Exclusive Field with respect to any Co-Licensed Other JHU Patent Rights or Future Co-Licensed Patent Rights, such Patent Rights shall be treated as if they were "PATENT RIGHTS" as defined in the other Party's Existing JHU License Agreement for purposes of determining JHU's and such other Party's respective rights and obligations with respect to the Prosecution and Maintenance thereof, (ii), for clarity, JHU shall continue to have the primary responsibility for Prosecution and Maintenance of, (A) in accordance with the GrayBug/JHU Agreement, the GrayBug Current Patent Rights, the GrayBug-Licensed Other Patent Rights, any Transferred Patent Rights that are transferred to GrayBug and any GrayBug-Licensed Future Patent Rights, and (B) in accordance with the Kala/JHU Agreement, the Kala Current Patent Rights, the Kala-Licensed Other Patent Rights, any Transferred Patent Rights that are transferred to Kala and any Kala-Licensed Future Patent Rights, and (iii) Kala shall continue to have the primary responsibility for Prosecution and Maintenance of the Kala/JHU Jointly Owned Patent Rights in accordance with the Kala/JHU Agreement; provided, however, that:
  - GrayBug shall have the primary interactions with JHU (as defined below) regarding Prosecution and Maintenance of the GrayBug Current Patent Rights in accordance with the GrayBug/JHU Agreement; <u>provided</u>, <u>however</u>, that JHU or GrayBug, as applicable, shall timely copy Kala on all correspondence that JHU or GrayBug, respectively, receives from or files with any patent office regarding the GrayBug Current Patent Rights and Kala may provide comments for consideration by JHU and GrayBug with respect thereto.
  - Kala shall have the primary interactions with JHU (as defined below) regarding Prosecution and Maintenance of the Kala Current Patent Rights in accordance with the Kala/JHU Agreement; provided, however,

- that JHU or Kala, as applicable, shall timely copy GrayBug on all correspondence that JHU or Kala, respectively, receives from or files with any patent office regarding the Kala Current Patent Rights and GrayBug may provide comments for consideration by JHU and Kala with respect thereto.
- 3. With regard to Prosecution and Maintenance of the Kala/JHU Jointly Owned Patent Rights, Kala shall timely copy JHU and GrayBug on all correspondence that Kala receives from or files with any patent office regarding the Kala/JHU Jointly Owned Patent Rights, and GrayBug and JHU may provide comments for consideration by Kala with respect thereto.
- 4. As used in this Section 3(a), "primary interaction with JHU" means that, to the extent that JHU receives conflicting input from each Party with respect to the relevant Current Patent Rights, JHU shall notify both Parties of such conflicting input but, unless the Parties subsequently agree with respect to their input to JHU, JHU shall weigh more heavily the input of the Party directly licensed by JHU under such Current Patent Rights.
- 5. JHU will select patent counsel (the "JHU Patent Counsel") responsible for the Prosecution and Maintenance of the JHU Patent Rights that is acceptable to JHU and both Parties. Kala will select patent counsel (the "Other Patent Counsel") responsible for Prosecution and Maintenance of the Kala/JHU Jointly Owned Patent Rights that is acceptable to JHU and both Parties. For the avoidance of doubt, this Agreement does not require JHU or Kala, respectively, to change any current JHU Patent Counsel or Other Patent Counsel.
- 6. If JHU or GrayBug elects not to have a patent application filed in any particular country or not to Prosecute and Maintain (or pay expenses associated with Prosecuting or Maintaining) any GrayBug Current Patent Right or claims therein and the other such party (GrayBug or JHU) does not, pursuant to its rights under the GrayBug/JHU Agreement, require such patent application to be filed, Prosecuted and Maintained or such expenses to be paid, then (A) JHU or GrayBug, respectively, shall within [\*\*] Business Days provide written notice thereof to GrayBug or JHU, as the case may be, and to Kala, (B) such Patent Right shall thereafter no longer be licensed by JHU to GrayBug and shall thereafter no longer be considered a GrayBug Current Patent Right hereunder, (C) Kala may elect (by written notice sent to JHU within [\*\*] Business Days after

receipt of the notice in clause (A)) to include such Patent Right as a "PATENT RIGHT" under the Kala/JHU Agreement, to be exclusively licensed to Kala in all fields, in which case such Patent Right shall not be licensed or sublicensed to GrayBug, with any sublicenses granted by GrayBug subject to Section 2(a)(v).

- 7. If JHU or Kala elects not to have a patent application filed in any particular country or not to Prosecute and Maintain (or pay expenses associated with Prosecuting and Maintaining) any Kala Current Patent Right or claims therein and the other such party (Kala or JHU) does not, pursuant to its rights under the Kala/JHU Agreement, require such patent application to be filed, Prosecuted and Maintained or such expenses to be paid, then (A) JHU or Kala, respectively, shall within [\*\*] Business Days provide notice thereof to Kala or JHU, as the case may be, and to GrayBug, (B) such Patent Right shall thereafter no longer be icensed by JHU to Kala and shall thereafter no longer be considered a Kala Current Patent Right hereunder, (C) GrayBug may elect (by written notice sent to JHU within [\*\*] Business Days after receipt of the notice in clause (A)) to include such Patent Right as a "PATENT RIGHT" under the GrayBug /JHU Agreement, to be exclusively licensed to GrayBug in all fields, in which case such Patent Right shall not be licensed or sublicensed to Kala, with any sublicenses granted by Kala subject to Section 2(a)(v).
- ii. At least [\*\*], one representative from each Party and JHU shall meet (in person or by tele-conference or video-conference) along with the JHU Patent Counsel to discuss the status, strategy, and any other relevant matters with respect to the Prosecution and Maintenance of the JHU Patent Rights in which both Parties retain a license or sublicense, <u>provided that</u> at least [\*\*] shall be in person. Each Party and JHU may replace its representative at any time upon written notice to the other Party(ies) and JHU, as applicable.
- iii. To the extent possible and legally permissible during Prosecution and Maintenance of JHU Patent Rights, at the request of either GrayBug or Kala, patent claims shall be split out in separate applications to facilitate each Party's separate commercial purposes.
- iv. JHU shall give reasonable consideration to any comment, suggestion, recommendation or other request (a "Request") of each Party regarding the Prosecution and Maintenance of the Co-Licensed Other JHU Patent Rights and Future Co-Licensed Patent Rights; provided that such Request is received by JHU in a reasonably sufficient time before the applicable filing date to consider such

Request. For Requests under this subsection, Kala and GrayBug shall work together to coordinate and compile their respective Requests into a joint Request (each a "Joint Request"), and for each Co-Licensed Other JHU Patent Right and Future Co-Licensed Patent Right, will designate which Party shall take the lead on conveying the Parties' Joint Request to JHU. If the Parties are unable to agree on a Joint Request for a matter, then the Parties shall jointly contact JHU to discuss their respective positions and JHU will in its sole discretion decide what to advise the JHU Patent Counsel with regard to that matter. For administrative purposes only, a "lead" Party will be nominated to coordinate the Parties' Joint Request for each Co-Licensed Other JHU Patent Right and Future Co-Licensed Patent Right, with the initial nominations decided at the first Prosecution and Maintenance meeting between the Parties after the Settlement Agreement Effective Date, and revised as necessary or appropriate at future meetings.

- With respect to Co-Licensed Other JHU Patent Right and the Future Co-Licensed Patent Rights, each Party (i.e., Kala and GrayBug) shall be responsible for [\*\*]percent ([\*\*]%) of JHU's costs (subject to clause (viii) below) incurred in the Prosecution and Maintenance thereof, and each Party shall reimburse JHU for such amount in accordance with the timelines and processes set forth in the Party's Existing JHU License Agreement, including for such patent costs with respect to such Patent Rights incurred prior to execution of this Agreement. From the point in time at which Other JHU Patent Rights or Future Co-Licensed Patent Rights are exclusively licensed to only one Party hereunder, such Party alone shall be solely responsible for JHU's costs incurred thereafter in the Prosecution and Maintenance of such Other JHU Patent Rights and Future Co-Licensed Patent Rights, in accordance with such Party's Existing JHU License Agreement. In the event a Party or a Third Party(ies) licenses from JHU any rights in the Remaining Field under the Co-Licensed Other JHU Patent Right, the patent costs incurred by JHU with respect to the Prosecution and Maintenance of such Patent Right, whether prior to or after the granting of such license to such Third Party(ies), shall be prorated amongst Kala, GrayBug, and such Third Party(ies), and the relevant Party ((the "Remaining Field Party") if one Party, but not both Parties, have been granted a license by JHU in the Remaining Field), based on the total number of licensees (counting such Remaining Field Party twice (once for its right in its Exclusive Field and once for its right in the Remaining Field) under such Patent Right and JHU shall refund to Kala and GrayBug the relevant amount received from such Third Party(ies) or such Remaining Field Party, as applicable.
- vi. The Parties and JHU each acknowledge the duties set forth in 37 C.F.R. § 1.56.

- vii. The Parties and JHU each acknowledge that patents falling within Patent Rights sublicensed or co-licensed under this Agreement may be eligible for patent term extension under 35 U.S.C. § 156, applications for which must be submitted by the owner of record or their agent. If a Party obtains approval of a New Drug Application ("NDA") from the U.S. Food and Drug Administration ("FDA") for a product and believes a patent falling within Patent Rights hereunder may be eligible for patent term extension, it shall notify JHU and the other Party, and JHU and the NDA-holding Party shall cooperate in connection with preparing and timely filing an application for patent term extension.
- viii. Notwithstanding anything to the contrary herein, a Party which has received, directly or indirectly, only non-exclusive rights in such Party's Exclusive Field under a Future Patent Right in a country shall be responsible for [\*\*] percent ([\*\*]%) of the Prosecution and Maintenance costs such Party would otherwise be responsible for with respect to such Future Patent Right in such country; provided, however, that, in the event a Party or a Third Party(ies) licenses from JHU any rights in such Future Patent Right, the patent costs incurred by JHU with respect to the Prosecution and Maintenance of such Patent Right, whether prior to or after the granting of such license to such Party or Third Party(ies), shall be prorated amongst each Party with a license from JHU under such Future Patent Right and such Third Party(ies).

#### b. <u>Enforcement.</u>

- i. Orange Book Listings & Hatch Waxman Litigation.
  - 1. The Parties acknowledge that the owner of an NDA has a non-delegable duty to determine which patents should be listed in the U.S. FDA "Approved Drug Products with Therapeutic Equivalence Evaluations" (sometimes referred to as the "Orange Book") and any foreign counterparts which require such, including the so-called "Patent Register" listings required in Canada. Prior to such listings and acts, the Parties will meet to evaluate and identify whether any patents falling within Patent Rights sublicensed or co-licensed to the non-NDA holding Party under this Agreement should be included, and the NDA-holding Party shall reach its own decision regarding which patents within such Patent Rights to list in the Orange Book or any foreign counterpart.
  - 2. If an NDA-holding Party lists in the Orange Book or foreign counterpart a patent also licensed to the non-NDA holding Party, it shall give the non-NDA holding Party written notice within [\*\*] Business Days.

- 3. If the NDA-holding Party receives written notice from a generic company under 21 U.S.C. 355(j)(2) (A)(vii)(IV) (a "Paragraph IV Certification") that it has filed an Abbreviated New Drug Application to sell a product covered by the NDA, and has certified that a patent listed in the Orange Book that falls within Patent Rights sublicensed or co-licensed to the non-NDA holding Party under this Agreement is either not infringed, invalid, or not enforceable, the NDA-holding Party shall give written notice to the non-NDA holding Party within [\*\*] Business Days, with a copy of the Paragraph IV Certification. The Parties acknowledge that time is of the essence in such matter given the statutory requirement that an NDA-holding Party bring a lawsuit within [\*\*] days of receiving the Paragraph IV Certification.
- The NDA-holding Party shall promptly discuss with JHU and the non-NDA-holding Party the advisability of filing a lawsuit against the generic company under 21 U.S.C. § 355 (the "Hatch Waxman Act"). If GrayBug is the NDA-holding Party and the Patent Right challenged is a GrayBug Current Patent Right, then GrayBug shall control the litigation, and if it chooses not to litigate, then JHU shall have the secondary right of litigation. If Kala is the NDA-holding Party and the Patent Right challenged is a Kala Current Patent Right or a Kala/JHU Joint Patent Right, then Kala shall control the litigation, and if it chooses not to litigate, then JHU shall have the secondary right of litigation. If GrayBug or Kala is the NDA-holding Party and the Patent Right challenged is a Co-Licensed Other JHU Patent Right or a Future Co-Licensed Patent Right, then GrayBug and Kala shall meet to discuss terms under which the litigation shall take place, which terms shall include that the NDA-holding Party shall pay the cost of the litigation, the Non-NDA holding Party shall have the right of close communication and review of filed litigation documents (except to the extent marked highly business confidential), participation in (but not control over) litigation strategy, and if the NDA-holding Party receives an adverse judgment of invalidity from a court of final jurisdiction, the NDA-holding Party shall compensate the non-NDA-holding Party for all fees and costs it has invested in the litigated patent historically as well as its litigation attorney's fees.

- ii. Current Patent Rights (Other Than Hatch Waxman).
  - 1. Each Party and JHU will notify the other Party(ies) and JHU, as applicable, promptly in writing when any infringement of any Current Patent Right by a Third Party is uncovered or suspected.
  - 2. Except as otherwise provided in Section 3(b)(i), with respect to an actual or suspected infringement of a Current Patent Right in the Exclusive Field of the Party directly licensed by JHU under such Current Patent Right, the Party directly licensed by JHU under such Current Patent Right shall have the sole right, at its own expense, to institute and control suit against any Third Party with respect to such infringement and recover any damages, awards or settlements resulting therefrom (subject to any obligations to JHU with respect to such amounts pursuant to the applicable Existing JHU License Agreement). Such Party (or JHU, as applicable) shall keep the other Party and JHU reasonably informed of the status of any such enforcement action. Such other Party and JHU shall reasonably cooperate in any such litigation at the enforcing Party's request and expense. If required by applicable law, the other Party and JHU shall permit action under this Section 3(b)(ii) to be brought in its name, including being joined as party-plaintiff. To the extent permitted by applicable law, the other Party or JHU may join such suit at its own expense. No settlement, consent judgment or other voluntary final disposition of such suit that concedes the invalidity or unenforceability of any patent within Current Patent Rights may be entered into without the prior written consent of JHU, which consent shall not be unreasonably withheld.
  - 3. With respect to an actual or suspected infringement of a Current Patent Right in the Exclusive Field of the Party sublicensed under such Current Patent Right, if such sublicensed Party desires to enforce such Current Patent Right against such infringement, such sublicensed Party shall send a written request for such enforcement right to the other Party and the Parties shall discuss in good faith the terms under which such sublicensed Party may enforce such Current Patent Right, including the allocation between the Parties of any damages, awards or settlements resulting therefrom (subject to any obligations to JHU with respect to such amounts pursuant to the applicable Existing JHU License Agreement). If the Parties agree on such enforcement terms, such terms shall be documented in writing and the Parties and JHU shall be bound thereby. If the Parties do not, within a reasonable period of time, agree on such enforcement terms, the Parties shall discuss such enforcement with JHU and any such enforcement action shall be

determined in the sole discretion of JHU, such determination to include a fair and reasonable allocation between the Parties of any damages, awards or settlements resulting therefrom, after taking into consideration litigation expenses and costs.

# iii. Kala/JHU Joint Patent Rights (Other Than Hatch Waxman).

- Each Party and JHU will notify the other Party(ies) and JHU, as applicable, promptly in writing
  when any infringement of any Kala/JHU Jointly Owned Patent Right by a Third Party is uncovered
  or suspected.
- 2. Except as otherwise provided in Section 3(b)(i), with respect to any actual or suspected infringement of a Kala/JHU Jointly Owned Patent Right, Kala shall have the sole right, at its own expense, to institute and control suit against any Third Party with respect to such infringement and recover any damages, awards or settlements resulting therefrom (subject to any obligations to JHU with respect to such amounts pursuant to the applicable Existing JHU License Agreement). Kala shall keep GrayBug and JHU reasonably informed of the status of any such enforcement action. GrayBug and JHU shall reasonably cooperate in any such litigation at Kala's request and expense. If required by applicable law, GrayBug and JHU shall permit action under this Section 3(b)(iii)(2) to be brought in its name, including being joined as party-plaintiff. To the extent permitted by applicable law, GrayBug or JHU may join such suit at its own expense. No settlement, consent judgment or other voluntary final disposition of such suit that concedes the invalidity or unenforceability of any patent within Kala/JHU Jointly Owned Patent Rights may be entered into without the prior written consent of JHU, which consent shall not be unreasonably withheld.
- 3. With respect to an actual or suspected infringement of a Kala/JHU Jointly Owned Patent Right in the GrayBug Field of Use, if GrayBug desires to enforce such Kala/JHU Jointly Owned Patent Right against such infringement, GrayBug shall send a written request for such enforcement right to Kala and the Parties shall discuss in good faith the terms under which GrayBug may enforce such Kala/JHU Jointly Owned Patent Right, including the allocation between the Parties of any damages, awards or settlements resulting therefrom (subject to any obligations to JHU with respect to such amounts pursuant to Kala's Existing JHU License Agreement). If the Parties agree on such enforcement terms, such terms shall be documented in writing and the Parties and JHU shall be bound thereby. If the Parties do not, within a

reasonable period of time, agree on such enforcement terms, the Parties shall discuss such enforcement with JHU and any such enforcement action shall be determined by Kala and JHU, such determination to include a fair and reasonable allocation between the Parties of any damages, awards or settlements resulting therefrom, after taking into consideration litigation expenses and costs.

# iv. Co-Licensed Patent Rights (Other Than Hatch Waxman).

- Each Party and JHU will notify the other Party(ies) and JHU, as applicable, promptly in writing
  when any infringement of any Co-Licensed Other JHU Patent Right or Future Co-Licensed Patent
  Right (each, a "<u>Co-Licensed Patent Right</u>") by a Third Party is uncovered or suspected.
- 2. Except as otherwise provided in Section 3(b)(i), with respect to an actual or suspected infringement of a Co-Licensed Patent Right, the Parties shall discuss in good faith whether enforcement of such Co-Licensed Patent Right is appropriate and the terms under which a Party(ies) may enforce such Co-Licensed Patent Right, including the allocation between the Parties of any damages, awards or settlements resulting therefrom (subject to any obligations to JHU with respect to such amounts pursuant to the applicable Existing JHU License Agreement). If the Parties agree on such enforcement terms, such terms shall be documented in writing and the Parties and JHU shall be bound thereby. If the Parties do not, within a reasonable period of time, agree on such enforcement terms, the Parties shall discuss such enforcement with JHU and any such enforcement action shall be determined in the sole discretion of JHU, such determination to include a fair and reasonable allocation between the Parties of any damages, awards or settlements resulting therefrom, after taking into consideration litigation expenses and costs.
- c. <u>Parallel Imports</u>. For so long as a Party retains a license or sublicense to a Patent Right in one country but no longer has a license or sublicense of a related Patent Right in another country, the other Party covenants not to assert against the Party related Patent Rights in other countries for parallel imports into such other countries of otherwise infringing products sold by or on behalf of the Party in the Party's Exclusive Field.
- d. <u>Declaratory Judgment</u>. If a declaratory judgment action is brought naming a Party or JHU as a defendant and alleging invalidity of any of the JHU Patent Rights, the provisions

of the relevant Existing JHU License Agreement(s) shall apply, unless the action or claim arises in connection with a litigation under 21 U.S.C. § 355, the Hatch Waxman Act, or similar United States or foreign statutory provisions, in which case, Section 3(b)(i) shall apply.

# 4. <u>Financial Terms</u>.

- a. Kala and GrayBug acknowledge that the cross-licenses and releases provided in this Agreement constitute full consideration for the Agreement. Neither Kala nor GrayBug shall owe the other any royalties, milestone payments, or other payments with respect to the sublicenses granted to the other Party hereunder.
- b. JHU agrees that neither Party shall be responsible for paying to JHU any sublicense fees or other payments on Exhibit A of such Party's Existing JHU License Agreement (including milestone payments and royalties) that may otherwise have arisen under such Existing JHU License Agreement as a result of such Party's granting to the other Party any sublicenses under this Agreement (or such sublicensed Party granting any further sublicenses), or, with respect to any of the Kala/JHU Jointly Owned Patent Rights, Kala's granting to GrayBug any licenses under this Agreement under Kala's ownership interest in such Kala/JHU Jointly Owned Patent Rights (or GrayBug granting any sublicenses). For the avoidance of doubt, Kala and GrayBug shall each remain responsible to JHU for sublicense fees and other payments for any other sublicenses it grants under its Existing JHU License Agreements.
- c. Each Party acknowledges that, except as otherwise expressly set forth in this Agreement, any JHU Patent Rights sublicensed to such Party by the other Party pursuant to this Agreement shall be treated as if they were "PATENT RIGHTS" (as defined in the sublicensed Party's Existing JHU License Agreement) for purposes of payment obligations, including patent payments, and the royalty, sublicense income and milestone payments owed by such sublicensed Party to JHU under items 3, 4 and 5 of Exhibit A of such sublicensed Party's Existing JHU License Agreement; however, all such payments shall be subject to Section 4(d), below.
- d. JHU agrees that the milestones and minimum annual royalties (items 2 and 5 of Exhibit A) of each Existing JHU License Agreement are hereby amended such that all payments due to JHU thereunder during the term of this Agreement are reduced by twenty-five percent (25%). Furthermore, the following additional changes to the financial terms in each Existing JHU License Agreement shall apply (all other financial terms, such as running royalty, sublicensing consideration, and patent costs, shall be unaffected except as otherwise expressly provided herein):

- i. minimum annual royalties which would otherwise be due to JHU at any time between the Settlement Agreement Effective Date and the [\*\*] anniversary of the Settlement Agreement Effective Date are hereby eliminated, and, thereafter, if, at the time the minimum annual royalties are due to JHU, the only Patent Rights owned (in whole or in part) by JHU licensed to the paying Party are those with respect to which JHU's rights in such Patent Right are only non-exclusively licensed to such Party in its Exclusive Field or those which are jointly owned by such Party and JHU, then such minimum annual royalty payment shall be reduced by [\*\*] percent ([\*\*]%);
- ii. the development milestones in Section 5 of Exhibit A (e.g., those milestones which do not relate to first commercial sale) under such agreement shall not apply to the second and third "LICENSED PRODUCT(S)" (as defined in such agreement) to achieve such milestone, but the payments (after the application of the [\*\*] percent ([\*\*]%) reduction described above) applicable to the first "LICENSED PRODUCT" to achieve any such milestone shall apply, and the due date of payment for such milestone is extended [\*\*] months from the due date set forth in the applicable Existing JHU License Agreement as of the Settlement Agreement Effective Date; provided, however, that, if, during such [\*\*] month period, the relevant Party terminates the development of such first "LICENSED PRODUCT" which achieved such milestone, such milestone payment shall not be due unless such terminated development is reinitiated after the extended twelve months, under which circumstance the payment shall be reduced by [\*\*]%) percent and due [\*\*] months after reinitiation;
- iii. with respect to each "LICENSED PRODUCT", the milestone payments with respect to first commercial sale of such "LICENSED PRODUCT" in the United States shall be reduced by an additional [\*\*] percent ([\*\*]%), for a total reduction of [\*\*] percent ([\*\*]%) from the rates set forth in the applicable Existing JHU License Agreement prior to the Settlement Agreement Effective Date;
- iv. if the only "VALID CLAIMS" (as defined in such agreement) owned (in whole or in part) by JHU that cover the relevant "LICENSED PRODUCT" or "LICENSED SERVICE" (as defined in such agreement) are those with respect to which JHU's rights are only non-exclusively licensed to such Party or those which are jointly owned by such Party and JHU, then the milestone payments (including any development and first commercial sale milestone payments) and royalties due to JHU with respect to such "LICENSED PRODUCT" or "LICENSED SERVICE" shall be further reduced by [\*\*] percent ([\*\*]%), provided that by no operation(s)

(except Section 4(d)(ii) above) shall the payments to JHU be reduced by more than [\*\*] percent ([\*\*]%);

- v. if a payment of "SUBLICENSE CONSIDERATION" (as defined in such agreement) is received by a Party only as a result of a Patent Right owned (in whole or in part) by JHU with respect to which JHU's rights are only non-exclusively licensed to such Party or which Patent Right is jointly owned by such Party and JHU, then the percentage of such "SUBLICENSE CONSIDERATION" due to JHU shall be further reduced by [\*\*] percent ([\*\*]%), provided that by no operation(s) shall such payments to JHU fall below [\*\*] percent ([\*\*]%);
- vi. the following sales-based milestones shall be added to Exhibit A, as a new Section 6, of each Existing JHU License Agreement, with the terms which are included but not defined below having the meaning set forth in such Existing JHU License Agreement:
  - "6. Sales-Based Milestones: In addition to the milestones payable under item 5 of this Exhibit A, Company shall pay the following sales-based milestones:
  - i. with respect to the second LICENSED PRODUCT (other than the first LICENSED PRODUCT for which (y) the Company achieves NET SALES of at least [\*\*] dollars (\$[\*\*]) in a calendar year or (y) a SUBLICENSEE achieves royalty income from sales of such LICENSED PRODUCT and Company has received some form of consideration from such SUBLICENSEE (the "First Sold Licensed Product")) for which (A) the Company achieves NET SALES of at least [\*\*] dollars (\$[\*\*]) in a calendar year or (B) a SUBLICENSEE achieves royalty income from sales of such LICENSED PRODUCT and Company has received some form of consideration from such SUBLICENSEE (the "Second Sold Licensed Product"), Company shall pay JHU as follows:
    - 1. if the event in clause (A) occurs before the event in clause (B): JHU shall be owed an amount equal to the sum of [\*\*] dollars (\$[\*\*]), plus the amount by which, pursuant to Section 4(d) (ii) of the Settlement and License Agreement entered into as of October 24, 2014, by and between GrayBug, LLC, Kala Pharmaceuticals, Inc. and The Johns Hopkins University (the "Settlement Agreement"), each of the development milestones in Section 5 of this Exhibit A for the second "LICENSED PRODUCT" to achieve such milestone was reduced, plus the amount by which, pursuant to Section 4(d)(iii) of the Settlement Agreement, the milestone payment with respect to first commercial sale of the second "LICENSED PRODUCT" in the United States was reduced (such sum, the "First Deferral Reimbursement Amount"); [\*\*]

of such amount to be paid to JHU by [\*\*] after which the Company obtained such NET SALES, and the remainder due by [\*\*]; or

- 2. if the event in clause (B) occurs before the event in clause (A): JHU shall be owed an amount equal to the First Deferral Reimbursement Amount, which shall be paid by increasing the applicable sublicensee royalty rates set forth in Section 4 of Exhibit A (as modified by the Settlement Agreement) due to JHU with respect to sales of the Second Sold Licensed Product by [\*\*]%, until the full amount of such First Deferral Reimbursement Amount has been paid to JHU.
- ii. with respect to the third LICENSED PRODUCT (other than the First Sold Licensed Product and the Second Sold Licensed Product) for which (A) the Company achieves NET SALES of at least [\*\*] dollars (\$[\*\*]) in a calendar year or (B) a SUBLICENSEE achieves royalty income from sales of such LICENSED PRODUCT and Company has received some form of consideration from such SUBLICENSEE (the "Third Sold Licensed Product"), Company shall pay JHU as follows:
  - 1. if the event in clause (A) occurs before the event in clause (B): JHU shall be owed an amount equal to [\*\*] dollars (\$[\*\*]), plus the amount by which, pursuant to Section 4(d)(ii) of the Settlement Agreement, each of the development milestones in Section 5 of this Exhibit A for the third "LICENSED PRODUCT" to achieve such milestone was reduced, plus the amount by which, pursuant to Section 4(d)(iii) of the Settlement Agreement, the milestone payment with respect to first commercial sale of the third "LICENSED PRODUCT" in the United States was reduced (such sum, the "Second Deferral Reimbursement Amount"); [\*\*] of such amount to be paid to JHU by [\*\*] after which the Company obtained such NET SALES, and the remainder due by [\*\*]; or
  - 2. if the event in clause (B) occurs before the event in clause (A): JHU shall be owed an amount equal to the Second Deferral Reimbursement Amount, which shall be paid by increasing the applicable sublicensee royalty rates set forth in Section 4 of Exhibit A of (as modified by the Settlement Agreement) due to JHU with respect to sales of the Third Sold Licensed Product by [\*\*]%, until the full amount of such Second Deferral Reimbursement Amount has been paid to JHU.
- iii. The parties agree and acknowledge that (i) the Second Sold Licensed Product may be a different LICENSED PRODUCT than the LICENSED PRODUCT to

be second to achieve one or more of the milestones described in Section 4(d)(ii) or 4(d)(iii) of the Settlement Agreement, and (ii) the Third Sold Licensed Product may be a different LICENSED PRODUCT than the LICENSED PRODUCT to be third to achieve one or more of the milestones described in Section 4(d)(ii) or 4(d)(iii) of the Settlement Agreement, and (iii) the First Sold Licensed Product may be a different LICENSED PRODUCT than the LICENSED PRODUCT to be first to achieve one or more of the milestones described in Section 4(d)(iii) or 4(d)(iii) of the Settlement Agreement.

#### Releases.

- a. Each Party, on behalf of itself and its predecessors, successors, assigns, employees, officers and directors (each, a, "Releasor"), fully, finally and forever releases, relinquishes, acquits and discharges the other Party and each of its predecessors, successors, assigns, employees, officers and directors (each, a "Releasee"), of and from, and covenants not to sue, not to assign to any other Person a right to sue and not to authorize any other Person to sue any Releasee for, any and all Losses of every name and nature, both at law and in equity, accrued or unaccrued, that (i) any of such Party's Releasors is aware of as of the Settlement Agreement Effective Date, (ii) arise out of the actions of the Parties or JHU leading to the execution of this Agreement and (iii) relate to the subject matter hereof. This Section 5(a) shall not prevent or impair the right of a Releasor to bring a proceeding in court or any other forum to enforce its rights under this Agreement or its Existing JHU License Agreement (except as explicitly or implicitly amended by this Agreement).
- b. Each Party (on behalf of itself and its other Releasors) fully, finally and forever releases, relinquishes, acquits and discharges JHU, The Johns Hopkins Hospital, The Johns Hopkins Health System Corporation, and their affiliated entities, their present and former trustees, officers, INVENTORS (as defined in such Party's Existing JHU License Agreement), agents, faculty, employees and students (each, a "JHU Releasee") of and from, and covenants not to sue, not to assign to any other Person a right to sue and not to authorize any other Person to sue any JHU Releasee for, any and all Losses of every name and nature, both at law and in equity, accrued or unaccrued, that (i) any of such Releasors is aware of as of the Settlement Agreement Effective Date, (ii) arise out of the actions of the Parties or JHU Releasees leading to the execution of this Agreement and (iii) relate to the subject matter hereof. This Section 5(b) shall not prevent or impair the right of a Releasor to bring a proceeding in court or any other forum to enforce its rights under this Agreement or its Existing JHU License Agreement (except as explicitly or implicitly amended by this Agreement).
- c. Each Party and JHU acknowledges that the releases and other terms of this Agreement settle all known disputes as of the Settlement Agreement Effective Date relating to

subject matter licensed to a Party from JHU, and after the Settlement Agreement Effective Date, neither Party nor JHU may assert a claim that the other Party or JHU, or their respective predecessors, successors, assigns, directors, officers, employees, trustees, INVENTORS (as defined in each Party's Existing JHU License Agreement), faculty, students or agents released hereunder is in violation of any Existing JHU License Agreement in connection with such acts or omissions prior to the Settlement Agreement Effective Date.

# 6. <u>Representations and Warranties</u>.

- a. Each Party represents, warrants and covenants to the other Party that:
  - as of the Settlement Agreement Effective Date, it has the authority and right to enter into and perform this
    Agreement and grant the rights and licenses herein, and it is not aware of any legal impediment that could
    inhibit its ability to perform its obligations under this Agreement;
  - ii. as of the Settlement Agreement Effective Date, its execution, delivery and performance of this Agreement does not conflict with, or constitute a breach of, any order, judgment, agreement or instrument to which it is a party or is otherwise bound;
  - iii. as of the Settlement Agreement Effective Date, no consent of any third party (other than JHU) is required for such Party to grant the licenses and rights granted to the other Party under this Agreement or to perform its obligations hereunder; and
  - iv. it will not, after the Settlement Agreement Effective Date, enter into any written or oral contractual obligation with any third party (including JHU) that would be inconsistent with the obligations that arise on its part out of this Agreement or that would deprive the other Party of the benefits of or rights granted under this Agreement.
- b. GrayBug represents, warrants and covenants to Kala that the GrayBug Current Patent Rights are all the Patent Rights licensed to GrayBug by JHU as of the Settlement Agreement Effective Date. Kala represents, warrants and covenants to GrayBug that the Kala Current Patent Rights and the Kala/JHU Jointly Owned Patent Rights are all the Patent Rights licensed to Kala by JHU as of the Settlement Agreement Effective Date.
- C. EXCEPT AS EXPRESSLY SET FORTH IN THIS SECTION 6 OR AS EXPRESSLY PROVIDED BY JHU TO THE APPLICABLE PARTY UNDER AN EXISTING JHU LICENSE AGREEMENT, EACH PARTY AGREES THAT THE PATENT RIGHTS LICENSED HEREUNDER ARE PROVIDED "AS IS", AND THAT NEITHER JHU NOR THE OTHER PARTY MAKES ANY REPRESENTATION OR

WARRANTY WITH RESPECT TO THE PERFORMANCE OF LICENSED PRODUCTS AND LICENSED SERVICES, INCLUDING THEIR SAFETY, EFFECTIVENESS, OR COMMERCIAL VIABILITY. EACH PARTY AND JHU DISCLAIM ALL WARRANTIES WITH REGARD TO PRODUCTS AND SERVICES LICENSED UNDER THIS AGREEMENT, INCLUDING ALL WARRANTIES, EXPRESSED OR IMPLIED, OF MERCHANTABILITY AND FITNESS FOR ANY PARTICULAR PURPOSE.

# 7. <u>Confidentiality</u>.

- a. The Parties agree that during the term of this Agreement, and for a period of [\*\*] years thereafter, each Party shall (i) maintain in confidence the Confidential Information of the other Party to the same extent such Party maintains its own confidential information, (ii) not disclose such Confidential Information to any Third Party without the prior written consent of the other Party, and (iii) not use such Confidential Information for any purpose except those permitted by this Agreement.
- b. <u>Authorized Disclosure</u>. Notwithstanding Section 7(b), a Party may disclose Confidential Information of the other Party to the extent such disclosure is reasonably necessary in the following instances:
  - disclosure to each applicable country's patent office to the extent necessary for the prosecution and maintenance of Patent Rights in accordance with this Agreement;
  - making necessary filings with regulatory authorities with respect to Licensed Products or Licensed Services in the receiving Party's Exclusive Field;
  - iii. complying with applicable laws or submitting information to governmental authorities; <u>provided that</u> if a Party is required by law to make any public disclosure of Confidential Information of the other Party, to the extent it may legally do so, it will give reasonable advance notice to the other Party of such disclosure and will use its reasonable efforts to secure confidential treatment of such Confidential Information prior to its disclosure (whether through protective orders or otherwise); or
  - iv. to the extent mutually agreed to in writing by the Parties.
- c. Each Party may disclose the existence and terms of this Agreement to its Affiliates and to prospective and actual acquirers, licensors, collaborators, licensees, sublicensees, lenders, financing sources, investment bankers, employees, contractors, agents, accountants, lawyers, consultants, auditors, advisors and investors, on a need to know basis, each of whom prior to disclosure must be bound by written or professional ethical obligations of confidentiality and non-use equivalent in scope to those set forth in this

Section 7 and that are of reasonable duration in view of the circumstances of the disclosure.

- d. JHU agrees that any JHU Confidential Information may be disclosed by a Party in accordance with this Section 7.
- e. Nothing herein grants JHU the right to disclose to a Party (i) the Existing JHU License Agreement of the other Party or (ii) any proprietary materials, information, records or data of the other Party.
- f. Nothing herein requires either Party to disclose or produce its Existing JHU License Agreement to the other Party.
- g. In no event shall this Section 7 (Confidentiality) limit JHU's ability to disclose or publish its own Confidential Information.

# 8. <u>Indemnification.</u>

- a. Each Party shall indemnify, hold harmless and defend the other Party, its Affiliates, and all of their respective officers, directors, employees, agents, licensors and shareholders (collectively, the "<u>Indemnitees</u>") from and against any and all losses, damages, liabilities, judgments, fines, amounts paid in settlement, expenses and costs of defense (including reasonable attorneys' fees and witness fees) resulting from any demand, claim, action or proceeding brought or initiated by a Third Party (each a "<u>Third Party Claim</u>") against any Indemnitee to the extent arising out of (a) the research, development, making, having made, use, having used, importation, having imported, offering to sell, having offered for sale, sale and having sold Licensed Products, and the provision and having provided Licensed Services, by or on behalf of the Indemnifying Party, in the United States and worldwide, (b) a breach of this Agreement by such Party, or (c) the gross negligence or willful misconduct of such Party.
- b. As a condition to indemnification hereunder, (i) the indemnified Party shall provide the indemnifying Party with prompt notice of the Third Party Claim giving rise to the indemnification obligation under Section 8(a), (ii) the Indemnitees shall reasonably cooperate with the indemnifying Party, at the indemnifying Party's reasonable request and reasonable expense with respect to out-of-pocket costs, in the defense of such claim, and (iii) the Indemnitees shall permit the indemnifying Party to exclusively defend or settle any such claim; provided however that the indemnifying Party shall not enter into any settlement for damages, or that imposes upon any Indemnitee any obligation or liability, without such Indemnitee's prior written consent, such consent not to be unreasonably withheld, delayed or conditioned. The Indemnitees shall have the right to participate, at its own expense and with counsel of its choice, in the defense of any claim or suit that has been assumed by the indemnifying Party.

c. Each Party shall indemnify JHU with respect to that Party's research, development, making, having made, use, having used, importation, having imported, offering to sell, having offered for sale, sale and having sold Licensed Products in the indemnifying Party's Exclusive Field, and the provision and having provided Licensed Services in the indemnifying Party's Exclusive Field, in the United States and worldwide, in accordance with the indemnification procedures set forth in such Party's Existing JHU License Agreement, mutatis mutandis.

#### 9. Termination.

- a. This Agreement shall remain in effect until the expiration of all JHU Patent Rights, Other JHU Patent Rights, Future Patent Rights and Kala/JHU Jointly Owned Patent Rights unless earlier terminated in accordance with this Section 9.
- b. If a Party (the "<u>Disparaging Party</u>"), or its employees, officers, directors, agents or representatives, Disparages a Current Patent Right or Kala/JHU Jointly Owned Patent Right directly licensed by JHU to the other Party (the "<u>Disparaged Party</u>"), then the Disparaged Party may, upon written notice to the Disparaging Party, immediately terminate one or more of its sublicenses granted to the Disparaging Party under this Agreement, including any further sublicenses granted by the Disparaging Party. "<u>Disparagement</u>" (including, with correlative meanings, "<u>Disparage</u>") means, with respect to a Patent Right, offensively opposing (including filing an opposition or declaration judgment action) or taking an independent action to attempt to invalidate, or formal steps prevent from issuing, such Patent Right, including by making written statements that such Patent Right is unpatentable; <u>provided</u>, <u>however</u>, that Disparagement shall not include any comments or arguments made by patent counsel for a Party before a patent office during the course of prosecution of any patent application owned by or licensed to such Party anywhere in the world. Notwithstanding the foregoing, each Party shall have the right to defend itself, including in a litigation brought by the other Party, using any applicable position. For example, if a Party sues the other Party for infringement, the purported infringer can defend itself with a position that the patentee's asserted claim is invalid.
- c. Each Party may terminate any of the rights and licenses granted to it under this Agreement or under such Party's Existing JHU License Agreement, on a Patent Right-by-Patent Right and country-by-country basis, by providing thirty (30) calendar days prior written notice thereof to the other Party and JHU, following which, but subject to Section 9(d), the terminating Party's rights and obligations hereunder and under its Existing JHU License Agreement shall terminate with respect to such Patent Right in such country. For clarity, termination of a Party's rights and licenses under a Patent Right in a country shall not be deemed to have occurred merely because a particular

- patent application has been abandoned if such patent application continues to be Prosecuted and Maintained, or issues, through a Counterpart.
- d. Expiration or termination of this Agreement, or of any of the licenses under this Agreement, shall not relieve the Parties of any obligation accruing prior to such expiration or termination. Termination of such licenses shall be in addition to, and shall not prejudice, the Parties' remedies at law or in equity, including the Parties' ability to receive legal damages or equitable relief with respect to any Disparagement.
- e. All remaining applicable provisions of this Agreement shall survive its expiration or termination.

#### 10. <u>Miscellaneous</u>.

- a. <u>Affirmative Amendment; Inconsistent Terms</u>. This Agreement affirmatively amends each Party's Existing JHU License Agreement as necessary to reflect the terms of this Agreement.
- b. <u>Assignment</u>. Neither this Agreement nor any right or obligation hereunder may be assigned or otherwise transferred by any Party without the consent of the other Party; <u>provided</u>, <u>however</u>, that any Party may, without such consent, assign this Agreement, in whole or in part: (a) to any of its Affiliates; <u>provided that</u> the assigning Party shall remain jointly and severally liable with such Affiliate(s) in respect of all obligations so assigned unless such Affiliate(s) are fully capable of fulfilling all the assigning Party's obligations under this Agreement; or (b) to any successor in interest by way of merger, acquisition or sale of all or substantially all of its assets to which this Agreement relates. Any assignment not in accordance with the foregoing shall be void.
- c. No Waivers; Severability. No waiver of any breach of this Agreement shall constitute a waiver of any other breach of the same or other provision of this Agreement, and no waiver shall be effective unless made in writing. Any provision hereof prohibited by or unenforceable under any applicable law of any jurisdiction shall as to such jurisdiction be deemed ineffective and deleted herefrom without affecting any other provision of this Agreement. Should any provision contained herein be held by any governmental agency or court of competent jurisdiction to be void, illegal and unenforceable, the Parties shall negotiate in good faith for a substitute term or provision which carries out the original intent of the Parties.
- d. <u>Entire Agreement; Amendment</u>. The Parties acknowledge that they have read this entire Agreement and that this Agreement constitutes the entire understanding and contract between the Parties with respect to the subject matter hereof and supersedes any and all prior or contemporaneous oral or written communications with respect to the subject matter hereof (including the Prior CDA). It is expressly understood and

agreed that there being no expectations to the contrary between the Parties, no usage of trade, verbal agreement or another regular practice or method of dealing within any industry or between the Parties shall be used to modify, interpret, supplement or alter in any manner the express terms of this Agreement. This Agreement shall not be modified, amended or in any way altered except by an instrument in writing signed by both of the Parties and, with respect to any provision applicable to JHU's rights, JHU.

- e. <u>No Inconsistent Amendments</u>. Neither Party shall amend its Existing JHU License Agreement in a manner that is inconsistent with its or JHU's obligations to the other Party under this Agreement.
- f. <u>No Partnership</u>. Nothing in this Agreement shall be construed to create any agency, employment, partnership, joint venture or similar relationship between the Parties other than that of a licensor/licensee. Neither Party shall have any right or authority whatsoever to incur any liability or obligation (express or implied) or otherwise act in any manner in the name or on the behalf of the other, or to make any promise, warranty or representation binding on the other.
- g. <u>Governing Law</u>. This Agreement shall be construed, and legal relations between the parties hereto shall be determined, in accordance with the laws of the State of New York applicable to contracts solely executed and wholly to be performed within the State of New York without giving effect to the principles of conflicts of laws.
- h. <u>Dispute Resolution</u>. Each Party will appoint one representative (each a "<u>Representative</u>") to serve as the main point of contact for activities relating to this Agreement. Neither Party will file suit against the other Party with respect to a dispute arising in connection with this Agreement except as permitted under this Section 10(h). In the event that any dispute arises between the Parties in connection with this Agreement, the Parties will, upon written notice from either Party to the other Party (the "<u>First Notice</u>"), first attempt to resolve it through good faith negotiations among the Representatives. If the Representatives fail to resolve the dispute within [\*\*] days of the First Notice, then upon written notice from one Party to the other (the "<u>Second Notice</u>"), the Parties will escalate the dispute to their respective CEOs who will also attempt to resolve the dispute by good faith negotiations. In the event the Parties are unable to resolve such dispute within [\*\*] days of the Second Notice, either Party may, subject to Section 10(g), proceed to file suit regarding the dispute without further obligations under this Section 10(h) with respect to such dispute (including any claims and counterclaims which may be subsequently added or modified). Notwithstanding the foregoing, a Party may at any time seek or obtain preliminary, interim, or conservatory measures from a court.

i. Notice. All notices or communication required or permitted to be given by either Party hereunder shall be deemed sufficiently given if sent by overnight courier providing evidence of delivery, such as Federal Express or United States Postal Service (USPS) overnight, to the other Party at its respective address set forth below or to such other address as one Party shall give notice of to the other from time to time hereunder, and a courtesy copy shall be sent by email to the email address(es) shown below or to such other email address(es) as one Party shall give notice of to the other from time to time hereunder. Notices shall be deemed received the following Business Day.

If to GrayBug: GrayBug, LLC

[\*\*]

Baltimore, MD 21203

Attn: Chief Executive Office (currently Michael O'Rourke)

Email: [\*\*]

With a Copy to: [\*\*]

If to Kala: Kala Pharmaceuticals, Inc.

100 Beaver Street

Suite 201

Waltham, MA 02453

Attn: President (currently Charlie McDermott, Interim President)

Email: [\*\*]

With a Copy to: Chief IP Counsel (at the same address)

Email: [\*\*]

Notices to JHU shall be provided by the relevant Party in accordance with the provisions of its Existing JHU License Agreement.

- j. <u>Compliance with All Laws</u>. In all activities undertaken pursuant to this Agreement, both Parties covenant and agree that each will in all material respects comply with such federal, state and local laws and statutes, as may be in effect at the time of performance and all valid rules, regulations and orders thereof regulating such activities. Each Party has confirmed that it is not required to make a filing under the HSR Act with respect to the rights granted to it as of the Settlement Agreement Effective Date under this Agreement.
- k. <u>Delays or Omissions</u>. Except as expressly provided herein, no delay or omission to exercise any right, power or remedy accruing to any Party shall impair any such right, power or remedy to such Party nor shall it be construed to be a waiver of any such breach or default, or an acquiescence therein, or in any similar breach or default be

deemed a waiver of any other breach or default theretofore or thereafter occurring. Any waiver, permit, consent or approval of any kind or character on the part of any Party of any breach or default under this Agreement, or any waiver on the part of any Party of any provisions or conditions of this Agreement, must be in writing and shall be effective only to the extent specifically set forth in such writing. All remedies either under this Agreement or by law or otherwise afforded to any Party, shall be cumulative and not alternative.

- No Third Party Beneficiaries. Nothing in this Agreement shall be construed as giving any Person, other than the
  Parties, JHU (to the extent set forth herein) and their successors and permitted assigns, any right, remedy or claim
  under or in respect of this Agreement or any provision hereof.
- m. Headings. Section headings are for convenient reference and not a part of this Agreement.
- n. <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original and all of which when taken together shall be deemed but one instrument.

[Signature Page Follows]

IN WITNESS WHEREOF, this Agreement shall take effect as of the Settlement Agreement Effective Date when it has been executed below by the duly authorized representatives of the Parties.

GrayBug, LLC

Kala Pharmaceuticals, Inc.

BY: /s/ Michael O'Rourke BY: /s/ Charles McDermott

NAME: Michael O'Rourke NAME: Charles McDermott

TITLE: President and Chief Executive Officer TITLE: Chief Business Officer and Interim President

[JHU Signature Page Attached]

#### JHU Signature Page, including Kala and GrayBug Signatures

WHEREAS, there is a dispute over the rights licensed to each Party under certain Patent Rights owned by JHU;

WHEREAS, the Parties and JHU now wish to settle such dispute in an amicable manner which is intended to settle and clarify rights granted by JHU to each Party and to permit each Party to have the freedom to effectively develop pharmaceutical products that will benefit patients;

WHEREAS, at JHU's request, GrayBug and Kala have undertaken extensive time and effort to work cooperatively to negotiate the terms of the Settlement and License Agreement to which this JHU Signature Page is attached (the "Settlement Agreement") that sets forth the terms and conditions for settling all currently outstanding disputes between JHU and the Parties, and each of their respective directors, officers, employees, and agents; and

WHEREAS, the Settlement Agreement provides for resolution of all known disputes as of the Settlement Agreement Effective Date relating to the scope of subject matter licensed to a Party from JHU, including releases of the Parties and JHU, and their respective predecessors, successors, assigns, employees, officers and directors, as set forth in Section 5 of the Settlement Agreement, which include releasing JHU in connection with any allegations of breach of contract or other potential causes of action with respect to such subject matter.

JHU acknowledges, accepts, and agrees with GrayBug and Kala that:

- The capitalized terms used but not defined in on this JHU Signature Page have the meaning ascribed to them in the Settlement Agreement.
- Kala, GrayBug and JHU each agrees to, and shall comply with, all the terms and conditions of the Settlement Agreement to the extent relevant or applicable to such Party or JHU, as applicable. Without limitation, it is confirmed that the provisions of Sections 1-7 and 10 of the Settlement Agreement shall apply to JHU and to the paragraphs set forth on this JHU Signature Page to the same extent as they apply to either Party.
- Each Existing JHU License Agreement is hereby amended to reflect the terms set forth in the Settlement Agreement, to the extent necessary to implement the intent, terms, and provisions of the Settlement Agreement, including Sections 2, 3, 4, 7, and 9 of the Settlement Agreement.
- As set forth in more detail and in accordance with Section 5(b) of the Settlement Agreement, Kala (on behalf of itself and its
  other Releasors (as defined in Section 5(a) of the Settlement Agreement)) fully, finally and forever releases, relinquishes,
  acquits and discharges the JHU Releasees (as defined in Section 5(b) of the Settlement Agreement) of and from, and covenants
  not to sue, not to assign to any other Person a right to sue and not to authorize any other Person to sue the JHU Releasees for,
  any and all Losses of every name and nature, both at law

and in equity, accrued or unaccrued, that (i) any of such Releasors is aware of as of the Settlement Agreement Effective Date, (ii) arise out of the actions of the Parties or JHU Releasees leading to the execution of this Agreement and (iii) relate to the subject matter hereof.

- As set forth in more detail and in accordance with Section 5(b) of the Settlement Agreement, GrayBug (on behalf of itself and its other Releasors (as defined in Section 5(a) of the Settlement Agreement)) fully, finally and forever releases, relinquishes, acquits and discharges the JHU Releasees (as defined in Section 5(b) of the Settlement Agreement) of and from, and covenants not to sue, not to assign to any other Person a right to sue and not to authorize any other Person to sue the JHU Releasees for, any and all Losses of every name and nature, both at law and in equity, accrued or unaccrued, that (i) any of such Releasors is aware of as of the Settlement Agreement Effective Date, (ii) arise out of the actions of the Parties or JHU Releasees leading to the execution of this Agreement and (iii) relate to the subject matter hereof.
- JHU (on behalf of itself and the JHU Releasees (as defined on Section 5(b) of the Settlement Agreement)) fully, finally and forever releases, relinquishes, acquits and discharges each Party and each of its predecessors, successors, assigns, employees, officers and directors (each, a "Party Releasee"), of and from, and covenants not to sue, not to assign to any other Person a right to sue and not to authorize any other Person to sue any Party Releasee for, any and all Losses of every name and nature, both at law and in equity, accrued or unaccrued, that (i) any of such JHU Releasees is aware of as of the Settlement Agreement Effective Date, (ii) arise out of the actions of the Parties or JHU Releasees leading to the execution of the Settlement Agreement and (iii) relate to the subject matter thereof. This paragraph shall not prevent or impair the right of JHU to bring a proceeding in court or any other forum to enforce its rights under this Agreement or its Existing JHU License Agreement with a Party (except as explicitly or implicitly amended by the Settlement Agreement).
- JHU acknowledges that the releases and amendments provided in the Settlement Agreement and continued consideration under
  the Parties' Existing JHU License Agreements constitute full consideration for the Settlement Agreement and this JHU
  Signature Page. Notwithstanding any obligation that either Party may have to JHU regarding responsibility for sublicensees
  under an Existing JHU License Agreement, neither Party shall have any liability to JHU for the acts or omissions of the other
  Party, whether as a "Sublicensee" as defined under an Existing JHU License Agreement or otherwise.
- JHU represents and that: (i) it has good and marketable title to its interest in each of the inventions claimed under any PATENT RIGHTS (as defined in each of the Existing JHU License Agreements), GrayBug Current Patent Rights, Kala Current Patent Rights, Other JHU Patent Rights, Kala/JHU Jointly Owned Patent Rights (all such patent rights, collectively, the "<u>LICENSED PATENT RIGHTS</u>"), including by acquiring, through written assignment, the rights of the relevant inventors in such inventions and patent rights therein and having taken all necessary actions in accordance with the Bayh-Dole Act to elect to retain title to such inventions, subject to certain

retained rights (other than ownership) of the United States Government which may apply to such JHU-owned inventions and LICENSED PATENT RIGHTS if any part of the JHU research was funded in whole or in part by the United States Government; (ii) as of the Settlement Agreement Effective Date, it has not entered any contract, license or similar agreement with a third party that restricts, contradicts, interferes with or otherwise limits the grant of rights under this Settlement Agreement; (iii) it has proper authority to enter into this Settlement Agreement and to grant the rights set forth herein; (iv) it has not granted any rights or licenses that may conflict with this Settlement Agreement or the rights and licenses granted herein; (v) to its knowledge, neither JHU nor any assignee of or inventor listed in any of the LICENSED PATENT RIGHTS has received any notice from any third party that the exercise of the LICENSED PATENT RIGHTS infringes or reasonably could be expected to infringe the rights of any third party; and (vi) to its knowledge, all inventors have been named in the LICENSED PATENT RIGHTS.

# ACKNOWLEDGED, ACCEPTED, AND AGREED:

#### THE JOHNS HOPKINS UNIVERSITY

BY: /s/ Jill E. Uhl

NAME: Jill E. Uhl

TITLE: Interim Executive Director

KALA PHARMACEUTICALS, INC.

BY: /s/ Charles McDermott

NAME: Charles McDermott
TITLE: Interim President & CEO

GRAYBUG, LLC

BY: /s/ Michael O'Rourke

NAME: Michael O'Rourke TITLE: President & CEO



# Johns Hopkins Technology Transfer

100 N. Charles Street, 5th Floor Baltimore MD 21201 410-516-8300 / Fax 410-516-4411 www.jhtt.jhu.edu

January 8, 2015

MEMORADUM: CO-INVENTORS

To Whom It May Concern,

As of the Settlement and License Agreement date, October 24, 2014, all potential Co-Inventors listed in the definition of Future Patent Rights were additionally identified as "Pending Co-Inventors" on Schedule 2c(ix). This letter is to inform that [\*\*]Pending Co-Inventors have agreed in writing to become Co-Inventors and subject to the relevant provisions of the Settlement and License Agreement. These Co-Inventors are as follows:

[\*\*]

Sincerely,

/s/ Jill E. Uhl

Jill Uhl Interim Executive Director

Johns Hopkins Technology Transfer



JHIT - Bringing the Benefits of Discovery to the World

#### **CERTIFICATIONS**

- I, Mark Iwicki, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Kala Pharmaceuticals, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 5, 2021 By: /s/ Mark Iwicki

Mark Iwicki
President and Chief Executive Officer
(Principal Executive Officer)

#### **CERTIFICATIONS**

- I, Mary Reumuth, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Kala Pharmaceuticals, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 5, 2021 By: /s/ Mary Reumuth

Mary Reumuth Chief Financial Officer (Principal Financial and Accounting Officer)

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of Kala Pharmaceuticals, Inc. (the "Company") for the period ended June 30, 2021 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned, Mark Iwicki, President and Chief Executive Officer of the Company hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes Oxley Act of 2002, that to the best of his knowledge on the date hereof:

- (1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 5, 2021 /s/ Mark Iwicki

Mark Iwicki
President and Chief Executive Officer
(Principal Executive Officer)

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of Kala Pharmaceuticals, Inc. (the "Company") for the period ended June 30, 2021 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned, Mary Reumuth, Chief Financial Officer of the Company hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes Oxley Act of 2002, that to the best of her knowledge on the date hereof:

- (1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 5, 2021 /s/ Mary Reumuth

Mary Reumuth
Chief Financial Officer
(Principal Financial and Accounting Officer)